Washington State Survey of Adolescent Health Behaviors 2000

Analytic Report

Prepared for

Office of Superintendent of Public Instruction

Old Capitol Building PO Box 47200 Olympia, WA 98504-7200

Prepared by

RMC Research Corporation 522 SW Fifth Avenue, Suite 1407 Portland, OR 97204

May 2001

Submitted to:

Office of Superintendent of Public Instruction Old Capitol Building PO Box 47200 Olympia, WA 98504-7200

Department of Social and Health Services 1115 Washington Street Southeast PO Box 45000 Olympia, WA 98504-5000

Office of Community Development 906 Columbia Street Southwest PO Box 48350 Olympia, WA 98504-8350

> Department of Health 1112 Southeast Quince PO Box 47890 Olympia, WA 98504-7890

> > Submitted by:

RMC Research Corporation 522 SW Fifth Avenue, Suite 1407 Portland, OR 97204-2131

> Eric L. Einspruch, Ph.D. Dennis D. Deck, Ph.D. Philip R. Nickel, M.A. Gwen Hyatt, M.S.

> > May 2001

The Safe and Drug-Free Schools and Communities Act of 1994 (P.L. 103-382); Division of Alcohol and Substance Abuse General State Funds; and National Institutes of Health, National Institute on Drug Abuse Grant Number 42USC241-42-CFR-52 Research jointly funded this project.

Analytic Report

Washington State Survey of Adolescent Health Behaviors (2000)

Dr. Terry Bergeson State Superintendent of Public Instruction

> Marcia Riggers Assistant Superintendent Operations and Support

Denise Fitch Director Learning and Teaching Support

Martin Mueller Program Supervisor Learning and Teaching Support

May 2001

This material is available in alternative format upon request. Contact Safe and Drug-Free Schools, (306) 753-5595 or TDD (360) 664-3631. The Office of Superintendent of Public Instruction complies with all federal and state rules and regulations and does not discriminate on the basis of race, color, national origin, sex, disability, age, or marital status.

Department of Social and Health Services

Dennis Braddock, Secretary

Division of Alcohol and Substance Abuse

Kenneth D. Stark, Director

Fred Garcia Chief of Service Operations

Antoinette Krupski Research and Evaluation Supervisor

Michael Langer Prevention Services Supervisor

Stephen Smothers Prevention Services Lead Worker

and

Office of Community Development

Busse Nutley, Director

Local Government Division

Steve Wells Assistant Director

Paul Perz Managing Director Safe and Drug-Free Communities Unit

Susan Roberts Program Supervisor Community Mobilization Program

and

Department of Health

Mary Selecky, Secretary

Maxine Hayes State Health Officer

Jack Williams Assistant Secretary Community and Family Health Julia Dilley, Epidemiologist Tobacco Prevention & Control Program

Contents

Figures and Tables	iii
Acknowledgements	.vii
Executive Summary	ix
Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use	X
Current Use of Alcohol	xi
Тоbассо	xi
Marijuana	xiii
Attitudes Toward Alcohol, Tobacco, and Other Drug Use	xiii
Intentional Injury Behaviors: Fighting and Weapon Carrying	Xiii
School Climate	xiv
Risk and Protective Factors for Adolescent Health Behaviors	XV
Chapter 1: Introduction	1
Participation	4
Purpose of This Report	7
Caution	8
Chapter 2: Alcohol, Tobacco, and Other Drug Use	13
Introduction	13
Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use	15
Current Use of Alcohol, Tobacco, and Other Drugs	22
Alcohol	28
Тоbассо	34
Marijuana	42
Attitudes Toward Alcohol, Tobacco, and Other Drug Use	45
Chapter 3: Intentional Injury Behaviors	49
Chapter 4: School Climate	57
Chapter 5: Risk and Protective Factors	64
Community Domain	67
School Domain	70
Peer-Individual Domain	73
Summary	79
Chapter 6: Characteristics of the Students Surveyed	81
Gender	81
Geographic Region	82
Urban/ Suburban/Rural School	83
Race/Ethnicity	85
Adults and Other Children at Home	86
Working at a Part-Time Job	.87
Chapter 7: Conclusion	89
References	92
Appendix A Item-Level Frequencies	99

	Appendix B	Selected Results by	Ethnic Group1	35
--	------------	---------------------	---------------	----

Figures and Tables

Figure	1:	Alcohol Use: Grades 6, 8, 10, and 12	. 29
Figure	2:	Trend in Prevalence of Binge Drinking in Past Two Weeks in Grades 6, 8, 10, and 12	. 30
Figure	3:	Composite Scale: Alcohol Use in Grades 6, 8, 10, and 12	.33
Figure	4:	Smoking Tobacco Use in Grades 6, 8, 10, and 12	.35
Figure	5:	Trends in 30-Day Prevalence of Cigarette Use: Grades 6, 8, 10, and 13	236
Figure	6:	Smokeless Tobacco Use in Grades 6, 8, 10, and 12	. 37
Figure	7:	30-Day Smokeless Tobacco Use	.38
Figure	8:	Perceived Risk of Secondhand Smoke	. 39
Figure	9:	Students Who Practiced Tobacco Refusal Skills in Class	. 40
Figure	10:	Students Who Had Discussed the Dangers of Tobacco Use With Parents or Guardians	.41
Figure	11:	Student Exposure to Antismoking Television and Radio Ads in the Past 30 Days	. 42
Figure	12:	Trends in 30-Day Prevalence of Marijuana Use in Grades 6, 8, 10 and 12	. 44
Figure	13:	Trends in Perceived Risk and Binge Drinking in the Past Two Weeks Among Grade 8 Students	. 46
Figure	14:	Trends in Perceived Risk and 30–Day Use of Marijuana Among Grade 8 Students in Washington	. 47
Figure	15:	Weapon Carrying in the Past 30 Days	. 50
Figure	16:	Prevalence of Attacking Someone With the Idea of Seriously Hurting That Person	9 .51
Figure	17:	Percent of Students Who Had Belonged to a Gang	. 52
Figure	18:	Prevalence of Violent and Delinquent Behavior in the Past Year	. 54
Figure	19:	Students Reporting Experience of Depressive Feelings	. 55
Figure	20:	Students Who Experienced Bullying Behaviors at School	.61
Figure	21:	Students' Reactions in Bullying Situations	. 62
Figure	22:	Students Who Carried a Weapon to School in the Past Month	. 63
Figure	23:	The Relationship Between Alcohol and Drug Use and the Number of Risk Factors Reported by Washington Students	.78

Figure 24:	The Relationship Between Alcohol and Drug Use and the Number of Protective Factors Reported by Washington Students	.79
Figure 25:	Geographic Regions	.82
Figure 26:	Distribution of Students in the State Sample by Rural/Urban Characteristics of School	.84
Figure 27:	Survey Respondents' Race and Ethnicity	.85
Figure 28:	Number of Children at Home	.87
Figure 29:	Hours per Week Worked at a Part-Time Job	.88

Table 1:	Number of Schools and Students in Statewide Sample by Grade	6
Table 2:	Grade 6 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1988–2000	17
Table 3:	Grade 8 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1988–2000	18
Table 4:	Grade 10 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1988–2000	19
Table 5:	Grade 12 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1990–2000	20
Table 6:	Average Age of First Use	22
Table 7:	Grade 6 Prevalence of Alcohol, Tobacco, and Other Drug Use in the Past 30 Days: 1990–2000	25
Table 8:	Grade 8 Prevalence of Alcohol, Tobacco, and Other Drug Use in the Past 30 Days: 1990–2000	26
Table 9:	Grade 10 Prevalence of Alcohol, Tobacco, and Other Drug Use in the Past 30 Days: 1990–2000	27
Table 10:	Grade 12 Prevalence of Alcohol, Tobacco, and Other Drug Use in the Past 30 Days: 1990–2000	28
Table 11:	Methods of Obtaining Alcohol in Grades 6, 8, 10, and 12	33
Table 12:	Perceived Safety at School	59
Table 13:	Profile of Community Risk and Protective Factors	68
Table 14:	Correlation of Community Risk and Protective Factors With Health Behavior Scales	70
Table 15:	Profile of School Risk and Protective Factors	72

Table 16:	Correlations of School Risk and Protective Factors With Health Behavior Scales
Table 17:	Profile of Peer-Individual Risk and Protective Factors74
Table 18:	Correlations of Peer-Individual Risk and Protective Factors With Health Behavior Scales
Table 19:	Actual and Weighted Proportions of Survey Participants by Geographic Region
Table B1:	Prevalence of 30-Day Cigarette Use by Ethnic Group
Table B2:	Prevalence of 30-Day Alcohol Use by Ethnic Group
Table B3:	Prevalence of 30-Day Marijuana Use by Ethnic Group
Table B4:	Prevalence of Two Week Binge Drinking Use by Ethnic Group140
Table B5:	Prevalence of 30-Day Weapon Carrying by Ethnic Group141
Table B6:	Prevalence of 12-Month Depressive Feelings by Ethnic Group

This report is part of a series of reports produced by RMC Research Corporation that summarize a major survey effort in the state of Washington. The planning and implementation of the 2000 administration of the Washington State Survey of Adolescent Health Behaviors (WSSAHB) were the products of an immense collaborative effort among the authors, the Washington State Survey Policy Committee, local educators and health professionals, and community members throughout the state of Washington. The summary reports are a collaborative product of the authors and the Survey Policy Committee.

RMC Research thanks the students, school administrators, parents, and local prevention and health professionals who encouraged and supported school participation in the survey. The survey would be of little use or consequence if these individuals had not shown interest in addressing the health behaviors and risk and protective factors at issue in the survey.

The survey planning and implementation involved many professionals from agencies and disciplines across the state. The following staff were, however, most consistently involved: Denise Fitch and Martin Mueller at the Office of Superintendent of Public Instruction, Steve Smothers and Linda Becker at the Department of Social and Health Services, Susan Roberts at the Office of Community Development, and Julia Dilley at the Department of Health. Thomas Taggart and Li Yang at the University of Washington's Office of Educational Assessment and Michael Arthur and John Briney of the University of Washington's Social Development Research Group also helped with the survey effort. Although all of these experts were active and influential in the survey methods and procedures, the authors bear responsibility for the content of this report and the companion reports. This survey benefited from the experience and guidance of Roy Gabriel, and this report benefited from the editing of Karla Wadeson and the word processing of Michelle Hutchens, all three of RMC Research Corporation.

The 2000 Washington State Survey of Adolescent Health Behaviors (WSSAHB) is the sixth statewide survey assessing the health-related attitudes and behaviors of Washington's public school students. A total of 17,870 students in 98 elementary, middle, and high schools across the state participated in the sample, which was designed to represent all Grade 6, 8, 10, and 12 students across the state. Of those schools asked to participate in the survey, 63 percent with Grade 6 students, 73 percent with Grade 8 students, and 62 percent with Grade 10 and Grade 12 students took part in the survey. As necessary to complete each cell of the sampling design, 13 nonsampled schools that participated on a volunteer basis were included with the sampled schools. The characteristics of the sample are sufficiently representative of Washington's students, which makes generalizing the survey results to the statewide population at the four grade levels reasonable.

Notably, another 84,662 students in 472 schools participated in the survey on a volunteer basis to obtain school-specific results to use in planning and evaluating prevention and intervention programs. The results of these surveys are not included in this report because they were not selected for the statewide scientific sample of students in public schools. More than twice as many students, and nearly twice as many schools, participated in the 2000 administration of the WSSAHB as compared to 1998, and that administration included more than twice as many students and schools as the 1995 administration. This continued increase in participation may reflect increasing interest in health-related information and is a tribute to the effective collaboration among the sponsoring agencies and local community members.

The survey was a cooperative effort of the Office of Superintendent of Public Instruction (OSPI); the Washington State Department of Social and Health Services' Division of Alcohol and Substance Abuse (DASA) and Research and Data Analysis (RDA); the Office of Community Development (OCD); the Department of Health (DOH); and the contractor, RMC Research Corporation.

Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use

As in all of the previous state surveys, alcohol was reportedly the substance of choice among Washington's students. Smoking tobacco and marijuana followed. The lifetime prevalence of alcohol, smoking tobacco, smokeless tobacco, and marijuana use appeared to drop considerably from 1998 to 2000. The lifetime prevalence for these substances was, however, assessed differently in 2000 than in previous survey administrations. Readers are therefore cautioned about interpreting changes over time for these substances.

The lifetime prevalence of psychedelic drug use appears to have decreased from 1998 to 2000, but the term *psychedelic* was used in 2000, whereas the term *hallucinogen* had been used in previous administrations. Thus this apparent decrease must be interpreted with caution. The lifetime prevalence of inhalant use declined from 1998 to 2000 to its lowest point since 1988 (2.5 percent of Grade 6 students, 9.6 percent of Grade 8 students, and 11.5 percent of Grade 10 students). The lifetime prevalence of amphetamine use declined for students in Grade 8 (4.3 percent), Grade 10 (8.4 percent), and Grade 12 (10.0 percent). The lifetime prevalence of methamphetamine use declined for students in Grade 10 (5.3 percent) and Grade 12 (7.5 percent).

The 2000 survey was the first administration that asked students about the use of party drugs (e.g., ecstasy). Almost none of the Grade 6 students (0.9 percent) had ever tried party drugs. This percentage increased among older students: 4.8

percent of Grade 8 students, 9.3 percent of Grade 10 students, and 13.5 percent of Grade 12 students had tried party drugs at some time in their lives.

The younger the age of drinking onset, the greater the chance that an individual will develop a clinically defined alcohol disorder at some point in life. The average ages of the first use of alcohol, tobacco, and marijuana were virtually identical to those reported in 1998, indicating that the average age of first use has not changed over the past two years. Among Grade 12 students who had ever used the substance, the average age of first use was 13.9 years for alcohol, 13.0 years for tobacco, and 14.3 years for marijuana.

Current Use of Alcohol

Alcohol has been reported as the substance of choice (i.e., the most frequently used substance) among Washington's students in each of the six administrations of the WSSAHB. Alcohol use in the past 30 days was reported by 6.6 percent of Grade 6 students, 22.8 percent of Grade 8 students, 37.6 percent of Grade 10 students, and 46.8 percent of Grade 12 students. Because the survey question regarding current use of alcohol changed in 2000, the results are not comparable to those from previous survey administrations.

After an essentially steady increase from 1992 to 1998, binge drinking decreased among students in Grades 6, 8, and 10 and leveled off among Grade 12 students. The rates of binge drinking remained, however, alarmingly high: 4.7 percent of Grade 6 students, 14.9 percent of Grade 8 students, 23.2 percent of Grade 10 students, and 31.8 percent of Grade 12 students reported binge drinking in the past two weeks.

Tobacco

Cigarette use in the past 30 days among Grade 6 students remained essentially unchanged from 1990 to 2000 (4.0 percent in 2000). Among Grade 8 students

cigarette use in the past 30 days increased from 12.1 percent in 1990 to 18.8 percent in 1995 and then decreased to 12.5 percent in 2000. Among Grade 10 students cigarette use climbed from 1990 (15.1 percent) to 1995 (20.9 percent) and has remained steady since then (19.8 percent in 2000). Among Grade 12 students cigarette use climbed from 1990 (20.7 percent) to 1998 (28.6 percent), and leveled in 2000 (27.6 percent).

Smokeless tobacco use in the past 30 days decreased from 1998 to 2000 among students in Grades 8, 10, and 12. Smokeless tobacco use decreased from 6.7 to 2.1 percent among Grade 8 students, from 9.6 to 4.6 percent among Grade 10 students, and from 12.4 to 8.8 percent among Grade 12 students.

Students were asked whether they think that smoke from other people's cigarettes (secondhand smoke) is harmful. More than 80 percent of students in all grades thought that secondhand was *probably* or *definitely* harmful. Students were also asked whether during the past year they had practiced in their classes ways to say *no* to tobacco—in role plays, for example. About two-thirds (69.7 percent) of Grade 6 students indicated that they had practiced in class ways to say *no* to tobacco during the past year. This percentage, however, decreased at each grade level, and only about one in seven (14.8 percent) Grade 12 students had practiced tobacco refusal in class.

Students were asked how often during the past 30 days they had seen antismoking commercials on television or heard them on the radio. About twothirds (67.8 percent) of Grade 6 students reported having seen or heard antismoking ads at least once a week during the past 30 days. About three fourths of the students in the higher grades indicated having seen or heard antismoking ads at least once a week during the past 30 days (77.1 percent of Grade 8 students, 77.7 percent of Grade 10 students, and 73.2 percent of Grade 12 students).

Marijuana

Marijuana use is of concern given its prevalence in adolescent treatment. Nationally, in 1998, 57 percent of youth aged 12 to 17 admitted to treatment reported marijuana as the primary substance of abuse. Thus it is critical to monitor current adolescent use of marijuana.

Marijuana use in the past 30 days decreased from 1998 to 2000 from 16.5 to 12.0 percent among Grade 8 students, from 26.6 to 21.9 percent among Grade 10 students, and from 28.7 to 24.4 percent among Grade 12 students. These declines represent a substantial decrease in the number of marijuana users. For example, given the October 2000 enrollment of 68,580 Grade 12 students, this decline represents a decrease of over 2,900 marijuana users in Grade 12.

Attitudes Toward Alcohol, Tobacco, and Other Drug Use

Among Grade 8 students, from 1988 to 1992 an increase in perceived risk was associated with decreased prevalence of binge drinking. From 1992 to 1995 decreased perception of risk was associated with increased prevalence of binge drinking. More recently, from 1995 to 2000, a continued increase in perceived risk was associated with a leveling and then decreasing prevalence of binge drinking. A similar pattern was observed for marijuana use.

Intentional Injury Behaviors: Fighting and Weapon Carrying

The WSSAHB contained a question that asked students how many times in the past 30 days they had carried a weapon, such as a gun, knife, or club, for self-protection or because they thought they might need it in a fight. Grade 6 students showed a steady decline in weapon carrying from 1992 to 1998, but were not asked this question in 2000. Grade 8 students showed a decrease in weapon carrying from 1992 to 1995 and then again from 1998 to 2000. Grade 10 and Grade 12 students reported decreased weapon carrying from 1992 to 1995,

but the prevalence of weapon carrying among these students has been level since then (although the results for Grade 10 students suggest a small continued decline). In 2000 about one in ten Grade 8, 10, and 12 students reported weapon carrying in the past 30 days.

The survey asked students how many times in the past year they had attacked someone with the idea of seriously hurting that person. This behavior decreased from 1998 to 2000 among Grade 8, 10, and 12 students (Grade 6 students were not asked this question). One in eight (12.4 percent) Grade 8 students and about one in ten Grade 10 (10.8 percent) and Grade 12 (9.1 percent) students reported in 2000 that they had attacked someone with the idea of seriously hurting them during the past year.

The WSSAHB contained one question related to depression in 2000. Nearly one in four (23.1 percent) of the Grade 6 students reported having experienced depressive feelings during the past year. Among Grade 8, 10, and 12 students, about one in three reported having experienced depressive feelings during the past year.

School Climate

Students were asked how safe they feel in several areas in school, including the classroom, the halls or stairs, the bathroom, the locker rooms, the playground or school grounds, the lunchroom, on the bus, on the way to school, and on the way home from school. Most students reported feeling generally safe in school. In particular, students reported that the classroom is a safe area (Grade 8 students were most likely to report feeling a little or very unsafe in the classroom; 8.9 percent of Grade 8 students reported this feeling). Students were more likely to report feeling and on the playground or school grounds. In general, younger students were less likely than older students to feel safe at school. For example, one in four Grade 6 students (24.8 percent)

reported feeling a little or very unsafe on the playground or school grounds, compared to 23.5 percent of Grade 8 students, 18.7 percent of Grade 10 students, and 12.8 percent of Grade 12 students.

Many students reported experiencing bullying at school, although most did not. Among Grade 6 students, one in five (20.9 percent) reported that other students at school put them down verbally a lot or every day. Younger students were more likely than older students to report experiencing bullying at school. Students were asked when they last carried a gun; knife or razor; or club, stick, pipe, or other weapon on school property for self-protection or because they thought they might need it in a fight. The percentage of students who reported this behavior in the past month was quite low, although the behavior does occur. One percent or less of the students at each grade level reported carrying a gun to school during the past month.

Risk and Protective Factors for Adolescent Health Behaviors

Decades of research have shown that *risk factors* are associated with increased likelihood of health risk behaviors, including alcohol, tobacco, and other drug abuse and violence and delinquent behaviors. Similarly, *protective factors* exert a positive influence or buffer against the negative influence of risk. This survey included items that assessed risk and protective factors as the state has an interest in this model of prevention.

Within the community domain of risk and protective factors, as students get older they are at considerably increased risk on perceived availability of drugs risk factor. Grade 6 students are more likely than older students to be resilient on the rewards for prosocial involvement risk factor. The strongest correlations between community risk factors and health risk behaviors involve the perceived availability of drugs and laws and norms favorable toward drug use. The correlations themselves, and their relative strengths, were generally similar in 1998 and 2000.

In the school domain, the percentage of students who were at risk on the little commitment to school risk factor increased for older students, from 35.2 percent of Grade 6 students to 47.3 percent of Grade 12 students. In addition, the percentage of students who resilient on the rewards for conventional involvement factor dropped from 60.1 percent of Grade 6 students to 45.0 percent of Grade 12 students. In the school domain there were moderate correlations between the risk factor of little commitment to school and alcohol use.

In the peer-individual domain, students report being at risk on the early initiation of drug use, early initiation of problem behavior, favorable attitudes toward drug use, perceived risk of drug use, and friends' use of drugs risk factors. The strongest peer-individual correlates with health risk behaviors were early initiation of drug use, attitudes favorable toward drug use, and friends' use of drugs. Consistent with previous administrations of the WSSAHB, protective factors showed their strongest relationships with health risk behavior in the peerindividual domain. An internal belief in the moral order and positive social skills were associated with lower levels of alcohol use, drug use, delinquent behavior, and violent behavior.

A clear association between the number of risk factors present and the prevalence of lifetime and 30-day alcohol and other drug use was evident. In addition, levels of protection (i.e., the presence of several protective factors in students) were clearly associated with lower rates of alcohol and other drug use. People across the nation are actively interested in their health and the healthy development of their children. They are concerned about the dangers of alcohol, tobacco, and other drug (ATOD) use. In contrast to the general public's health-related concern, adolescence is often filled with experimentation, risk-taking, and influences not always in the interest of good health. Too often youth suffer the consequences of violence, substance use, and other related risk behaviors. Professionals, parents, and policymakers need to actively promote responsible behavior and lifestyles among adolescents, recognizing the interrelationships of the full range of health risk behaviors. Although focusing on any one behavior in isolation may lead to short-term success, such narrow efforts will almost inevitably fall short of promoting long-term improvements and healthy lifestyles.

The Washington State Survey of Adolescent Health Behaviors (WSSAHB) is an effort to recognize the interdependencies of alcohol and other drug use, violence, and related risk and protective factors. The survey estimates the prevalence of major adolescent health risk behaviors and provides crucial information to school officials, health professionals, human service agencies, policymakers, and parents as they work together to ensure the optimum health of the young people across the state. The results presented in this report estimate the current status of these health risk behaviors, relate them to characteristics of the students engaging in them, and examine trends in the behaviors over the past 12 years.

The survey results also provide important needs assessment data for program planning and offer a global look at the effectiveness of statewide prevention and health promotion initiatives based on a range of education and healthrelated goals at the federal and state levels. One of the National Education Goals states that by the year 2000 every school in the United States will be free of drugs, violence, and the unauthorized presence of firearms and alcohol and will offer a disciplined environment conducive to learning. The current President's statement on education, No Child Left Behind (Bush, 2001), addresses the importance of school safety. Goal 1 of the 1999 National Drug Control Strategy (Office of National Drug Control Policy, 1999) is "Educate and enable America's youth to reject illicit drugs as well as alcohol and tobacco," Goal 2 is "Increase the safety of America's citizens by substantially reducing drug related crime and violence," and Goal 3 is "Reduce health and social costs to the public of illegal drug use by reducing the treatment gap." In 1998 the U.S. Department of Education, Safe and Drug-Free Schools and Communities Program issued Safe and Drug-Free Schools and Communities Act (SDFSCA) Principles of Effectiveness (U.S. Department of Education, 1998). The first principle states that each SDFSCA program shall "base its programs on a thorough assessment of objective data about the drug and violence problems in the schools and communities served."

The Washington Education Reform Act of 1993 established common learning goals intended to raise academic standards and student achievement for all students. The Commission on Student Learning subsequently approved the essential academic learning requirements (EALRs) that define the specific academic skills and knowledge the state's students will be required to meet. The EALRs in health and fitness (Washington State Commission on Student Learning, 1998) "establish the concepts and skills necessary for safe and healthy living and, in turn, for successful learning." One of the EALRs states that students will "identify physical, emotional, and legal consequences of using nicotine, alcohol, and other drugs and apply skills to resist any harmful use of substances." The Washington State Board of Health (1998) recommended eight Priority Health Goals for 1999–2001. Two of these priority goals relate to alcohol, tobacco, and

other drug use and one relates to violence: reduce tobacco use and exposure to secondhand smoke, reduce misuse of alcohol and other drugs, and reduce the incidence and impact of violence and preventable injuries. The Washington State Governor's Council on Substance Abuse has developed long-term goals to reduce substance abuse in the state. The Council's goals in the area of prevention include "prevent and reduce the misuse of alcohol, tobacco, and other drugs," "focus on outcome based prevention strategies to increase the effectiveness of prevention efforts," and "increase community ownership and responsibility for the prevention and misuse of alcohol, tobacco, and other drugs."

The results of the WSSAHB are intended to meet a wide variety of information needs by producing:

- Empirical needs assessment data necessary for planning prevention and early intervention programs.
- Data for studying trends of student substance use and abuse and associated risk and protective factors.
- Information on the progress of drug education programs funded under the federal Safe and Drug-Free Schools and Communities Act and the state Omnibus Controlled Substance and Alcohol Abuse Act.
- Data to measure the progress towards attainment of the state's targeted benchmarks for substance abuse prevention established by the Governor's Substance Abuse Prevention Advisory Committee.
- Information on the progress of programs implemented pursuant to the state's Youth Violence Act, E2SHB 2319.
- Data for the state's comprehensive, cross-agency database on youth violence developed by the Departments of Health and Social and Health Services.

- Data that can contribute information to local community profiles.
- Data to describe risk and protective factors that can be used by local school and community members as they plan or refine school- and community-based prevention and intervention programs.

The 2000 WSSAHB represents a cooperative effort among the Office of Superintendent of Public Instruction (OSPI); the Washington State Department of Social and Health Services' Division of Alcohol and Substance Abuse (DASA) and Research and Data Analysis (RDA); the Office of Community Development (OCD); the Department of Health (DOH); and the contractor, RMC Research Corporation. These agencies composed the Washington State Survey Policy Committee, which advised every aspect of the survey development and implementation. In addition, staff from the University of Washington's Social Development Research Group provided consultation on the risk and protective factors assessment portion of the survey.

This administration is the sixth biennial survey of Washington's students in Grades 6, 8, 10, and 12. The first two administrations (Deck and Nickel, 1989; Gabriel, 1991) included questions only about alcohol, tobacco, and other drug use and associated risk and protective factors. The 1992 and 1995 surveys included coverage of a variety of other health risk behaviors (Einspruch and Pollard, 1993; Gabriel, Deck, Einspruch, and Nickel, 1995). The 1998 survey once again focused on alcohol, tobacco, and other drug use and related risk and protective factors (Einspruch, Gabriel, Deck, and Nickel, 1998).

Participation

A random sample of schools, stratified by geographic region, school size, and percentage minority enrollment, was selected at each grade level to constitute a representative sample of Washington's Grade 6, 8, 10, and 12 students (schools were also ordered by community type within sampling cells to ensure inclusion along that dimension). When a given school declined to participate, due to reasons such as conflicts with other student assessments or objections to the survey content or length, another school of the same geographic region, size, percent minority enrollment, and community type was added to take its place. Of those schools asked to participate in the survey, 63 percent with Grade 6 students, 73 percent with Grade 8 students, and 62 percent with Grade 10 and Grade 12 students took part in the survey. (As necessary to complete each cell of the sampling design, 13 nonsampled schools that participated on a volunteer basis were included with the sampled schools: 4 schools for Grade 6, 1 school for Grade 8, and a total of 8 schools for Grades 10 and 12.) The data from the participating schools were adjusted using a statistical weighting procedure that realigned the proportionality of responses to reflect the actual statewide enrollment. The resultant weighted responses provide accurate statewide estimates of Washington students' health-related behaviors. Readers interested in a detailed account of the technical methodology of the survey are encouraged to review the forthcoming companion Technical Report prepared by RMC Research Corporation.

RMC Research's analysis of the survey information included a series of quality control steps to remove data that were incomplete, obviously inaccurate, or internally inconsistent. Table 1 shows the resultant sample sizes—that is, the number of participating schools and students—at each of the four grade levels. The precision of the estimates presented in this report is a direct result of the size and design of the sample. The results presented in this report are not perfect estimates—rather, there exists a certain margin of error for interpretation. When readers examine the results presented here and make comparisons between 1998 and 2000 results; they must consider this notion of margin of error. The sample sizes in Table 1 indicate that for statewide comparisons at each grade, the prevalence rates in this report can be interpreted within plus or minus 3.0 percent at Grades 6 and 8 and within plus or minus 3.5 percent at Grades 10 and 12. These margins of error allow 95 percent confidence in the interpretation of the results.

Grade Level	Number of Schools	Number of Students
6	37	4,312
8	32	4,980
10	33	4,820
12	33	3,758
Total	98	17,870

Table 1:Number of Schools and Students in Statewide Sample by Grade

Note. More than one grade may have been surveyed in a given school; thus the total number of schools is less than the sum across grades. Numbers represent percent of students responding.

In addition to the number of schools and students included in the statewide sample, 84,662 students in 472 schools participated in the survey on a volunteer basis. These schools receive reports of their own results, but their results are not included in this statewide report because these schools were not part of the representative statewide sample. More than twice as many students, and nearly twice as many schools, participated in the 2000 administration of the WSSAHB as compared to 1998, and that administration included more than twice as many students and schools as the 1995 administration. This continued increase in participation may reflect increasing interest across the state in health-related information and is a tribute to the collaboration among the sponsoring agencies and local community members.

Within the sample on which the results presented in this report are based, some key student characteristics include these:

 Nearly equal percentages of males and females at each grade, closely matching the percentages in the population statewide.

- Close representation of students in urban, suburban, and rural schools for Grade 8 students compared to the population distribution, although suburban Grade 6 students were somewhat overrepresented and rural high school students were somewhat overrepresented.
- Close representation of students by ethnic group to the population distribution, although White, non-Hispanic students were a little underrepresented at Grade 6 and Grade 8.

Purpose of This Report

This report provides the results of the 2000 administration of the WSSAHB. Beyond this introduction, the Analytic Report contains 6 additional chapters that address the adolescent health behaviors of Washington's students. Chapter 2 presents the prevalence of alcohol, tobacco, and other drug use. These questions, used in all previous statewide surveys dating back to 1988, are derived primarily from the National Institute on Drug Abuse (NIDA) Monitoring the Future (Johnston, O'Malley and Bachman, 1994; National Institute on Drug Abuse, 2001) survey, the Centers for Disease Control and Prevention's Youth Risk Behavior Survey (YRBS; Centers for Disease Control, 1999), and the Youth Tobacco Survey (YTS, Centers for Disease Control, 2000). Chapter 3 discusses violence, weapon carrying, and other delinquent behaviors. These questions are derived largely from the Centers for Disease Control and Prevention's Youth Risk Behavior Survey. Chapter 4 presents the results from questions regarding school climate. These questions were included for the first time on the 2000 survey. Chapter 5 presents the results of the assessment of risk and protective factors. These questions are derived from a risk and protective factors assessment instrument developed by the Social Development Research Group at the University of Washington (Arthur, Hawkins, Catalano, and Pollard, 1998). Chapter 6 presents background and school- and community-related characteristics of the sample

of students to better describe the nature of this sample as it mirrors statewide enrollment at the grade levels surveyed. Chapter 7 concludes this report.

In reporting the results of this survey, the authors provide two comparative frames of reference. First, trends over time—comparisons with results of previous surveys—are presented. These comparisons allow readers to view the trends over past years' reports of health risk behaviors among Washington's students at the same grade levels. Second, using Healthy People 2010 (U.S. Department of Health and Human Services, 2000a, 2000b) as a starting point, the state has established a specific set of objectives for substance abuse prevention, many of which are measured with student survey data. Where available, the targets for those objectives can be compared to the results of the current survey.

The results presented in this report are the major findings from the WSSAHB survey. Many more possibilities for comparative analysis of the data resulting from this large, statewide effort exist. Appendix A features response frequencies to each of the items contained in the survey. Appendix B features response frequencies by ethnic group to selected items contained in the survey. Readers are referred to the appendixes for more detailed information. In addition, other reports of these results will be forthcoming from the state agencies and RMC Research Corporation.

Caution

Readers should bear in mind several cautions regarding the survey data contained in this report.

Representativeness

Every attempt was made to ensure the sample's representativeness of the students in Washington public schools. Although the response rate to the survey was good (62 to 73 percent of recruited schools participated), the possibility

exists that the results are not representative of the student population of the state as a whole. The replacement schools were selected to be similar to the refusing schools, although the use of randomly selected replacement schools is common and accepted survey practice. Despite this effort, the participating and refusing schools might differ on characteristics that might be related to responses to the survey. The sample included representation from across the state, including the Seattle and Tacoma School Districts. Thus generalizing the results to all students at these grade levels in Washington public schools may be reasonable.

Trends

In comparing the results of the 2000 survey and earlier surveys, readers should remember that certain factors may influence apparent trends. For example, information about the characteristics of the 1988 and 1990 samples is not readily available. Comparisons with the 1992 surveys might be influenced by the inclusion of volunteer schools in the data, although comparisons between the sampled and volunteer schools that year revealed similar levels of substance use. Also, the trends for substance use in Washington from 1988 to 2000 appear to be similar to national trends.

School Dropouts

In interpreting differences between grade levels, readers should remember that some reported behaviors and risk factors may appear more prevalent in Grade 8 and Grade 10 compared to Grade 12 because of increased school dropouts after age 16 (i.e., prior to Grade 12). It is generally accepted that the results for high school seniors in surveys such as this one are underestimates of young people of that age group because many of the students most likely to engage in these kinds of behaviors may have dropped out of school (Johnston et al., 1994). Thus the authors recommend interpreting results for high school seniors with some caution, particularly when their prevalence rates differ markedly from those of students at earlier grades.

The school dropout issue is not new and has also existed in previous Washington State surveys. Unless data to suggest that the current high school population includes a larger or behaviorally different collection of dropouts than in previous years, the bias in Grade 12 estimates is likely similar to what it has been in the past. This fact means that although any given year's data on health risk behaviors among Grade 12 students may be an underestimate, the year-toyear comparisons are likely to be less affected by this bias (Johnston et al., 1994).

Developmental Changes

In interpreting differences between grade levels, readers should remember that developmental changes may influence students' perceptions and accuracy of reporting.

Self-Report Data

The survey measures self-reports, which may be influenced by factors including problems in remembering, social desirability, reading ability, and developmental changes.

Correlational Data

Interrelationships among the variables should not be interpreted as indicating that one variable caused the other. Although this causal relationship might be the case, the reverse might also be true or an apparent relationship might be due to some other measured or unmeasured cause.

Extrapolations

Estimates of the number of students who engage in a particular behavior should be regarded only as estimates because of the issues already described and other issues such as the noninclusion of students who are not attending public schools (including students in private schools, students receiving home schooling, students in correctional facilities, school dropouts, and others).
Introduction

Alcohol, tobacco, and other drug use among young people has both shortand long-term implications. In the short term, alcohol, tobacco, and other drug use interferes with positive, healthy physical, emotional, and social development. Relationships within families and among friends and satisfactory progress in school are all casualties of substance use. In the long term, alcohol, tobacco, and other drug use is associated with delinquency and criminal activity, unintended injuries, and a variety of severe health complications, including shorter life expectancy (Centers for Disease Control and Prevention, 1994). For example, of the 11.1 million victims of violent crime, almost one in four (2.7 million) report that the offender had been drinking alcohol prior to committing the crime (Greenfeld, 1998). Of the more than 2 million deaths each year in the United States, approximately one in four is attributable to alcohol, tobacco, and illicit drug use (tobacco causes about 430,700 deaths, alcohol causes about 100,000 deaths, and illicit drugs cause about 16,000 deaths)—thus substance abuse is the single largest preventable cause of death in this country (Brandeis University, 2001).

Schools experience the consequences of alcohol, tobacco, and other drug use in a variety of ways, some quite dramatic. The Center on Addiction and Substance Abuse (1997) reported that a national survey of teachers, principals, teens, and parents revealed that 29 percent of the high school students surveyed had indicated that a student in their school had died from an alcoholor drug-related incident in the past year. The survey also illustrated dramatic differences between students' perceptions of drugs in schools compared to those of teachers and principals (the students thought that drugs were commonplace in their schools, whereas teachers and principals were often unaware of their prevalence).

The economic costs of alcohol, tobacco, and other drug abuse are enormous. Nationwide, the economic cost of alcohol and drug abuse was \$276 billion in 1995, in terms of health care, motor vehicle crashes, crime, lost productivity, and other adverse outcomes of alcohol and other drug abuse (Harwood, Fountain, and Livermore, 1998). Wickizer, Wagner, Atherly, and Beck (1993) studied the economic costs of alcohol and other drugs to Washington State in 1990. The authors estimated that the economic losses in Washington due to alcohol and other drugs was \$1.81 billion, or \$372 for every man, woman, and child living in the state. The authors also found that alcohol abuse—not other drug abuse—had the greatest economic impact and that for every dollar the state collected in tax revenue from alcohol sales, over seven dollars were spent as a result of alcohol abuse. In 1999 Wickizer updated this figure, estimating that the economic cost of alcohol, tobacco, and other drug use had been \$2.54 billion in Washington State in 1996. This figure represents a 39 percent increase over the 1990 cost estimate.

Alcohol use, tobacco use, and other drug use are preventable behaviors. Current research on alcohol, including research into its effects on the brain, genetic and psychosocial influences, medical consequences, prevention, and treatment can be found in the 10th Special Report to the U.S. Congress on Alcohol and Health (National Institute on Alcohol Abuse and Alcoholism, 2000). The national health objectives for the year 2010 include reductions in alcohol and other drug use as a high priority (U.S. Department of Health and Human Services, 2000a). Although schools can play an important role in substance abuse prevention, they cannot do so without the support and effort of the communities in which they exist. The Healthy People 2010 objective that supports this idea is "Increase the number of communities using partnerships or coalition models to conduct comprehensive substance abuse prevention efforts" (U.S. Department of Health and Human Services, 2000a, 2000b).

Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use

Lifetime prevalence for most substances was assessed differently in 2000 than in previous survey administrations. In 1998 and earlier administrations, students were asked "Have you ever, even once in your lifetime, used any of the following drugs?" In the 2000 administration students were asked this same question in reference to many substances, although for other substances lifetime prevalence was determined from the students' responses to the question "How old were you when you first . . . ?" followed by a list of drugs and behavioral descriptors (rather than the names of the drugs only—for example, "had more than a sip or two of beer, wine, or hard liquor" rather than "drank alcohol"). Although the new approach appears comparable to earlier approaches, the changes in the results from 1998 to 2000 are significant enough that that they may be due to measurement rather than behavioral changes. Readers are therefore cautioned against comparing changes over time for these substances.

As in all previous WSSAHB surveys (Deck and Nickel, 1989; Einspruch and Pollard, 1993, Einspruch et al., 1998, Gabriel et al., 1995; Gabriel, 1991), the respondents reported alcohol as the substance of choice. Smoking tobacco and marijuana followed. These three drugs are often termed *gateway substances*, which refers to initial experimentation with illegal drugs that often leads to more frequent use of these and other substances. Although the majority of people who use marijuana do not go on to use more addictive substances, studies on the influence of gateway drugs indicate that young people who use marijuana are 85 times more likely to use cocaine than those who have never used marijuana (Center on Addiction and Substance Abuse, 1994).

Tables 2 through 5 detail lifetime prevalence for students in Grades 6, 8, 10, and 12 from 1988 through 2000. Lifetime prevalence of inhalant use declined from 1998 to 2000 to its lowest point since 1988 (2.5 percent for Grade 6 students, 9.6 percent for Grade 8 students, and 11.5 percent for Grade 10 students). Lifetime prevalence of amphetamine use declined for students in Grade 8 (4.3 percent), Grade 10 (8.4 percent), and Grade 12 (10.0 percent). Lifetime prevalence of methamphetamine use declined for students in Grade 10 (5.3 percent) and Grade 12 (7.5 percent). Lifetime prevalence of psychedelic drug use appears to have decreased substantially from 1998 to 2000, but the term *psychedelic* was used in 2000, whereas *hallucinogen* had been used in previous administrations. Thus this apparent decrease must be interpreted with caution.

		Change					
Substance	1988	1990	1992	1995	1998	2000	1998-2000
Alcohol	51.4	33.0	33.0	33.2	39.8	21.2ª	b
Tobacco, smoking	12.4	11.3	11.7	20.6	25.7	15.1ª	b
Tobacco, smokeless	9.5	5.4	5.5	7.1	7.8	1.8ª	b
Marijuana	3.6	1.7	1.9	4.9	7.0	2.2ª	b
Hallucinogens (Psychedelics)	1.5	.8	1.2	1.1	2.6	0.8	-1.8
Inhalants	13.0	7.5	7.7	3.9	7.0	2.5	-4.5
Over-the-counter drugs	-	7.0	7.8	2.0	-	-	b
Cocaine	0.8	0.9	1.1	1.3	2.3	-	b
Steroids	1.7	1.2	1.1	1.2	2.6	-	b
Other illegal drugs	-	_	1.4	1.6	_	2.4	0.8c
Heroin		_	-	_	1.7	-	b
Amphetamines	-	_	-	-	3.4	-	b
Methamphetamines	_	_	_	_	2.3	_	b
Party drugs	_	_	_	_	_	0.9	b

Table 2: Grade 6 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1988– 2000

Note. Dashes indicate a substance was not represented on that particular year's survey. Statistically significant changes are highlighted in bold.

^aThe presentation of the question changed for the 2000 administration.

^bNo data were available to compare change between 1998 and 2000 administrations.

^c Figure reflects a change from the 1995 administration year and the 2000 administration year. No data available

for the 1998 administration.

	Percent of Students Responding						Change
Substance	1988	1990	1992	1995	1998	2000	1998-2000
Alcohol	68.9	60.2	55.3	58.1	62.7	45.7ª	b
Tobacco, smoking	29.8	32.5	31.0	48.9	48.2	37.1ª	b
Tobacco, smokeless	16.6	13.9	13.1	22.9	14.8	5.2ª	b
Marijuana	14.4	11.2	9.0	27.2	28.2	1 9.7 ª	b
Hallucinogens (Psychedelics)	4.0	5.0	5.6	9.3	8.7	4.7	-4.0
Inhalants	17.3	17.1	17.4	14.5	14.3	9.6	-4.7
Over-the-counter drugs	-	13.8	11.1	11.6	-	-	b
Cocaine	2.0	3.0	2.0	5.5	5.2	_	-0.3c
Steroids	3.0	2.0	1.0	2.5	2.6	2.2	-0.4
Other illegal drugs	_	-	4.0	8.4	-	_	b
Heroin	_	-	-	_	2.6	1.4	-1.2
Amphetamines	-	-	-	-	8.4	4.3	-4.1
Methamphetamines	-	-	-	-	4.6	2.0	-2.6
Party drugs	_	_	_	_	_	4.8	b

Table 3:Grade 8 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1988–2000

Note. Dashes indicate a substance was not represented on that particular year's survey. Statistically significant changes are highlighted in bold.

^aThe presentation of the question changed for the 2000 administration.

^bNo data were available to compare change between 1998 and 2000 administrations.

^cFigure reflects a change from the 1995 administration year and the 1998 administration year. No data

available for the 2000 administration.

	Percent of Students Responding						Change
Substance	1988	1990	1992	1995	1998	2000	1998-2000
Alcohol	84.1	75.7	70.3	70.5	79.7	65.0ª	b
Tobacco, smoking	43.1	43.4	43.7	55.7	63.4	52.2ª	b
Tobacco, smokeless	21.5	22.1	23.2	30.7	25.8	14.3ª	b
Marijuana	32.7	21.5	22.8	39.1	49.5	37.6ª	b
Hallucinogens (Psychedelics)	12.1	9.1	11.1	15.4	18.8	10.7	-8.1
Inhalants	19.5	17.7	15.6	12.3	15.3	11.5	-3.8
Over-the-counter drugs	-	23.2	18.4	12.3	-	-	b
Cocaine	8.1	4.3	3.5	7.4	9.4	6.0	-3.4
Steroids	4.9	3.0	2.2	2.1	3.1	2.9	-0.2
Other illegal drugs	-	_	7.9	11.6	_	_	b
Heroin	-	-	-	-	3.9	1.9	-2.0
Amphetamines	-	-	-	-	14.6	8.4	-6.2
Methamphetamines	-	_	-	-	9.8	-	b
Party drugs	_	_	_	_	_	9.3	b

Table 4: Grade 10 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1988– 2000

Note. Dashes indicate a substance was not represented on that particular year's survey. Statistically significant changes are highlighted in bold.

^aThe presentation of the question changed for the 2000 administration.

^bNo data were available to compare change between 1998 and 2000 administrations.

Table 5: Grade 12 Lifetime Prevalence of Alcohol, Tobacco, and Other Drug Use: 1990– 2000

	Per	Percent of Students Responding					
Substance	1990	1992	1995	1998	2000	1998-2000	
Alcohol	83.0	79.8	82.1	84.2	76.0ª	b	
Tobacco, smoking	51.7	52.6	64.7	68.4	60.9 a	b	
Tobacco, smokeless	28.5	27.9	37.7	35.0	24.8 a	b	
Marijuana	34.0	32.9	43.5	55.1	50.5 ª	b	
Hallucinogens (Psychedelics)	13.7	16.8	18.7	23.8	15.1	-8.7	
Inhalants	16.4	13.1	11.0	13.3	13.1	-0.2	
Over-the-counter drugs	27.2	22.3	11.6	-	-	b	
Cocaine	7.8	4.6	7.6	9.7	9.2	-0.5	
Steroids	3.2	2.4	2.4	3.0	2.9	-0.1	
Other illegal drugs	_	9.5	11.1	-	-	b	
Heroin	_	-	_	3.6	2.4	-1.2	
Amphetamines	_	-	_	14.9	10.0	-4.9	
Methamphetamines	_	_	-	11.0	7.5	-3.5	
Party drugs	_	_	_	_	13.5	b	

Note. Dashes indicate a substance was not represented on that particular year's survey. In 1988 Grade 12 was not surveyed. Statistically significant changes are highlighted in bold.

^aThe presentation of the question changed for the 2000 administration.

^bNo data were available to compare change between 1998 and 2000 administrations.

The 2000 survey was the first administration that asked students about their use of party drugs (e.g., ecstasy). Almost none of the Grade 6 students (0.9 percent) had ever tried party drugs. This percentage increased among the older students: 4.8 percent of Grade 8 students, 9.3 percent of Grade 10 students, and 13.5 percent of Grade 12 students had tried party drugs at some time in their lives.

Students begin experimenting with alcohol and other drugs at an early age. The younger the age of drinking onset, the greater the chance that an individual will develop a clinically defined alcohol disorder at some point in life. For example, Grant and Dawson (1997) found that young people who began drinking before age 15 were four times more likely to develop alcohol dependence than those who began drinking at age 21. The state's substance abuse prevention target is "Increase the average age of first use of alcohol, tobacco, and marijuana to age 16."

Table 6 shows the average age of first use for those respondents who had ever tried a given substance the row labeled shows the average age of first use, the row labeled *n* shows the number on which this average was computed, and the row labeled *SD* shows the standard deviation). On average, Grade 6 students reported having more than a sip or two of beer, wine, or hard liquor at 10.5 years of age, whereas Grade 12 students reported being 13.9 years of age. Grade 6 students reported that they began drinking alcoholic beverages at least once or twice a month at 10.7 years of age, whereas Grade 12 students reported being 15.3 years of age. Grade 6 students were 10.3 years of age when they first smoked a cigarette (even just a puff), whereas Grade 12 students were 13.0 years of age. Grade 6 students were 14 years of age. These average ages are virtually identical to the averages reported in 1998, indicating that average age of first use has not changed over the past two years.

Factor		Grade 6	Grade 8	Grade 10	Grade 12
More than a sip of	X	10.5	11.5	12.8	13.9
beer, wine, or hard liquor	n	825	2,140	2,954	2,725
	SD	0.7	1.3	1.3	2.2
Began drinking at	X	10.7	12.1	13.9	15.3
least once or twice a	n	157	751	1,538	1,712
	SD	1.0	1.3	1.5	1.6
Smoked a cigarette,	X	10.3	11.1	12.0	13.0
even just a puff	n	590	1,750	2,412	2,190
	SD	0.6	1.2	1.7	2.1
Smoked marijuana	X	10.8	11.9	13.3	14.3
	n	88	932	1,740	1,819
	SD	1.1	1.2	1.6	1.8

Table 6: Average Age of First Use

Current Use of Alcohol, Tobacco, and Other Drugs

Although lifetime prevalence trends are of great concern, readers must remember that these tends reflect in part experimental use. Lifetime prevalence is the percentage of students who have ever tried a substance, even if only on one occasion. An indicator of more current use is represented by students' responses to questions about substance use in the past 30 days.

Tables 7 through 10 detail Grade 6, 8, 10, and 12 students' alcohol, tobacco, and other drug use in the past 30 days. Because the survey question regarding alcohol changed in 2000, the results are not comparable to those from previous survey administrations. The question regarding hallucinogens also changed: in 2000 the term *psychedelic* was used, whereas *hallucinogens* had been used in previous administrations. Alcohol is clearly the substance of choice among students, followed by cigarettes and marijuana. In addition, students reported increased prevalence as they get older. For example, alcohol use in the past 30 days was reported by 6.6 percent of Grade 6 students and by 46.8 percent of Grade 12 students. Cigarette use in the past 30 days was reported by 4.0 percent of Grade 6 students and 27.6 percent of Grade 12 students. 1.5 percent of Grade 6 students and 24.4 percent of Grade 12 students reported marijuana use in the past 30 days.

Smokeless tobacco and marijuana were the only substances that showed a change in 30-day prevalence. Smokeless tobacco use decreased from 6.7 to 2.1 percent among Grade 8 students, from 9.6 to 4.6 percent among Grade 10 students, and from 12.4 to 8.8 percent among Grade 12 students. Marijuana use decreased from 16.5 to 12.0 percent among Grade 8 students, from 26.6 to 21.9 percent among Grade 10 students, and from 28.7 to 24.4 percent among Grade 12 students.

The Governor's Substance Abuse Prevention Advisory Committee's objectives for current substance use include:

- Reduce the prevalence of alcohol use to 6 percent for Grade 6 students, 15 percent for Grade 8 students, 24 percent for Grade 10 students, and 34 percent for Grade 12 students.
- Reduce the prevalence of marijuana use to 1 percent for Grade 6 students, 8 percent for Grade 8 students, 10 percent for Grade 10 students, and 14 percent for Grade 12 students.
- Reduce the prevalence of cigarette use to 3 percent for Grade 6 students, 10 percent for Grade 8 students, 15 percent for Grade 10 students, and 34 percent for Grade 12 students.

 Reduce the prevalence of binge drinking in the past two weeks to 4 percent for Grade 6 students, 12 percent for Grade 8 students, 18 percent for Grade 10 students, and 20 percent for Grade 12 students.

	Per	Change				
Substance	1990	1992	1995	1998	2000	1998-2000
Alcohol	11.8	12.8	12.2	13.8	6.6ª	b
Tobacco, smoking	2.4	2.8	4.3	4.7	4.0	-0.7
Tobacco, smokeless	_	_	3.6	3.5	0.8	-2.7
Marijuana	1.3	1.3	3.1	3.4	1.5	-1.9
Hallucinogens (Psychedelics)	-	-	-	1.3	0.6	-0.7
Inhalants	_	_	2.7	3.2	1.4	-1.8
Cocaine	-	-	1.0	1.1	_	b
Other illegal drugs	-	1.4	1.3	-	1.0	-0.3c
Heroin	-	-	-	0.6	_	b
Amphetamines	-	_	-	1.4	_	b
Methamphetamines	_	_	-	0.9	-	b
Party drugs	_	_	_	_	0.7	b

Table 7:Grade 6 Prevalence of Alcohol, Tobacco, and Other Drug Usein the Past 30 Days: 1990–2000

Note. Dashes indicate a substance was not represented on that particular year's survey. Statistically significant changes are highlighted in bold.

^aThe presentation of the question changed for the 2000 administration. ^bNo data were available to compare change between 1998 and 2000 administrations.

^cFigure reflects a change from the 1995 administration year and the 2000 administration year. No data available for the 1998 administration.

Table 8:Grade 8 Prevalence of Alcohol, Tobacco, and Other Drug Use in the Past 30Days: 1990–2000

	Percent of Students Responding					
Substance	1990	1992	1995	1998	2000	1998-2000
Alcohol	29.1	24.0	30.1	31.0	22.3ª	b
Tobacco, smoking	12.1	10.3	18.8	15.2	12.5	-2.7
Tobacco, smokeless	_	_	11.5	6.7	2.1	-4.6
Marijuana	7.6	6.1	16.2	16.5	12.0	-4.5
Hallucinogens (Psychedelics)	-	-	-	3.8	3.1	-0.7
Inhalants	-	-	7.3	6.6	4.9	-1.7
Cocaine	3.1	2.0	3.6	2.5	1.5	-1.0
Other illegal drugs	5.4	5.0	6.9	-	_	b
Heroin	_	_	-	1.3	0.8	-0.5
Amphetamines	_	_	-	3.9	2.7	-1.2
Methamphetamines	-	-	-	2.3	1.2	-1.1
Party Drugs	_	_	-	-	3.4	b

Note. Dashes indicate a substance was not represented on that particular year's survey. Statistically significant changes are highlighted in bold.

^aThe presentation of the question changed for the 2000 administration. ^bNo data were available to compare change between 1998 and 2000 administrations.

Table 9:Grade 10 Prevalence of Alcohol, Tobacco, and Other Drug Use in the Past 30Days: 1990–2000

	Per	Change				
Substance	1990	1992	1995	1998	2000	1998-2000
Alcohol	44.0	40.0	37.0	44.9	37.6ª	b
Tobacco, smoking	15.5	17.1	20.9	21.8	19.8	-2.0
Tobacco, smokeless	_	-	15.3	9.6	4.6	-5.0
Marijuana	10.6	13.2	23.0	26.6	21.9	-4.7
Hallucinogens (Psychedelics)	-	-	-	5.8	5.8	0
Inhalants	_	-	5.4	3.9	3.6	-0.3
Cocaine	2.1	2.1	3.2	3.2	2.6	-0.6
Other illegal drugs	7.2	7.3	6.1	_	-	b
Heroin	_	-	-	1.3	1.0	-0.3
Amphetamines	_	-	_	5.6	4.5	-1.1
Methamphetamines	_	_	_	3.8	2.6	-1.2
Party Drugs	_	-	_	-	6.2	b

Note. Dashes indicate a substance was not represented on that particular year's survey. Statistically significant changes are highlighted in bold.

^aThe presentation of the question changed for the 2000 administration. ^bNo data were available to compare change between 1998 and 2000 administrations.

Table 10:Grade 12 Prevalence of Alcohol, Tobacco, and Other Drug Use in the Past 30Days: 1990–2000

	Per	Percent of Students Responding							
Substance	1990	1992	1995	1998	2000	1998-2000			
Alcohol	52.0	51.8	44.8	52.0	46.8ª	b			
Tobacco, smoking	20.7	22.3	24.0	28.6	27.6	-1.0			
Tobacco, smokeless	-	-	18.2	12.4	8.8	-3.6			
Marijuana	15.9	17.3	23.3	28.7	24.4	-4.3			
Hallucinogens (Psychedelics)	-	-	-	6.0	6.5	0.5			
Inhalants	-	-	2.7	2.3	2.4	0.1			
Cocaine	2.6	2.0	1.9	2.7	2.8	0.1			
Other illegal drugs	8.8	8.2	5.1	-	-	b			
Heroin	-	-	-	0.7	0.8	0.1			
Amphetamines	-	-	-	3.6	4.0	0.4			
Methamphetamines	-	-	-	2.9	2.9	0.0			
Party drugs	_	_	_	_	6.8	b			

Note. Dashes indicate a substance was not represented on that particular year's survey. Statistically significant changes are highlighted in bold. ^aThe presentation of the question changed for the 2000 administration. ^bNo data were available to compare change between 1998 and 2000 administrations.

The 2000 survey was the first administration in which students were asked about use of party drugs (e.g., ecstasy). Almost no Grade 6 students (0.7 percent) had used party drugs in the past 30 days, but this percentage increased among older students: 3.4 percent of Grade 8 students, 6.2 percent of Grade 10 students, and 6.8 percent of Grade 12 students had used party drugs in the past 30 days.

Alcohol

Alcohol has been reported as the substance of choice (i.e., the substance most frequently used) among Washington's students in each of the six administrations

of the WSSAHB. Figure 1 presents three standard indicators of alcohol use: lifetime prevalence, 30-day use, and binge drinking. This figure shows that rates of alcohol use increase by grade level. Among the Grade 6, students 21.2 percent had tried alcohol at some time in their lives, 6.6 percent reported alcohol use in the past 30 days, and 4.7 percent reported binge drinking (i.e., consuming five or more drinks in a row) during the past two weeks. These rates increase at each grade level. Seventy–six percent of the Grade 12 students had tried alcohol at some time in their lives, 46.8 percent reported alcohol use in the past 30 days, and 31.8 percent reported binge drinking during the past two weeks.



Figure 1: Alcohol Use: Grades 6, 8, 10, and 12

Changes in alcohol use over time can be traced from 1988 to 1998, but the year 2000 marks the beginning of a new trend line due to changes in the wording of the question regarding alcohol use in the past 30 days. Nationally, 22.4 percent of Grade 8 students, 41.0 percent of Grade 10 students, and 50.0 percent of Grade 12 students reported alcohol use in the past 30 days.

Figure 2 illustrates changes in binge drinking among Washington students over the past 12 years. After an essentially steady increase from 1992 to 1998, binge drinking decreased among students in Grades 6, 8, and 10 and leveled off among students in Grade 12. The rates of binge drinking remained, however, alarmingly high in 2000: 4.7 percent of Grade 6 students, 14.9 percent of Grade 8 students, 23.2 percent of Grade 10 students, and 31.8 percent of Grade 12 students reported binge drinking in the past two weeks. Nationally, 14.1 percent of Grade 8 students, 26.2 percent of Grade 10 students, and 32.3 percent of Grade 12 students reported binge drinking in the past two weeks.

Figure 2: Trend in Prevalence of Binge Drinking in Past Two Weeks in Grades 6, 8, 10, and 12



Patterns of alcohol use by gender varied by grade level. Among Grade 6 students, males were more likely than females to report alcohol use in the past 30 days (7.9 percent compared to 5.5 percent). Among Grade 8 students, females were more likely than males to report alcohol use in the past 30 days (24.2 percent compared to 20.0 percent). Among Grade 10 students, 30-day prevalence was similar for males (36.7 percent) and females (38.1 percent), but males were more likely than females to report having had 10 or more drinks during the past 30 days (7.8 percent compared to 4.7 percent). Among Grade 12 students, males were more likely than females to report alcohol use in the past 30 days (49.2 percent compared to 44.5 percent). Although statistically significant, these differences are not great enough to be of much practical significance.

Patterns of binge drinking by gender also varied by grade level. Among Grade 6 students, males were more likely than females to report binge drinking in the past two weeks (5.7 percent compared to 3.8 percent). Among Grade 8 students, males and females were equally likely to report binge drinking (14.2 percent compared to 15.4 percent). Among Grade 10 students, binge drinking rates were similar for males (23.4 percent) and females (22.7 percent), but males were more likely than females to report binge drinking 10 or more times during the past two weeks (3.0 percent compared to 1.5 percent). Among Grade 12 students, males were more likely than females to report binge drinking in the past two weeks (37.8 percent compared to 25.6 percent).

Patterns of alcohol use also varied by ethnic group. American Indian/Alaskan Natives reported the highest levels of alcohol use (59.1 percent of Grade 12 students), followed by Hispanics (52.9 percent of Grade 12 students), Black or African American non-Hispanics (47.3 percent of Grade 12 students), White non-Hispanics (47.0 percent of Grade 12 students), and Asian/Pacific Islanders (37.8 percent of Grade 12 students). Readers are, however, cautioned against overinterpreting these results due to the small numbers of survey respondents included in some ethnic groups. Detailed results by ethnic group are in Appendix B of this report.

The prevalence of the students' experimentation with alcohol and the quantity and frequency with which they consumed alcohol were combined to form a composite index of alcohol use for interpretive purposes. Combining the data from several items that ask about different aspects of alcohol use produces for school and health officials a more global assessment of the prevalence of varying levels of problematic alcohol use. (Specific details about the scaling of this composite index are available in the Technical Report.) Because of the addition of new items or changes in wording of previously used items, this scale is different from the one used by the authors for previous WSSAHB administrations. A consequence of this difference is that comparing the 2000 results on this composite scale of alcohol use with the results from previous years' composite scales of alcohol use is not possible.

The composite alcohol use scale consists of four levels: (a) *no use*—never having tried one full drink of any form of alcohol, (b) *prior use*—some experimentation with alcohol, but no use in the past 30 days, (c) *some recent use*—having had at least one full drink in the past 30 days, and (d) *frequent use*—having drunk alcohol ten or more times in the past 30 days or binge drinking three or more times in the past two weeks. Figure 3 shows the results from this composite scale, reiterating the finding that students use alcohol more as they get older. In particular, whereas only 5.6 percent of Grade 6 students reported recent use and 1.8 percent reported frequent use, among Grade 12 students 32.6 percent reported recent use and 15.8 percent reported frequent use.

Figure 3: Composite Scale: Alcohol Use in Grades 6, 8, 10, and 12



■No use ■Prior use □Recent use ■Frequent use

Students were asked how they usually obtain alcohol. Table 11 shows that about half of the Grade 6 students who reported drinking alcohol obtained their alcohol at home and their parents knew about it. In contrast, one-half to twothirds of Grades 8, 10, and 12 students who reported drinking obtained their alcohol from friends. In addition, as students got older the percentage who reported asking adults to buy alcohol for them or who purchased alcohol themselves increased. These patterns are similar to those found in the 1998 survey administration.

Table 11:Methods of Obtaining Alcohol in Grades 6, 8, 10, and 12

Method of obtaining	Grade 6	Grade 8	Grade 10	Grade 12
I don't drink	93.0	74.9	55.3	43.8
From home, parents don't know	1.2	4.7	4.0	1.5
Ask adults to purchase or	0.3	1.5	5.9	11.0

buy myself				
From home, parents know	3.8	6.5	6.7	6.5
From friends	1.7	12.4	28.1	37.3

Note. Figures represent percentages of students responding.

Tobacco

Smoking is the leading cause of preventable disease and death in the United States and the health consequences of smoking impose a considerable toll on society. Smoking is a major risk factor for heart disease, stroke, lung cancer, and chronic lung diseases. Direct medical costs related to smoking total at least \$50 billion per year (Centers for Disease Control and Prevention, 1994) and direct medical costs related to smoking during pregnancy are approximately \$1.4 billion per year (Centers for Disease Control and Prevention, 1997). Targets adopted by the Governor's Substance Abuse Prevention Advisory Committee related to tobacco use include:

- Reduce the proportion of Grade 6 students reporting cigarette use during the past 30 days to 3 percent.
- Reduce the proportion of Grade 8 students reporting cigarette use during the past 30 days to 10 percent.
- Reduce the proportion of Grade 10 students reporting cigarette use during the past 30 days to 15 percent.
- Reduce the proportion of Grade 12 students reporting cigarette use during the past 30 days to 20 percent.

Figure 4 illustrates cigarette use across grades as reported in the 2000 WSSAHB administration. As the students got older they were more likely to report having ever tried cigarettes and having used cigarettes in the past 30 days. For example, 15.1 percent of Grade 6 students, but 60.9 percent of Grade 12 students, had tried cigarettes at some point in their lives. Similarly, 4.0 percent of Grade 6 students, but 27.6 percent of Grade 12 students had smoked cigarettes in the past 30 days.



Figure 4: Smoking Tobacco Use in Grades 6, 8, 10, and 12

Figure 5 shows changes in student use of cigarettes in the past 30 days from 1990 through 2000. Cigarette use among Grade 6 students remained essentially unchanged (4.0 percent in 2000). Among Grade 8 students cigarette use in the past 30 days increased from 12.1 percent in 1990 to 18.8 percent in 1995 and then decreased to 12.5 percent in 2000. Among Grade 10 students cigarette use climbed from 1990 (15.1 percent) to 1995 (20.9 percent) and has remained steady since then (19.8 percent in 2000). Among Grade 12 students cigarette use climbed from 1990 (20.7 percent) to 1998 (28.6 percent) and leveled in 2000 (27.6 percent).

Figure 5: Trends in 30-Day Prevalence of Cigarette Use: Grades 6, 8, 10, and 12



Among Grade 6 and Grade 12 students, males and females reported similar patterns of cigarette smoking in the past 30 days. Grade 8 and Grade 10 females were, however, more likely than males to report cigarette use in the past 30 days. Among Grade 8 students, 14.5 percent of females smoked cigarettes in the past 30 days, compared to 10.3 percent of males. Among Grade 10 students, 21.9 percent of females smoked cigarettes in the past 30 days, compared to 17.5 percent of males.

Patterns of cigarette use also varied by ethnic group. American Indian/Alaskan Natives generally reported the highest levels of cigarette use (for example, 44.9 percent of Grade 12 students) and Asian/Pacific Islanders generally reported the least use (for example, 17.1 percent of Grade 12 students.) Cigarette use by members of other ethnic groups generally was within this range, although patterns varied by grade level. Readers are cautioned against overinterpreting these results due to the small numbers of survey respondents included in some ethnic groups. Similar to patterns observed for cigarette use, lifetime and 30-day prevalence of smokeless tobacco use increased as students got older. Figure 6 shows that 1.8 percent of Grade 6 students had ever tried smokeless tobacco and 0.8 percent had used it in the past month. Among Grade 12 students, however, 24.8 percent had ever tried smokeless tobacco and 8.8 percent had used it in the past month.



Figure 6: Smokeless Tobacco Use in Grades 6, 8, 10, and 12

Figure 7 shows changes in prevalence of 30-day smokeless tobacco use from 1995 (the first year this substance was included in the survey) to 2000. Students in Grades 8, 10, and 12 show marked declines in smokeless tobacco use over this period of time.



Figure 7: 30-Day Smokeless Tobacco Use

Exposure to secondhand smoke has serious health effects (California Environmental Protection Agency, 1997; U.S. Department of Health and Human Services, 1986; U.S. Environmental Protection Agency, 1992). Each year, secondhand smoke causes an estimated 3,000 nonsmokers to die of lung cancer, and 150,000 to 300,000 infants and children under age 18 months experience lower respiratory tract infections. Secondhand smoke exposure causes heart disease among adults (Glantz and Parmely, 1995; Pirkle et al., 1996). The surveyed students indicated whether they thought that smoke from other people's cigarettes (secondhand smoke) is harmful. More than 85 percent of students in all grades believed that secondhand smoke was *probably* or *definitely* harmful (see Figure 8).



Figure 8: Perceived Risk of Secondhand Smoke

Students were also asked whether during the past year they had practiced in their classes ways to say *no* to tobacco—in role plays, for example. About twothirds (69.7 percent) of the Grade 6 students indicated that they had practiced in class ways to say no to tobacco during the past year. This percentage decreased at each grade level, however, and only about one in seven (14.8 percent) Grade 12 students had practiced tobacco refusal. Figure 9 illustrates these findings.



Figure 9: Students Who Practiced Tobacco Refusal Skills in Class

The survey asked students whether their parents or guardians had discussed the dangers of tobacco use with them. About three fourths (77.8 percent) of Grade 6 students indicated that such a discussion had occurred. Students in each higher grade were somewhat less likely to answer that this had been the case; about two-thirds (63.8 percent) of Grade 12 students reported that they had discussed the dangers of tobacco use with their parents or guardians (see Figure 10).





Students reported how often during the past 30 days they had seen antismoking ads on television or heard them on the radio. About two-thirds (67.8 percent) of Grade 6 students reported having seen or heard antismoking ads at least once a week during the past 30 days. About three-fourths of the students in the higher grades indicated having seen or heard antismoking ads at least once a week during the past 30 days (77.1 percent of Grade 8 students, 77.7 percent of Grade 10 students, and 73.2 percent of Grade 12 students). Figure 11 shows these findings.

Figure 11: Student Exposure to Antismoking Television and Radio Ads in the Past 30 Days



Marijuana

Along with alcohol and tobacco, marijuana is a gateway drug with important health consequences. Marijuana use is of concern given its prevalence in adolescent treatment. Nationally, in 1998 57 percent of youth aged 12 to 17 admitted to treatment reported marijuana as the primary substance of abuse. The federal Center for Substance Abuse Treatment conducted the Cannabis Youth Treatment Experiment to identify effective adolescent treatment for marijuana use (Center for Substance Abuse Treatment, 2000). Targets adopted by the Governor's Substance Abuse Prevention Advisory Committee related to tobacco use include:

- Reduce the proportion of Grade 6 students reporting marijuana use during the past 30 days to 1 percent.
- Reduce the proportion of Grade 8 students reporting marijuana use during the past 30 days to 8 percent.

- Reduce the proportion of Grade 8 students reporting marijuana use during the past 30 days to 10 percent.
- Reduce the proportion of Grade 8 students reporting marijuana use during the past 30 days to 14 percent.

Current use (i.e., use in the past 30 days) of marijuana decreased from 1998 to 2000. Among Grade 8 students, marijuana use in the past 30 days more than doubled from 1992 to 1995, remained stable in 1998 (16.5 percent), and then decreased in 2000 (12.0 percent). Among Grade 10 students, marijuana use in the past 30 days climbed steadily from 1990 to 1998 (26.6 percent) and then decreased in 2000 (21.9 percent). Among Grade 12 students, marijuana use in the past 30 days also climbed steadily from 1990 to 1998 (28.7 percent) and then decreased in 2000 (24.4 percent). Although the recent decreases in current use are encouraging, use levels remain quite high: about one in eight Grade 8 students, one in five Grade 10 students, and one in four Grade 12 students reported marijuana use in the past 30 days (see Figure 12). These declines represent a substantial decrease in the number of marijuana users. For example, given the October 2000 enrollment of 68,580 Grade 12 students, this decline represents a decrease of over 2,900 marijuana users in Grade 12.

Figure 12: Trends in 30-Day Prevalence of Marijuana Use in Grades 6, 8, 10 and 12



At all four grades levels males were more likely than females to report any marijuana use and heavy marijuana use in the past 30 days (the differences are, however, very small among the Grade 6 students). Among Grade 8 students, 13.7 percent of males reported any marijuana use in the past 30 days, compared to 10.3 percent of females, and 4.1 percent of males reported using marijuana 10 or more times in the past 30 days, compared to 2.5 percent of females. Among Grade 10 students, 23.8 percent of males reported any marijuana use in the past 30 days, compared to 20.1 percent of females, and 10.2 percent of males reported using marijuana 10 or more times in the past 30 days, compared to 6.2 percent of females. Among Grade 12 students, 28.0 percent of males reported any marijuana use in the past 30 days, compared to 20.6 percent of females, and 12.3 percent of males reported using marijuana 10 or more times in the past 30 days, compared to 5.9 percent of females.

Patterns of marijuana use also varied by ethnic group, although the patterns also varied by grade level; thus no clear pattern can be associated with any particular ethnic group (again, readers are cautioned against overinterpreting these results due to the small number of survey respondents included in some ethnic groups).

Attitudes Toward Alcohol, Tobacco, and Other Drug Use

The rise in illicit drug use that occurred in the 1990s was at least partially attributed to the erosion of antidrug attitudes and norms (Johnston et al., 1994). One of the key attitudes influencing alcohol, tobacco, and other drug use is the perception of harm that smoking, excessive drinking, or regular use of marijuana causes. Bachman, Johnston, and O'Malley (1998) analyzed trends in marijuana use from 1975 to 1996 and noted:

The fundamental conclusion that was drawn from the present analyses, as well as earlier ones, is that attitudes about specific drugs—disapproval of use and perceptions of risk of harmfulness—are among the most important determinants of actual use . . . If we want to know why marijuana use is on the rise again . . . we need to ask why it is that [youth] have become less concerned in recent years about the risks of marijuana use, and why they do not disapprove of such use as strongly as students did just a few years earlier. The implication for prevention is that presenting such information [about risks and consequences of drug use] once does not finish the job; the messages must be repeated lest they be lost from one cohort to the next.

Fortunately, key attitudes and beliefs about drugs that have proven to be important determinants of use began to reverse in 1997 (Johnston, O'Malley and Bachman, 1997).

Figure 13 shows the association between the perceived risk of binge drinking and the prevalence of binge drinking in the past two weeks for Grade 8 students. From 1988 to 1992 an increase in the perceived risk was associated with a decreased prevalence of binge drinking. Then, from 1992 to 1995 a decreased perception of risk was associated with an increased prevalence of binge drinking. More recently, from 1995 to 2000, a continued increase in the perceived risk was associated with a leveling and then decreased prevalence of binge drinking.



Figure 13: Trends in Perceived Risk and Binge Drinking in the Past Two Weeks Among Grade 8 Students

Figure 14 shows the association between the perceived risk of marijuana use and the prevalence of marijuana use in the past month for Grade 6, 8, 10, and 12 students. From 1988 to 1992 a steady level of perceived risk was associated with an unchanging prevalence of marijuana use. Then, from 1992 to 1995 a decreased perception of risk was associated with an increased prevalence of marijuana use. More recently, from 1995 to 2000, a continued increase in perceived risk was associated with a leveling and then decreased prevalence of marijuana use.





The relationships between the perceived risk and actual binge drinking and marijuana use are not conclusive proof of the causal influence of attitudes on behavior. Indeed, some would argue that the behavior occurs first and attitudes are formed to support the behavior. The strong inverse association of these trends is, however, strongly suggestive of the close link between the perceived health risk and the actual behavior.
Chapter 3: Intentional Injury Behaviors

In 1984 the U.S. Surgeon General declared violence as much a current national public health issue as smallpox, tuberculosis, and syphilis had been decades earlier. Fundamental to the public health perspective on violence is a shift away from a reactive effort toward a proactive effort to change the social, behavioral, and environmental factors that cause violence (Mercy, 1993). Central to this approach is the objective measurement of the incidence and prevalence of violence and violence-related behaviors.

Injury is the leading cause of death for children aged 5 to 17 (National Center for Health Statistics, 1998), and violence claims the lives of many of the nation's youth each year. Fighting, weapon carrying, and attempted suicide are all health risk behaviors associated with threats to personal safety, future injury, and death. Healthy People 2010 objectives related to intentional injury and related risk behavior include "Reduce physical fighting in the past year among adolescents in Grades 9 through 12 to 32 percent" and "Reduce weapon carrying on school property during the past 30 days among adolescents in Grades 9 through 12 to 4.9 percent."

The WSSAHB contained a question that asked students how many times in the past 30 days they had carried a weapon, such as a gun, knife, or club, for self-protection or because they thought they might need it in a fight. Figure 15 illustrates changes in this behavior over time. Grade 6 students showed a steady decline from 1992 to 1998 in weapon carrying, but were not asked this question in 2000. Grade 8 students showed a decrease in weapon carrying from 1992 to 1995 and then again from 1998 to 2000. Grades 10 and 12 students reported decreased weapon carrying from 1992 to 1995, but a level prevalence since then (although results for Grade 10 students suggest a small continued decline).

In 2000 about one in ten Grade 8, 10, and 12 students reported carrying a weapon in the past 30 days. Unlike substance abuse behaviors, the prevalence of weapon carrying is similar across age groups.



Weapon Carrying in the Past 30 Days

Males were more likely than females to report weapon carrying. Among Grade 8 students, 16.9 percent of males reported weapon carrying in the past 30 days, compared to 5.6 percent of females. Among Grade 10 students, 17.2 percent of males reported weapon carrying in the past 30 days, compared to 3.7 percent of females. Among Grade 12 students, 15.7 percent of males reported weapon carrying in the past 30 days, compared to 5.7 percent of females. The survey asked the students how many times in the past year they had attacked someone with the idea of seriously hurting that person. Figure 16 shows a decrease in this behavior from 1998 to 2000 among Grade 8, 10, and 12 students (Grade 6 students were not asked this question). One in eight (12.4 percent) Grade 8 students and about one in ten Grade 10 (10.8 percent) and Grade 12 (9.1 percent) students reported that they had attacked someone with the idea of seriously hurting that person during the past year.

Figure 16: Prevalence of Attacking Someone With the Idea of Seriously Hurting That Person



Students were asked how old they were when they first belonged to a gang (sometimes referred to as an organization, click, clique, set, or posse). Figure 17 shows the percentages of the students who reported having ever belonged to a gang (i.e., they answered with an age of first belonging rather than stating *never have*). Among Grade 6 students, one in 20 (5.0 percent) reported having belonged to a gang. About one in nine Grade 8 (11.7 percent) and Grade 10 (11.2 percent) students reported having ever belonged to a gang. Among Grade 12 students, one in ten (10.3 percent) reported having ever belonged to a gang.



Figure 17: Percent of Students Who Had Belonged to a Gang

The WSSAHB included specific questions regarding different indicators of violent behavior (e.g., fighting, weapon carrying). Combining the information across indicators provides school and health officials with a composite index of violent behaviors that can be used as a general indicator of the magnitude of violent behavior in a given school. The authors combined the information from three WSSAHB questions to define the following three levels of violent or delinquent behavior: (a) *none*—no violent or delinquent behavior reported in the past 12 months, (b) *infrequent*—one or two violent or delinquent behaviors in the past 12 months, and (c) *frequent*—three or more violent or delinquent behaviors or the occurrence of a single violent incident or delinquent behavior ten or more times in the past 12 months. The three questions included in the violent behavior scale were 30-day weapon carrying, carrying a handgun in the past year, and attacked someone in the past year (Questions 96, 98b, and 98f). The three questions included in the delinquent behavior scale were been suspended from school in the past year, sold illegal drugs in the past year, and been arrested in the past year (Questions 98a, 98c, and 98e). Additional details about the construction of these scales are in the Technical Report. These scales are not comparable to the scales constructed in 1995 and 1998 due to changes to the questions.

Figure 18 shows the percentages of students at each grade level who exhibited some degree of violent and delinquent behavior (Grade 6 students were not asked all the questions necessary to compute these scales). The prevalence of at least some violent behavior is fairly consistent across grade levels—about one in five Grade 8, 10, and 12 students had engaged in at least one violent behavior in the past year. Delinquent behavior was also fairly consistent across grade levels—about one in six Grade 8, 10, and 12 students had engaged in at least one violent behavior in the past year.

Figure 18: Prevalence of Violent and Delinquent Behavior in the Past Year



People who are depressed experience a range of symptoms, including sadness, loss of usual interests and pleasures, sleep disturbance, weight or appetite disturbance, difficulty concentrating, intense feelings of guilt, and suicidal thoughts or behaviors (Keefe and Harvey, 1994). Individuals who experience depression may experience social or occupational impairment. Even if this impairment is not present, normal functioning requires markedly increased effort. People who experience a depressed mood or loss of interest for at least two weeks, accompanied by at least four additional symptoms of depression, are said to have a major depressive episode (American Psychiatric Association, 1994). This serious condition is associated with high mortality—up to 15 percent of individuals with severe major depressive disorder die by suicide.

The 2000 WSSAHB contained one question related to depression: "During the past 12 months, did you ever feel so sad or hopeless almost every day for two weeks or more in a row that you stopped doing some usual activities?" Although this question is not sufficient to diagnose depression, Figure 19 illustrates that by

the time the students reached Grade 6, nearly one in four (23.1 percent) reported having experienced depressive feelings during the past year. Among Grade 8, 10, and 12 students, about one in three students reported having experienced depressive feelings during the past year.



Figure 19: Students Reporting Experience of Depressive Feelings

Among Grade 6 students, males and females were about equally likely to report experiencing depressive feelings in the past year (24.0 percent of females and 21.7 percent of males). As the students got older, however, females were more likely than males to report that they had experienced depressive feelings during the past year. Among Grade 8 students, 33.9 percent of females reported having experienced depressive feelings, compared to 26.4 percent of males. Among Grade 10 students, 41.3 percent of females reported having experienced depressive feelings, compared to 25.9 percent of males. Among Grade 8 students, 38.3 percent of females reported having experienced depressive feelings, compared to 25.9 percent of males. Depression is an important condition to address due to its association with suicide. Attempted suicide heightens the risk of eventual suicide and is related to a host of other problem behaviors, such as substance abuse and delinquency. Although the 2000 WSSAHB did not include questions regarding suicide ideation and suicidal behavior, the 1992 and 1995 administrations did include such questions. In 1995 about one in five students had thought about suicide and about one in ten had made a plan to attempt suicide. A similar proportion of students, about one in ten, had actually attempted suicide. About 30 percent of these attempted suicides resulted in injury.

The 2000 administration of the WSSAHB included expanded coverage of school climate. Questions asked for the first time addressed perceived safety at school and bullying behavior. Other questions, which had been asked before, addressed weapon carrying at school. The Governor's Substance Abuse Prevention Advisory Committee aims to increase the percentage of adolescents reporting that they feel safe in school to 85 percent of Grade 6 students, 85 percent of Grade 8 students, 85 percent of Grade 10 students, 90 percent of Grade 12 students. In 1998, 79 percent of Grade 6 students, 74 percent of Grade 8 students, 79 percent of Grade 10 students, 86 percent of Grade 12 students reported that they felt safe at their school. In 2000 the percentage of Grade 6 students who reported feeling safe in their school increased to 86 percent and the percentage of Grade 8 students who reported feeling safe in their school increased to 77 percent.

In addition, students reported how safe they felt in several areas in school, including the classroom, the halls or stairs, the bathroom, the locker rooms, the playground or school grounds, the lunchroom, on the bus, on the way to school, and on the way home from school. Table 12 details the results. Although most students generally felt safe in school, too large a percentage reported feeling unsafe in at least some areas of school. In particular, students felt that the classroom is a safe area (Grade 8 students were most likely to report feeling a little or very unsafe in the classroom; 8.9 percent reported this feeling). Students were more likely to report feeling unsafe in the bathrooms and on the playground or school grounds. In general, younger students were less likely than older students to report feeling safe at school. For example, one in four Grade 6 students (24.8 percent) reported feeling a little or very unsafe on the playground or school grounds, compared to 23.5 percent of Grade 8, 18.7 percent of Grade 10, and 12.8 percent of Grade 12 students.

	Percent of Students Responding										
Location	Grade 6	Grade 8	Grade 10	Grade 12							
In class											
Very unsafe	1.8	2.3	2.7	2.3							
A little unsafe	4.8	6.6	5.0	3.2							
Mostly safe	29.9	44.0	41.4	37.0							
Very safe	63.5	47.1	50.9	57.4							
In the halls or stairs											
Very unsafe	4.5	5.0	4.7	3.1							
A little unsafe	16.4	18.5	14.5	10.2							
Mostly safe	41.1	50.0	49.1	44.7							
Very safe	38.0	26.5	31.7	42.0							
In the bathroom											
Very unsafe	6.0	7.1	6.3	4.4							
A little unsafe	18.3	18.3 15.6		8.8							
Mostly safe	36.3	45.2	45.1	42.4							
Very safe	39.4	32.1	35.9	44.5							
In the locker rooms											
Very unsafe	6.1	6.3	5.6	4.0							
A little unsafe	13.3	12.3	10.7	8.6							
Mostly safe	35.1	47.2	46.3	41.7							
Very safe	45.6	34.3	37.4	45.7							
On the playground or school grounds											
Very unsafe	6.4	6.9	5.7	3.9							
A little unsafe	18.4	16.6	13.0	8.9							
Mostly safe	39.2	48.3	48.3	44.2							
Very safe	36.1	28.2	33.0	43.0							
In the lunchroom											
Very unsafe	3.0	4.5	4.6	3.5							
A little unsafe	6.7	8.8	8.4	6.8							
Mostly safe	32.6	44.7	44.8	41.6							
Very safe	57.6	42.0	42.2	48.1							

Table 12:Perceived Safety at School

	Percentage of Students Responding								
Location	Grade 6	Grade 8	Grade 10	Grade 12					
On the bus									
Very unsafe	6.5	5.4	4.9	3.5					
A little unsafe	11.4	9.2	6.9	3.0					
Mostly safe	30.3	28.6	20.5	7.6					
Very safe	23.7	19.8	16.2	5.8					
Don't take the bus	28.0	36.9	51.6	80.1					
On the way to school									
Very unsafe	4.9	4.2	3.6	2.9					
A little unsafe	11.7	10.5	7.0	5.9					
Mostly safe	37.1	39.4	37.5	36.4					
Very safe	46.4	45.8	51.9	54.8					
On the way home after school									
Very unsafe	5.6	5.0	4.1	3.3					
A little unsafe	12.8	14.4	8.9	6.6					
Mostly safe	39.9	42.8	39.5	36.4					
Very safe	41.7	37.9	47.6	53.8					

Table 12, continued

Many students reported having experienced bullying at school, although most did not. For example, among the Grade 6 students, one in five (20.9 percent) reported that other students at school put them down verbally *a lot* or every day. Younger students were more likely than older students to report having experienced bullying at school. Figure 20 illustrates these results.

Figure 20: Students Who Experienced Bullying Behaviors at School



Students were asked what they would do if they saw a student bullying another student at school. Students in Grades 6, 8, and 10 were most likely to report that they would tell the bully to stop. The Grade 6 students were, however, more likely than the students in the other grades to report that they would tell an adult at school. In contrast, older students were more likely than younger ones report that they would either walk away or mind their own business or to stay and watch. Figure 21 illustrates these results.



Figure 21: Students' Reactions in Bullying Situations

Students reported the last time they had carried a gun; knife or razor; or club, stick, pipe, or other weapon on school property for self-protection or because they thought they might need the weapon in a fight. Figure 22 shows the percentages of students who reported that they had carried a weapon within the past month. The percentage of students who reported this behavior is quite low, although the behavior does occur, and Grade 8 and Grade 10 students were more likely than Grade 6 or Grade 12 students to carry a weapon to school. The Healthy People 2010 objectives related to intentional injury and related risk behaviors include "Reduce weapon carrying on school property during the past 30 days among adolescents in Grades 9 through 12 to 4.9 percent."



Figure 22: Students Who Carried a Weapon to School in the Past Month

Chapter 5: Risk and Protective Factors

The adolescent health risk behaviors addressed in this report have many implications for the students, families, schools, and communities in which they occur. Decades of research have shown that *risk factors* are associated with increased likelihood of health risk behaviors, including alcohol, tobacco, and other drug abuse (Hawkins et al., 1992), and violence and delinquent behaviors (Bensley and Van Eenwyk, 1995; Brewer, Hawkins, Catalano, and Neckerman, 1995). Similarly, *protective factors* exert a positive influence or buffer against the negative influence of risk.

The 1995, 1998, and 2000 WSSAHB administrations included substantial coverage of risk and protective factors using standardized assessment tools developed by the Social Development Research Group at the University of Washington (Arthur et al., 1998). The reliability analyses for the scales measuring these factors are presented in the Technical Report. Fourteen risk factors were assessed and organized into four domains of influence—community, family, school, and peer-individual:

Community

- Low neighborhood attachment.
- Laws and norms favorable toward drug use.
- Perceived availability of drugs.
- Perceived availability of handguns.

Family

- Poor family management. School
- Academic failure.
- Low commitment to school.

Peer-Individual

• Early initiation of drug use.

- Early initiation of problem behavior.
- Favorable attitudes toward antisocial behavior.
- Favorable attitudes toward drug use.
- Perceived risk of use.
- Friends' use of drugs.
- Rewards for antisocial involvement.

Because the family domain was optional on the 2000 WSSAHB, the number of students who answered the questions in this domain was smaller than the numbers of respondents for the other domains and not all of the sampled schools asked these questions. Thus the results for the family domain are not included in this report.

Another body of research has focused on the abilities of young people to overcome the odds that challenge them (Werner and Smith, 1989) and succeed in spite of a preponderance of risk in their environments. Benard (1991) summarized this literature on protective factors, citing the longitudinal research of Werner and Smith and Rutter (1979) in the formulation of a construct termed *resilience*. Resnick et al. (1997) found that parent-family connectedness and perceived school connectedness were protective against every health risk behavior measured in their study except history of pregnancy. Parental expectations regarding school achievement and school connectedness were also associated with lower levels of health risk behaviors (except in the case of suicide, in which only parent-family connectedness was protective).

The WSSAHB also assessed 8 protective factors (again, results for the optional family domain are not included in this report):

Community

- Opportunities for prosocial involvement.
- Rewards for prosocial involvement.

Family

- Opportunities for prosocial involvement.
- Rewards for prosocial involvement.

School

- Opportunities for prosocial involvement.
- Rewards for prosocial involvement.

Peer-Individual

- Social skills.
- Belief in the moral order.

Several researchers and government agencies have described a risk reduction and protective factor enhancement approach as the most promising approach to preventing problem behaviors (Hawkins et al., 1992; Institute of Medicine, 1994). The premise of this approach is that preventing a problem before it occurs necessitates addressing the factors that predict the problem. Ideally, doing so entails discovering the causes of the problem behavior and influencing the causes. Today, longitudinal research has identified several factors that are potential causes of problem behaviors. Further work is necessary to determine which of these factors are truly causal. In the interim, these risk and protective factors represent promising inputs for prevention and intervention programs and policies.

This chapter presents the 2000 results of the WSSAHB assessment of risk and protection at each grade level in each of three domains. The relationship between risk and protective factors and the major health risk behaviors of alcohol use, drug use, violent behavior, and delinquent behavior are also presented. Readers should remember that all results are based on student selfreport and therefore represent perceptions of risk and protection, which may not be accurate. Furthermore, the statistical relationships between risk and protective factors and health risk behaviors are not necessarily causal relationships. Rather, the statistical relationships indicate an association or cooccurrence of these factors and behaviors. Both the risk factor and the behavior may be associated with a third factor such as poverty or other factors that were not included in this study. Similarly, some apparent relationships may be confounded with age.

Community Domain

The WSSAHB assessed four risk factors and two protective factors in the community domain. For purposes of this report, these risk and protective factors are described as follows:

Risk Factors

- Low neighborhood attachment. Students who do not feel a part of the neighborhood in which they live and feel that what they do there does not makes a difference in their lives are at higher risk for crime and substance abuse.
- Laws and norms favorable toward drug use. The policies a community holds in relation to health and problem behaviors are communicated through laws, social practices, and expectations and are related to use.
- Perceived availability of drugs. Perceptions of the availability or access to alcohol and other drugs have been shown to predict use of these substances.
- Perceived availability of handguns. Perceptions of the availability or access to handguns may be related to the use of handguns.

Protective Factors

- Opportunities for prosocial involvement. Youth need opportunities to participate meaningfully in activities in the community.
- Rewards for prosocial involvement. Youth need rewards for positive participation in prosocial activities.

Each risk and protective factor scale is calculated as the average of one or more questions. Students whose scores placed them above a cut point determined by recent analyses conducted by the University of Washington Social Development Research Group were considered at risk on a given risk factor or resilient on a given protective factor. Table 13 details the percentages of students at risk and the percentages of students resilient on the risk and protective factor scales in the community domain. The 1998 results reflect a reanalysis of the 1998 survey data conducted by the Department of Social and Health Services' Research and Data Analysis based on updated cutpoints that determine whether a student is considered at risk. The Governor's Substance Abuse Prevention Advisory Committee has selected a number of key risk and protective factors for which to establish state targets. As this report is going to press, those targets have not been adjusted to reflect reanalysis based on updated cutpoints.

		Gra	de 6	Gra	de 8	Grac	le 10	Grac	de 12
Fa	ctor	1998	2000	1998	2000	1998	2000	1998	2000
	Low neighborhood attachment	53.1	48.6	40.1	35.0	44.7	43.8	48.2	48.2
Risk	Laws and norms favorable toward drug use	52.4	37.5	46.7	33.3	55.5	44.1	49.6	42.3
	Perceived availability of drugs	40.3	26.8	47.7	34.9	58.7	48.8	60.9	55.9
	Perceived availability of handguns	14.2	22.7	22.3	35.7	30.9	25.3	37.7	32.6
Prote	Opportunities for prosocial involvement	-	42.4	-	56.5	_	48.9	-	47.1

Table 13:Profile of Community Risk and Protective Factors

Rewards for	62.2	67.4	54.0	52.6	44.3	55.7	52.2	51.5
prosocial involvement								

Note. Figures indicate the percentage age of survey participants reporting the risk and protective factors. Dashes indicate this protective factor was not present in the 1998 administration.

Table 13 illustrates three noteworthy points. First, as students got older they were at considerably increased risk on the factor of perceived availability of drugs. Second, Grade 6 students were more likely than older students to be resilient on the factor of rewards for prosocial involvement. Third, changes occurred from 1998 to 2000 in the percentages of students at risk or resilient on some factors. In particular, the percentage of Grade 12 students at risk on the factors laws and norms favorable toward drug use and perceived availability of drugs decreased.

The correlations that appear in Table 14 show the strength of the relationships between the community risk and protective factors and global measures of health risk behaviors on the survey. Results of the four risk and two protective factor scales are correlated with four health behavior composite scales: alcohol use, drug use, violent behavior, and delinquent behavior. A positive correlation indicates a direct relationship. For example, being at higher risk on a risk factor is associated with higher alcohol use if the two are positively correlated. A negative correlation indicates an inverse relationship. For example, being at greater resilience on a protective factor is associated with less alcohol use if the two are negatively correlated. Correlations may range between -1 (prefect inverse correlation) and +1 (perfect direct correlation); a correlation of 0 indicates that the two variables are not associated with one another. Table 14 reports only statistically significant correlations, which provides the assurance that the observed relationships between the risk or protective factors with the health behavior scales are stronger than could be expected by chance or coincidence. Statistical significance proffers greater confidence that the relationship is real, replicable, and has implications for prevention programs.

Table 14:Correlation of Community Risk and Protective Factors With Health BehaviorScales

Fac	tor	Alcohol Use	Drug Use	Violent Behavior	Delinquent Behavior
	Low neighborhood attachment	0.20	0.20	0.16	0.15
tive Risk	Laws and norms favorable toward drug use	0.46	0.44	0.27	0.28
	Perceived availability of drugs	0.57	0.53	0.23	0.29
	Perceived availability of handguns	0.27	0.25	0.28	0.20
	Opportunities for positive involvement	-0.24	-0.24	-0.14	-0.18
Protec	Rewards for conventional involvement	-0.21	-0.21	-0.12	-0.14

Note. Figures are correlations between the community risk and protective factors and the health risk behaviors. A correlation of -1 indicates prefect inverse correlation, a correlation of +1 indicates perfect direct correlation, and a correlation of 0 indicates that the two variables are not associated with one another.

The strongest correlations between community risk factors and health risk behaviors involved the perceived availability of drugs and laws and norms favorable toward drug use. The correlations themselves, and their relative strength, were generally similar in 1998 and 2000.

School Domain

School is an environment in which young people spend a great deal of time. As a result, schools have the opportunity, although not the sole responsibility, to greatly influence adolescent development. The 2000 WSSAHB included two risk factors and two protective factors in the school domain:

Risk Factors

- Academic failure. Children fail in school for many reasons, but research indicates that the very experience of failure, regardless of whether the failure is linked to the students' abilities, places them at higher risk of negative behavior.
- Little commitment to school. When young people cease to see the school role as viable, they are at higher risk of engaging in the health risk behaviors.

Protective Factors

- Opportunities for prosocial involvement. When young people are given more opportunities to participate meaningfully in important activities at school, they are less likely to engage in problem behaviors.
- Rewards for prosocial involvement. When young people are recognized and rewarded for their contributions at school, they are less likely to be involved in health risk behaviors.

Table 15 details the percentages of students at risk and percentages of students resilient on the risk and protective factors in the school domain. The percentage of students who showed little commitment to school increased for older students, from 35.2 percent of Grade 6 students to 47.3 percent of Grade 12 students. In addition, the percentage of students who perceived rewards for conventional involvement dropped from 60.1 percent of Grade 6 students at risk on the factor low commitment to school increased from 1998 to 2000. The percentage of students resilient on the factor opportunities for prosocial involvement increased from 1998 to 2000 among Grade 6, 8, and 10 students. The percentage of students.

		Gra	<u>Grade 6</u>		<u>Grade 8</u>		<u>Grade 10</u>		le 12
Fa	Factor		2000	1998	2000	1998	2000	1998	2000
	Academic failure	40.2	39.9	42.5	41.4	41.8	38.2	38.0	41.3
R	Little commitment to school	36.7	35.2	33.8	39.4	31.7	42.5	33.1	47.3
P	Opportunities for involvement	54.4	59.2	54.3	60.5	53.6	57.4	55.9	57.7
Р	Rewards for conventional involvement	51.6	60.1	47.8	52.8	59.8	59.3	47.3	45.0

Table 15:Profile of School Risk and Protective Factors

Note. R = Risk. P = Protective. Figures indicate the percentage of survey participants reporting the risk and protective factors.

Table 16 details the correlations between risk and protective factors in the school domain and the four health behavior scales. In the school domain there were moderate correlations between the risk factor of little commitment to school and alcohol use.

Table 16:Correlations of School Risk and Protective Factors With Health Behavior Scales

Factor		Alcohol Use	Drug Use	Violent Behavior	Delinquent Behavior
	Academic failure	0.23	0.26	0.22	0.28
R	Little commitment to school	0.41	0.39	0.27	0.28
D	Opportunities for involvement	-0.21	-0.21	-0.18	-0.17
٢	Rewards for conventional involvement	-0.24	-0.22	-0.19	-0.14

Note. R = Risk. P = Protective. Figures are correlations between the school risk and protective factors and the health risk behaviors. A correlation of -1 indicates prefect inverse correlation, a correlation of +1 indicates perfect direct correlation, and a correlation of 0 indicates that the two variables are not associated with one another.

Peer-Individual Domain

The social environments of the school and community greatly influence young people's behavior. In addition, many characteristics of individuals and attributes of peer groups are powerful determinants of behavior. To study this, the survey included 7 risk factors and 2 protective factors in the peer-individual domain:

Risk Factors

- Early initiation of drug use. Research clearly shows that the earlier an individual begins using alcohol, tobacco, and other drugs, the more likely he or she is to develop drug use problems in adolescence.
- Early initiation of problem behavior. Research clearly shows that the earlier an individual begins engaging in delinquent and violent behavior, the more likely he or she is to develop delinquent or violent behavior problems in adolescence.
- Favorable attitudes toward antisocial behavior. Young people who accept or condone antisocial behavior are more likely to engage in health risk behaviors.
- Favorable attitudes toward drug use Young people who have positive or accepting attitudes toward drug use are more likely to engage in a variety of health risk behaviors.
- Perceived risk of use. Young people who do not perceive a risk in using alcohol, tobacco, and other drugs are at higher risk of engaging in substance use.
- Friends' use of drugs. Young people whose friends use drugs are more likely to engage in health risk behaviors.
- Rewards for antisocial involvement. Young people who believe that they are favorably perceived as a result of engaging in antisocial behavior are more likely to engage in that behavior.

Protective Factors

- Social skills. Young people who are socially competent and engage in positive interpersonal relations with their peers are less likely to participate in negative health risk behaviors.
- Belief in the moral order. Young people who have a belief in what is right or wrong are at lower risk for engaging in problem behaviors.

Table 17 shows the profile of the peer-individual risk and protective factors across grade levels. As students got older they were more likely to report being at risk in terms of early initiation of drug use, early initiation of problem behavior, favorable attitudes toward drug use, perceived risk of drug use, and friends' use of drugs. In addition, from 1998 to 2000 there was an increase in the percentages of students at risk on the factor of friends' use of drugs. Decreases on the factors of early initiation of problem behavior, favorable attitudes toward drug use, and rewards for antisocial involvement are also evident. The percentage of students resilient on the protective factor social skills increased among Grade 8 and Grade 10 students and increased on the protective factor belief in the moral order among Grade 6 and Grade 8 students.

		Grad	<u>de 6</u>	Grad	<u>Grade 8</u>		le 10	Grac	de 12
Fo	actor	1998	2000	1998	2000	1998	2000	1998	2000
	Early initiation of drug use	42.8	27.1	47.0	44.8	51.3	45.5	44.9	48.7
Risk	Early initiation of problem behavior	23.4	18.0	36.6	28.9	39.7	31.8	35.5	33.4
	Favorable attitudes toward antisocial behavior	36.3	32.3	43.4	36.6	40.9	43.4	34.5	41.9
	Favorable attitudes toward drug use	34.4	23.5	45.0	34.4	51.9	45.4	52.1	47.1
	Perceived risk of use	_	13.8	_	21.8	_	20.9	_	26.4

Table 17:Profile of Peer-Individual Risk and Protective Factors

	Friends' use of drugs	41.4	22.9	52.2	37.5	53.1	42.2	46.6	43.4
	Rewards for antisocial involvement	38.8	25.4	52.5	42.7	39.9	38.1	48.7	43.6
ctiv	Social skills	64.9		55.7	66.1	49.6	55.4	62.4	64.2
Prote	Belief in the moral order	45.3	56.8	54.1	64.4	66.4	69.2	54.5	57.4

Note. Figures indicate the percentage of survey participants reporting the risk and protective factors. Dashes indicate that the risk or protective factor was not computed for a given grade or year or not comparable across years.

Table 18 shows the strength of the relationships between peer-individual risk and protective factors and health risk behaviors. Consistent with previous administrations of the WSSAHB, in 2000 the strongest peer-individual correlates with health risk behaviors were early initiation of drug use, attitudes favorable toward drug use, and friends' use of drugs. As the research literature has shown abundantly, the earlier a student begins experimenting with any of these problem behaviors, the more likely he or she is to advance to higher levels of that behavior or experience problematic consequences in later adolescence, or both.

Factor		Alcohol Use	Drug Use	Violent Behavior	Delinquent Behavior
	Early initiation of drug use	0.76	0.64	0.36	0.41
Risk	Early initiation of antisocial behavior	0.34	0.41	0.57	0.57
	Attitudes favorable toward antisocial behavior	0.46	0.46	0.42	0.37
	Attitudes favorable toward drug use	0.65	0.65	0.33	0.40
	Friends' use of drugs	0.67	0.70	0.31	0.43
	Rewards for antisocial behavior	0.29	0.30	0.21	0.20

Table 18: Correlations of Peer-Individual Risk and Protective Factors With Health Behavior Scales

	Perceived risk of substance use	0.37	0.41	0.27	0.34
ctiv	Belief in the moral order	-0.46	-0.44	-0.37	-0.34
Prote	Social skills	-0.57	-0.56	-0.39	-0.42

Note. Figures are correlations between the peer-individual risk and protective factors and the health risk behaviors. A correlation of -1 indicates prefect inverse correlation, a correlation of +1 indicates perfect direct correlation, and a correlation of 0 indicates that the two variables are not associated with one another.

Consistent with previous administrations of the WSSAHB, in 2000 the protective factors showed their strongest relationships with health risk behaviors in the peerindividual domain. An internal belief in the moral order and positive social skills were associated with lower levels of alcohol use, drug use, delinquent behavior, and violent behavior.

The results presented thus far in this chapter have addressed risk and protective factors on an individual basis. The tables have shown the strength of the relationship of each of these factors with health risk behaviors and grade-to-grade differences in the severity of risk and the level of protection on each factor. The influence of specific risk or protective factors on health risk behaviors is important to demonstrate because it can provide useful guidance to state and local prevention efforts.

Research has also suggested a cumulative effect in the influence of risk and protection on these health risk behaviors (Bry, McKeon, and Pandina, 1982; Newcomb, Maddahian, and Skager, 1987; Werner and Smith, 1989). In addition to examining the specific influence of a given risk or protective factor, examining the relationship between multiple risk or protective factors and these behaviors is important. This examination helps illustrate whether students who are at high risk on more risk factors are more likely to engage in health risk behaviors than students who are at high risk on fewer factors. An examination of the relationship between multiple risk or protective factors and health risk behaviors also helps show whether students who are well protected are less likely to engage in these behaviors than students who are less protected.

Figure 23 displays the relationship between the number of risk factors present and the use of alcohol and other drugs. To assess this relationship, students participating in the survey were classified as to whether they were at high risk on none, one, two, three, or more of the risk factors assessed on the 2000 WSSAHB. Twelve risk factors were used as the maximum because too few students in the groups possessed more than 12 risk factors to provide stable estimates of alcohol and other drug use. Once these groups were formed, the prevalence rates of alcohol use (both lifetime and 30-day) and other drug use (both lifetime and 30day) were calculated. These group means were then plotted to display the relationship shown in the figure.

Perhaps the most obvious interpretation is the clear, linear relationship between the number of risk factors present and the prevalence of lifetime and 30-day alcohol and other drug use. Clearly, as the number of risk factors for individual students increased, the more likely they were to use alcohol and other drugs. These findings are consistent with the findings from the two previous WSSAHB administrations.

Figure 23: The Relationship Between Alcohol and Drug Use and the Number of Risk Factors Reported by Washington Students



Figure 24 is a similar display relating the presence of protective factors to alcohol and other drug use. Again, the overall relationship was a strong one, with increased levels of protection (i.e., the presence of several protective factors in students) clearly associated with lower rates of alcohol and other drug use. Protective factors have also been found to have a moderating effect on the presence of risk factors (DeWit, Silverman, Goodstadt, and Stoduto, 1995; Gabriel, Deck, Einspruch, Nickel, 1997; Jessor, Van den Bos, Vanderryn, Costa, and Trubin, 1995).

Figure 24: The Relationship Between Alcohol and Drug Use and the Number of Protective Factors Reported by Washington Students



Summary

In summary, the influence of the risk and protective factors on key health risk behaviors among adolescents is once again supported by the reports of Washington students in the 2000 survey effort. In general, the 2000 WSSAHB findings, which are similar to the results reported in both 1995 and 1998, indicate that:

- Of all risk and protective factors, those in the peer-individual domain show the strongest relationships with health risk behaviors.
- Alcohol and other drug use is also strongly correlated with the community risk factors laws and norms favorable to use and perceived availability of drugs and the school risk factor little commitment to school.
- The cumulative effect of risk and protective factors on alcohol and other drug use is evident among Washington students. Students at high risk on a larger number of risk factors were increasingly more likely to use alcohol

and other drugs, whereas students possessing a larger number of protective factors were increasingly less likely to use alcohol and other drugs.

The data presented in this chapter represent the state as a whole. The level of these indicators of risk and protection likely vary by community and the results for any given community can show the extent to which students in that community experience risk and protective factors. Communities can compare community-level data to state-level data, and to county-level data where available, to determine which risk and protective factors are priorities for their communities to address. These communities can then target specific populations or geographical areas where risk exposure is high and protection is low for intensive interventions.

Chapter 6: Characteristics of the Students Surveyed

The 2000 findings of the Washington State Survey of Adolescent Health Behaviors presented in this report are based on the responses of 17,970 students in Grades 6, 8, 10, and 12. These students were selected using a scientific sampling plan designed by the authors to represent the full population of students at these grade levels across the state. Students from all geographic regions, urban and rural schools, and students of all races/ethnicities were included in the survey. Full details of the sampling plan and its representativeness of the target population are included in the Technical Report.

This chapter describes the students participating in the survey in terms of a variety of background characteristics and their participation in activities in their schools and communities. The specific characteristics discussed in this chapter are the students' gender and race/ethnicity, the geographic region in which the students lived, the type of school the students attended (urban, rural, or suburban), the number of adults and other children living at the students' homes, and student employment (whether the students held a part-time job).

Gender

Males and females were nearly equally represented at all of the surveyed grade levels. The largest deviation from a 50–50 split occurred at the Grade 8 level, where the sample consisted of 51.9 percent females. This percentage closely matches the distribution of population of students in the state (the largest difference was at the Grade 8 level, where females were overrepresented by 3.4 percent). Gender is an important characteristic to consider when interpreting the incidence and prevalence of health behaviors.

Geographic Region

For the purpose of sampling, Washington was divided into the four geographic regions displayed in Figure 25. The eastern region included 20 counties and approximately 24 percent of the student population in the state. The southwest region included 11 counties and approximately 22 percent of the student population. The Puget Sound region included 3 counties and approximately 37 percent of the student population. The student population. The northwest region included 5 counties and approximately 16 percent of the student population.



Figure 25: Geographic Regions

Because of different survey return rates, survey participation represented a higher proportion of students in some regions, and a lower proportion in other

regions, than the sampling plan originally specified. Among Grade 6 students the eastern, southwest, and Puget Sound regions were underrepresented and the northwest region was overrepresented. Among Grade 8 and Grade 10 students the eastern and Puget Sound regions were underrepresented and the northwest region was overrepresented. Among Grade 12 students the southwest and Puget Sound regions were underrepresented and the northwest region was overrepresented. The responses of students from each region were weighted to achieve representation by region within about 1 percent of the actual population statistics. Table 19 details these proportions.

Table 19:Actual and Weighted Proportions of Survey Participants by Geographic Region

	East			S	Southwest			Puget Sound			Northwest		
Grad e	Actual	Weighte d	State										
6	20.8	23.7	24.2	14.6	21.8	22.0	37.8	38.5	37.4	26.9	16.0	16.4	
8	19.8	24.6	24.4	25.4	21.5	22.2	26.3	38.0	37.4	28.4	15.9	16.0	
10	19.8	24.6	24.0	21.4	22.2	22.5	31.9	37.8	37.0	26.8	15.4	16.5	
12	21.9	24.8	24.7	18.9	37.7	36.7	31.7	15.1	15.4	27.6	15.1	15.4	

Note. Figures represent percentage of survey participants.

Urban/ Suburban/Rural School

Three categories describing the urbanity or rurality of the participating schools were used to draw the sample. Urban schools were located in large urban centers and smaller cities that are urban in nature but with more modest population size. For the 2000 WSSAHB administration, schools in these cities accounted for about 26 percent of Washington's student population. Suburban schools were located in areas near large cities or were small cities. These schools accounted for about 37 percent of the student population. Rural schools were from identifiable communities away from urban areas with smaller populations than small cities or from areas with low population density. Schools in this category included about 37 percent of the student population. Figure 26 illustrates the percentages of students in the sample by the rural/urban characteristics of their schools. At the Grade 6 level students in urban schools were overrepresented and students in suburban schools were underrepresented, whereas the opposite was true at the Grade 8 level. At the Grade 10 and Grade 12 levels, students in urban schools were overrepresented and students in rural schools were underrepresented. The weighting procedure brought the sampled proportions in line with the population proportions for Grade 8 students, but suburban Grade 6 students were somewhat overrepresented and rural high school students were somewhat overrepresented.




Race/Ethnicity

Because of the sponsoring agencies' interest in estimating prevalence rates for different ethnic groups, schools were divided into high and low minority enrollment for the purpose of sampling. The survey asked the students to identify which racial/ethnic group they considered themselves to belong to. Two questions were posed to the students. The first question asked students to select a single category based on those used by the federal Department of Education Office of Civil Rights (and thus used by OSPI). The second question was based on the federal census categories and allowed students to select as many groups as they wished. Figure 27 shows the proportion of students who reported themselves as being in the five traditional categories.



Figure 27: Survey Respondents' Race and Ethnicity

Consistent with the sampling design, White, non-Hispanic students were somewhat underrepresented and members of other ethnic groups were slightly overrepresented. The weighting procedure brought the sample proportions close to the actual population proportions, although White, non-Hispanic students remained a little underrepresented at the Grade 6 and Grade 8 levels.

Adults and Other Children at Home

Most (86 to 90 percent) of the students who completed the survey reported living with their mother, and about two-thirds (67 to 71 percent) reported living with their father. About one in eight students (11 to 13 percent) reported living with a stepfather. Figure 28 illustrates the number of other children the survey respondents reported as living at home. Older students were less likely than younger students to report that no other children lived with them at home (39.7 percent of Grade 12 students compared to 14.9 percent of Grade 6 students). The influence of siblings and peers on many of the health behaviors under study in this survey is well documented. Like peers, older siblings can be models of prosocial behavior or they can influence younger children to engage in risk or problem behaviors.



Figure 28: Number of Children at Home

Working at a Part-Time Job

Having a part-time job represents important responsibility to a young person. Previous research indicates that a moderate amount of part-time work is associated with lower alcohol, tobacco, and other drug use (Johnston et al., 1994). Working too many hours, however, is associated with emotional stress and problem behaviors (Resnick et al., 1997). Figure 29 displays the distribution of the number of hours the students reported typically worked per week during the school year.

Figure 29: Hours per Week Worked at a Part-Time Job



The percentage of students who worked at a part-time job increased as students got older. One in five Grade 8 students worked at a part-time job, compared to three in five Grade 12 students.

Chapter 7: Conclusion

The 2000 administration of the Washington State Survey of Adolescent Health Behaviors continued the collaborative tradition of state agencies assessing the health of youth throughout the state. Sponsoring agencies included the Office of Superintendent of Public Instruction, the Division of Alcohol and Substance Abuse, the Office of Community Development, and the Department of Health. RMC Research Corporation conducted the survey. This survey was the sixth of its kind in the state since 1988 and the results in this report charted trends in health behaviors and related risk and protective factors over the past 12 years. The number of schools and students participating in the survey has increased substantially for each of the past three administrations.

Alcohol continues to be the substance of choice among Washington's students, followed by smoking tobacco and marijuana. This finding is true for both lifetime prevalence and 30-day use. The average age of first use of alcohol, tobacco, and marijuana did not change from 1998 to 2000. After an essentially steady increase from 1992 to 1998, binge drinking decreased among students in Grades 6, 8, and 10 and leveled off among students in Grade 12. Current use of cigarettes remained steady from 1998 to 2000. Smokeless tobacco use in the past 30 days decreased from 1998 to 2000 among students in Grades 8, 10, and 12.

Most students understand that secondhand smoke is harmful. Grade 6 students were much more likely than older students to report that they had practiced in class ways to say no to tobacco during the past year. About two-thirds of Grade 6 students and about three fourths of older students reported that they had seen or heard antismoking ads at least once a week during the past 30 days. Marijuana use in the past 30 days decreased from 1998 to 2000 among Grade 8, 10, and 12 students. This change was associated with a continued increase in the percentages of students who thought there is great risk in smoking marijuana. A similar result was found for binge drinking.

Grade 6 students showed a steady decline from 1992 to 1998 in weapon carrying for self-protection or because they thought they might need a weapon in a fight, but Grade 6 students were not asked this question in 2000. Grade 8 students showed a decrease in weapon carrying from 1992 to 1995 and then again from 1998 to 2000. Grade 10 and Grade 12 students reported decreased weapon carrying from 1992 to 1995, but a level prevalence of weapon carrying since then (although the results for the Grade 10 students suggest a small continued decline).

Most students generally feel safe in school. In particular, students reported feeling that the classroom is a safe area. Students were more likely to report feeling unsafe in the bathrooms and on the playground or school grounds. In general, younger students were less likely than older students to feel safe at school. Many students reported having experienced bullying at school, although most did not. Younger students were more likely than older students to report that they had experienced bullying at school. The percentage of students who reported carrying a weapon on school property was quite low, although this behavior does occur.

This report reaffirmed the relationship between substance use and risk and protective factors. In particular, the risk factors perceived availability of drugs, laws and norms favorable toward drug use, little commitment to school and alcohol use were associated with negative health behaviors. Early initiation of drug use, attitudes favorable toward drug use, and friends' use of drugs were associated with substance use. This report also reaffirmed the clear relationship between the number of risk and protective factors present and the prevalence of lifetime and 30-day alcohol and other drug use.

The 2000 WSSAHB is part of an ongoing effort to assess the health of youth throughout Washington State. The results of the survey will be used by stakeholders at the state, county, district, school, and community levels who are interested in developing and improving prevention and intervention programs to better the lives of youth. American Psychiatric Association. (1994). Diagnostic and statistical manual of mental disorders (4th ed.). Washington, DC: Author.

- Arthur, M.W., Hawkins, J.D., Catalano, R.F., and Pollard, J.A. (1998). Student survey of risk and protective factors and prevalence of alcohol, tobacco, and other drug use. Seattle, WA: Social Development Research Group.
- Bachman, J.G., Johnston, L.D., and O'Malley, P.M. (1998). Explaining recent increase in students' marijuana use: Impacts of perceived risks and disapproval, 1976 through 1996. American Journal of Public Health, 88(6), 887–892.
- Benard, B.L. (1991). Fostering resiliency in kids: Protective factors in the family, school, and community. San Francisco: Far West Laboratory for Educational Research and Development.
- Bensley, L.S. and Van Eenwyk, J. (1995). Youth violence and associated risk factors: An epidemiologic view of the literature. Olympia, WA: Washington State Department of Health, Office of Epidemiology.
- Brandeis University, Schneider Institute for Health Policy. (2001, February). Substance abuse: The nation's number one health problem. Key indicators for policy. Princeton, NJ: Robert Wood Johnson Foundation.
- Brewer, D.D., Hawkins, J.D., Catalano, R.F., and Neckerman, H.J. (1995).
 Preventing serious, violent, and chronic juvenile offending. In J.C. Howell,
 B. Krisberg, J.D. Hawkins, and J.J. Wilson (Eds.), A sourcebook: Serious,
 violent, and chronic juvenile offenders (pp. 61–141). Thousand Oaks, CA:
 Sage.

Bry, B.H., McKeon, P., and Pandina, R.J. (1982). Extent of drug use as a function of number of risk factors. *Journal of Abnormal Psychology*, 91, 273–279.

Bush, G.W. (2001, May). No child left behind. http://www.whitehouse.gov/news/reports/no-child-left-behind.html

- California Environmental Protection Agency. (1997). Health effects of exposure to environmental tobacco smoke: Final report. Sacramento, CA: California Environmental Protection Agency, Office of Environmental Health Hazard Assessment.
- Center for Substance Abuse Treatment. (2000, September). Effective treatments emerge for adolescent marijuana use. http://www.health.org/newsroom/releases/2000/sept00/3.htm
- Center on Addiction and Substance Abuse. (1994). Cigarettes, alcohol, and marijuana: Gateways to illicit drug use. NY: Columbia University.
- Center on Addiction and Substance Abuse. (1997). Back to school 1997— National survey of American attitudes on substance abuse: Teen and their parents, teachers, and principals. NY: Author.
- Centers for Disease Control and Prevention. (1994). Medical care expenditures attributable to cigarette smoking: United States, 1993. Morbidity and Mortality Weekly Report, 43(26), 469–472.
- Centers for Disease Control and Prevention. (1997). Medical care expenditures attributable to cigarette smoking during pregnancy: United States, 1995. *Morbidity and Mortality Weekly Report, 46*, 1048–1050.

Centers for Disease Control and Prevention. (1999). 1999 Youth risk behavior surveillance—National Alternative High School Youth Risk Behavior Survey: United States, 1998. http://www.cdc.gov/mmwr/preview/mmwrhtml/ss4807al.htm

Centers for Disease Control and Prevention. (2000). Youth tobacco surveillance: United States, 1998–1999. http://www.cdc.gov/mmwr/preview/mmwrhtml/ss4910al.htm

- Deck, D.D. and Nickel, P.N. (1989). Substance abuse among public school students in Washington. Olympia, WA: Office of Superintendent of Public Instruction.
- DeWit, D.J., Silverman, G., Goodstadt, M., and Stoduto, G. (1995). The construction of risk and protective factor indices for adolescent alcohol and other drug use. *The Journal of Drug Issues, 25*(4), 837–863.
- Einspruch, E.L., Gabriel, R.M., Deck, D.D., and Nickel, P.N. (1998). Washington State Survey of Adolescent Health Behaviors: Analytic report. Olympia, WA: Office of Superintendent of Public Instruction.
- Einspruch, E.L. and Pollard, J.P. (1993). Washington State Survey of Adolescent Health Behaviors: 1988–90. Olympia, WA: Office of Superintendent of Public Instruction.
- Gabriel, R.M. (1991). Substance abuse among public school students in Washington State: 1988–1990. Olympia, WA: Office of Superintendent of Public Instruction.
- Gabriel, R.M., Deck, D.D., Einspruch, E.L., and Nickel, P.N. (1995). The findings of the Washington State Survey of Adolescent Health Behaviors: Analytic report. Olympia, WA: Office of Superintendent of Public Instruction.

- Gabriel, R.M., Deck, D.D., Einspruch, E.L., and Nickel, P.N. (1997). *Risk and* protective factors associated with alcohol, tobacco, and other drug use and violence. Olympia, WA: Office of Superintendent of Public Instruction.
- Glantz, S.A. and Parmely, W.W. (1995) Passive smoking and heart disease: Mechanism and risk. *Journal of the American Medical Association, 273,* 1047–1053.
- Grant, B.F. and Dawson, D.A. (1997). Age of onset of alcohol use and its association with DSM-IV alcohol abuse and dependence: Results from the National Longitudinal Alcohol Epidemiologic Survey. *Journal of Substance Abuse, 9,* 103–110.
- Greenfeld, L.A. (1998). Alcohol and crime: An analysis of national data on the prevalence of alcohol involvement in crime. Washington, DC: U.S. Department of Justice.
- Harwood, H., Fountain, D., and Livermore, G. (1998). The economic costs of alcohol and drug abuse in the United States, 1992 (NIH Publication No. 98-4327). Rockville, MD: National Institutes of Health.
- Hawkins, J.D., Catalano, R.F., Jr., Barnard, K.E., Gottfredson, G.D., Holmes, A.B., Miller, J.Y. (1992). Communities that care: Action for abuse prevention. San Francisco: Jossey Bass.
- Institute of Medicine, Committee on Prevention of Mental Disorder. (1994). In P.H. Mrazek and R.J. Haggerty (Eds.), *Reducing risks for mental disorders: Frontiers for prevention intervention research*. Washington, DC: National Academy Press.

- Jessor, R., Van den Bos, J., Vanderryn, J., Costa, F.M., and Trubin, M.S. (1995). Protective factors in adolescent problem behavior: Moderator effects and developmental change. *Developmental Psychology*, 31(6), 923–933.
- Johnston, L.D., O'Malley, P.M., and Bachman, J.G. (1994). National survey results on drug use, the Monitoring the Future Study, 1975–1993. Volume I: Secondary students. Rockville, MD: National Institute on Drug Abuse.
- Johnston, L.D., O'Malley, P.M., and Bachman, J.G. (1997, December 18). Drug use among American teens shows some signs of leveling after a long rise (Press release). Ann Arbor: University of Michigan News and Information Services.
- Keefe, R.S.E. and Harvey, P.D. (1994). Understanding schizophrenia: A guide to the new research on causes and treatment. NY: The Free Press.
- Mercy, J.A. (1993). The public health impact of firearm injuries. American Journal of Preventive Medicine, 9, 8–11.
- National Center for Health Statistics. (1998, January). Injury visits to hospital emergency departments: United States, 1992–95. (DHHS Publication No. PHS 98–1792). Washington, DC: U.S. Government Printing Office.
- National Institute on Alcohol Abuse and Alcoholism. (2000). 10th special report to the U.S. Congress on alcohol and health. Washington, DC: National Institutes for Health.
- National Institute on Drug Abuse. (2001, May). Monitoring the Future: A continuing study of american youth. http://www.monitoringthefuture.org

- Newcomb, M.D., Maddahian, E., and Skager, R. (1987). Substance abuse and psychosocial risk factors among teenagers: Associations with sex, age, ethnicity, and type of school. *American Journal of Drug and Alcohol Abuse, 13, 413–433*.
- Office of National Drug Control Policy. (1999). 1999 National drug control strategy. http://whitehousedrugpolicy.gov/policy/99ndcs/99ndcs.pdf
- Pirkle, J.L., Flegal, K.M., Bernet, J.T., Brody, D.J., Etzel, R.A., and Maurer, K.R. (1996). Exposure of the U.S. population to environmental tobacco smoke. *Journal of the American Medical Association, 275,* 1233–1240.
- Resnick, M., Bearman, P.S., Blum, R.W., Bauman, K.E., Harris, K.K., Jones, J., Tabor, J., Beuhring, T., Sieving, R.E., Shew, M., Ireland, M., Bearinger, L.H., and Udry, J.R. (1997). Protecting adolescents from harm: Findings from the National Longitudinal Study on Adolescent Health. *Journal of the American Medical Association, 278*(10), 823–832.
- Rutter, M. (1979). Protective factors in children's responses to stress and disadvantage. In M.W. Kent and J.E. Rolf (Eds.), Primary prevention of psychopathology, Vol. 3. Social competence in children (pp. 49–74). Hanover, NH: University Press of New England.
- U.S. Department of Education. (1998, May). Safe and Drug-Free Schools and Communities Act—State grants for drug and violence prevention program: Nonregulatory guidance for implementing the SDFSCA principles of effectiveness. Washington, DC: Author.
- U.S. Department of Health and Human Services. (1986). The health consequences of involuntary smoking: A report of the Surgeon General. (DHHS Publication No. PHS 87–8398). Washington, DC: U.S. Government Printing Office.

- U.S. Department of Health and Human Services. (2000a, January). Healthy People 2010: Understanding and Improving Health (Conference edition). Washington, DC: Author.
- U.S. Department of Health and Human Services. (2000b, November). Healthy People 2010. Vol. 2. Washington, DC: Author.
- U.S. Environmental Protection Agency. (1992). Respiratory health effects of passive smoking: Lung cancer and other disorders (EPA Publication No. EPA/600/6-90/006F). Washington, DC: Author.
- Washington State Board of Health. (1998, March). Washington State Public Health Report 1998. Olympia, WA: Author.
- Washington State Commission on Student Learning. (1998, August). Essential academic learning requirements: Technical manual. Olympia, WA: Author.
- Werner, E. and Smith, R. (1989). Vulnerable but invincible: A longitudinal study of resilient children and youth. New York: Adams, Bannister, and Cox.
- Wickizer, T.M. (1999, March). The economic costs of drug and alcohol abuse in Washington State, 1996. Seattle: University of Washington.
- Wickizer, T.M., Wagner, T., Atherly, A., and Beck, M. (1993). The economic costs of drug and alcohol abuse in Washington State, 1990. Seattle: University of Washington, School of Public Health and Community Medicine, Department of Health Services.

Appendix A Item-Level Frequencies

1. How old are you?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,302)	(n = 4,962)	(n = 4,811)	(n = 3,747)
10 or younger	0.5%	0.0%	0.0%	0.0%
11	65.2	0.0	0.0	0.0
12	33.3	0.7	0.0	0.0
13	0.9	64.2	0.0	0.0
14	0.0	33.6	0.5	0.0
15	0.0	1.4	60.8	0.0
16	0.0	0.0	36.1	0.7
17	0.0	0.0	2.2	60.7
18	0.0	0.0	0.4	35.5
19 or older	0.0	0.0	0.0	3.0

[Item 2 was a check on student's grade level.]

3. Are you:

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,260)	(n = 4,778)	(n = 4,776)	(n = 3,686)
Female	50.0%	51.9%	51.6%	49.5%
Male	50.0	48.1	48.4	50.5

[Item 4 asked for student's zip code.]

5. Not counting you, how many other children under age 18 live in your home?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,274)	(n = 4,922)	(n = 4,804)	(n = 3,734)
None	14.9%	19.5%	31.0%	39.7%
1	41.4	41.5	36.4	33.3
2	24.9	22.8	19.9	16.2
3	10.9	9.3	7.2	5.8
4 or more	7.9	6.9	5.4	5.0

6. What do you consider yourself to be? (Choose one or more.)

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,312)	(n = 4,980)	(n = 4,820)	(n = 3,758)
Black or African American, not of Hispanic origin	5.8%	6.4%	3.9%	4.1%
White, not of Hispanic origin	65.8	70.6	75.6	77.0
American Indian or Alaskan Native	7.1	7.4	5.0	4.1
Mexican American	6.4	4.8	5.0	3.7
Mexican	3.0	3.4	3.5	2.6
Puerto Rican	0.9	0.8	0.7	0.6
Other Latino/Hispanic	2.2	2.7	2.7	2.3
Asian American	4.0	3.8	2.7	2.7
Chinese	1.9	1.9	1.4	1.2
Japanese	1.4	1.2	1.1	1.2
Filipino	2.1	2.1	2.8	2.1
Vietnamese	1.2	1.4	1.2	1.5
Korean	1.3	1.5	1.3	1.2
Samoan	0.5	0.6	0.6	0.4
Native Hawaiian	1.0	1.1	0.6	0.6
Other Asian or Pacific Islander	1.5	1.5	1.7	1.6
Other	14.4	10.5	6.9	5.7

7. Now, if you could only choose one, what do you consider yourself to be? (Choose one best ar	iswer.)
--	---------

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,007)	(n = 4,824)	(n = 4,692)	(n = 3,644)
Black or African American, not of Hispanic origin	6.6%	6.6%	4.0%	3.8%
Asian or Pacific Islander	8.8	8.7	7.6	8.2
American Indian or Alaskan Native	7.3	4.8	2.7	2.5
Hispanic	8.3	8.6	9.9	7.3
White, not of Hispanic origin	69.1	71.3	75.8	78.3

8.	Which of the following adults are living with you now	? (Choose al	l that apply.)

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,312)	(<i>n</i> = 4,980)	(n = 4,820)	(n = 3,758)
Mother	90.9%	89.9%	89.3%	86.3%
Father	71.0	69.1	67.4	68.2
Stepmother	4.2	4.4	4.7	4.8
Stepfather	11.4	13.2	13.2	11.8
Grandmother	5.9	6.0	3.9	3.1
Grandfather	3.3	2.9	2.3	1.6
Aunt	3.1	2.4	1.7	2.3
Uncle	3.3	3.0	2.0	1.8
Foster mother	0.3	0.5	0.3	0.3
Foster father	0.3	0.3	0.3	0.3
Adult brother	5.5	7.5	8.4	7.8
Adult sister	5.1	6.0	6.2	5.9
Other adults	2.6	3.2	2.6	3.7
None of the above	1.0	0.5	0.6	2.3

During the school year, how many hours a week do you work at a part-time job? 9.

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(<i>n</i> = 4,911)	(n = 4,784)	(n = 3,726)
I don't work	*	79.2%	74.3%	39.4%
1–4 hours	*	13.0	9.2	6.8
5–9 hours	*	4.5	6.6	8.9
10–20 hours	*	1.7	7.4	30.3
Over 20 hours	*	1.5	2.5	14.6
What language is usually spoken at home?				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 185)	(n = 4,825)	(n = 4,728)	(n = 3,692)
English	91.2%	92.5%	91.0%	91.4%
Spanish	3.7	3.0	4.6	3.1
Other	5.1	4.6	4.3	5.5
I like my neighborhood.				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,222)	(<i>n</i> = 4,910)	(n = 4,772)	(n = 3,722)
NO!	3.7%	4.2%	5.9%	3.8%
no	8.8	9.3	10.6	10.4
yes	50.7	53.5	56.7	54.8

26.8

31.0

33.0

36.8

10.

11.

yes YES!

12.	If I had to move, I would miss the neighborhood I	now live in.			
	-	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 4, 181)	(n = 4,867)	(n = 4,752)	(<i>n</i> = 3,696)
	NO!	8.6%	9.7%	12.7%	12.9%
	no	17.7	22.3	28.1	30.3
	yes	35.5	37.1	36.2	35.5
	YES!	38.2	30.9	22.9	21.4
13.	I'd like to get out of my neighborhood.				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 4, 156)	(n = 4,855)	(n = 4,727)	(n = 3,686)
	NO!	45.7%	35.8%	25.3%	22.4%
	no	33.8	40.8	44.9	42.1
	yes	13.2	14.5	19.2	24.0
	YES!	7.3	8.9	10.7	11.5
14.	I feel safe in my neighborhood.				
	, ,	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 4, 184)	(n = 4,868)	(n = 4,741)	(n = 3,700)
	NO!	5.5%	4.7%	3.9%	2.5%
	No	8.6	8.3	6.9	6.2
	Yes	41.0	45.8	48.9	45.2
	YES!	44.9	41.2	40.3	46.1

15. How wrong would most adults in your neighborhood think it was for kids your age:

A. To use marijuana

В.

C.

10 use manjaana	· ·			
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,166)	(n = 4,904)	(n = 4,765)	(n = 3,713)
Very wrong	88.2%	74.7%	62.4%	56.2%
Wrong	8.0	16.7	24.7	27.4
A little bit wrong	2.4	6.3	9.7	11.5
Not wrong at all	1.4	2.2	3.2	4.9
To drink alcohol				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 153)	(n = 4,889)	(n = 4,766)	(n = 3,703)
Very wrong	75.1%	58.3%	40.7%	32.6%
Wrong	17.5	27.9	35.9	37.5
A little bit wrong	5.6	10.5	18.2	23.3
Not wrong at all	1.8	3.3	5.2	6.7
To smoke cigarettes				
Ç	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 169)	(n = 4,895)	(n = 4,771)	(n = 3,699)
Very wrong	74.6%	57.6%	39.4%	25.9%
Wrong	15.1	26.1	32.8	33.5
A little bit wrong	7.1	11.6	20.3	27.4
Not wrong at all	3.1	4.7	7.5	13.2

16. About how many adults have you known personally who in the past year have:

A. Used marijuana, crack, cocaine, or other drugs?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 160)	(n = 4,883)	(n = 4,768)	(n = 3,730)
None	66.8%	55.3%	45.4%	36.8%
1 adult	14.5	15.8	15.3	12.4
2 adults	7.1	9.1	11.1	11.9
3 or 4 adults	5.7	8.7	10.7	14.0
5 or more adults	6.0	11.1	17.5	25.0
Sold or dealt drugs?				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 113)	(n = 4,871)	(n = 4,760)	(n = 3,730)
None	86.2%	74.0%	64.3%	53.8%
1 adult	7.6	12.0	13.4	13.0
2 adults	3.1	5.4	7.9	12.5
3 or 4 adults	1.5	4.0	6.4	8.8
5 or more adults	1.6	4.6	8.0	12.0

C. Done other things that could get them in trouble with the police like stealing, selling stolen goods, mugging or assaulting others, etc.?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 103)	(n = 4,874)	(n = 4,738)	(n = 3,710)
None	80.2%	69.3%	64.7%	57.5%
1 adult	11.3	13.9	12.2	12.1
2 adults	4.3	6.0	7.0	9.8
3 or 4 adults	1.9	3.9	5.8	7.2
5 or more adults	2.3	7.0	10.4	13.4

D. Gotten drunk or high?

Β.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 121)	(n = 4,890)	(n = 4,773)	(n = 3,730)
None	55.9%	39.4%	27.3%	20.8%
1 adult	22.4	19.1	14.7	10.0
2 adults	8.7	9.9	11.1	9.5
3 or 4 adults	4.8	8.3	10.3	11.9
5 or more adults	8.2	23.3	36.5	47.8

17. If a kid drank some beer, wine, or hard liquor (for example, vodka, whiskey, or gin) in your neighborhood would he or she be caught by the police?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,092)	(n = 4,823)	(<i>n</i> = 4,741)	(<i>n</i> = 3,704)
NO!	11.6%	21.7%	32.5%	39.2%
No	30.4	45.6	52.2	48.7
Yes	34.8	24.2	11.3	8.7
YES!	23.1	8.5	4.0	3.4

18. If a kid smoked marijuana in your neighborhood would he or she be caught by the police?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,085)	(n = 4,786)	(n = 4,731)	(n = 3,692)
NO!	7.9%	18.7%	30.5%	36.7%
No	22.9	38.2	49.0	48.3
Yes	36.2	29.7	15.5	11.2
YES!	33.1	13.5	5.1	3.8

19. If a kid carried a handgun in your neighborhood would he or she be caught by the police?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 105)	(n = 4,774)	(n = 4,712)	(n = 3,662)
NO!	5.9%	13.3%	18.9%	22.1%
No	13.6	26.5	36.8	39.1
Yes	30.4	33.4	29.2	25.4
YES!	50.0	26.8	15.1	13.4

20. If you wanted to get some beer, wine, or hard liquor (for example, vodka, whiskey, or gin), how easy would it be for you to get some?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 142)	(n = 4,872)	(n = 4,764)	(n = 3,726)
Very hard	62.2%	30.9%	11.9%	6.7%
Sort of hard	19.1	23.4	17.0	10.0
Sort of easy	11.0	23.5	31.5	28.7
Very easy	7.7	22.2	39.6	54.5

21. If you wanted to get some cigarettes, how easy would it be for you to get some?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 126)	(n = 4,864)	(n = 4,762)	(n = 3,724)
Very hard	61.0%	31.6%	13.3%	4.4%
Sort of hard	16.8	20.5	14.5	5.2
Sort of easy	11.0	18.1	23.1	11.1
Very easy	11.2	29.7	49.0	79.2

22. If you wanted to get some marijuana, how easy would it be for you to get some?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,097)	(n = 4,840)	(n = 4,739)	(<i>n</i> = 3,690)
Very hard	86.6%	53.2%	22.2%	12.5%
Sort of hard	6.6	16.1	16.5	13.7
Sort of easy	4.2	14.8	25.2	27.9
Very easy	2.6	15.9	36.1	45.9

23. If you wanted to get a drug like cocaine, LSD, or amphetamines, how easy would it be for you to get some?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 101)	(n = 4,826)	(n = 4,731)	(<i>n</i> = 3,670)
Very hard	90.2%	72.2%	43.2%	29.5%
Sort of hard	6.0	16.3	27.8	30.9
Sort of easy	2.3	7.4	19.0	24.0
Very easy	1.5	4.1	10.0	15.6

24. If you wanted to get a handgun, how easy would it be for you to get one?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 109)	(n = 4,838)	(n = 4,723)	(n = 3,677)
Very hard	77.3%	64.3%	50.6%	39.2%
Sort of hard	12.7	17.8	24.1	28.2
Sort of easy	5.2	9.0	14.0	15.9
Very easy	4.9	8.9	11.3	16.7

25. There are adults in my neighborhood I could talk to about something important.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,240)	(n = 4,879)	(n = 4,760)	(n = 3,714)
NO!	7.3%	11.1%	13.5%	12.9%
No	9.2	12.1	14.4	16.1
Yes	28.2	31.8	34.2	31.5
YES!	55.3	45.0	37.9	39.6

26. Which of the following activities do you regularly participate in?

A. Sports teams

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 187)	(n = 4,910)	(n = 4,787)	(n = 3,721)
Yes	72.0%	66.8%	61.8%	53.4%
No, even though this activity is available	22.8	30.6	35.7	43.9
No, because this activity is not available	5.2	2.5	2.5	2.7

Scouting (such as Cub Scouts, Boy Scouts, Girl Scouts, Brownies, Camp Fire Girls, etc.) Β.

Scouling (such as Cub Scouls, Boy Scouls, Girl	Scouts, Brownies,	Camp Fire Gi	ris, etc.)	
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,054)	(n = 4,790)	(n = 4,726)	(n = 3,678)
Yes	19.4%	8.6%	6.3%	5.9%
No, even though this activity is available	69.3	79.1	81.9	82.1
No, because this activity is not available	11.2	12.4	11.8	12.0

C. Arts groups (such as art, music, drama, dance, etc.)

This groups (such as art, music, drama, dance, etc.)				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,069)	(n = 4,789)	(n = 4,728)	(<i>n</i> = 3,689)
Yes	50.2%	44.8%	38.0%	36.6%
No, even though this activity is available	38.2	47.9	57.2	58.5
No, because this activity is not available	11.5	7.3	4.7	4.8

D. Service or social clubs (such as Boys and Girls Clubs, 4-H Clubs, church youth groups, etc.)

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,059)	(n = 4,824)	(n = 4,744)	(n = 3,696)
Yes	41.4%	37.2%	38.9%	39.3%
No, even though this activity is available	47.5	55.8	56.6	56.7
No, because this activity is not available	11.1	7.1	4.5	4.0

27. My neighbors notice when I am doing a good job and let me know.

	Grade 6 $(n = 4.234)$	Grade 8 $(n = 4.898)$	Grade 10 $(n = 4.779)$	Grade 12 $(n = 3.710)$
NO!	17.2%	26.3%	29.3%	32.3%
No	24.6	29.5	34.9	34.3
Yes	37.2	30.8	26.7	24.6
YES!	20.9	13.5	9.1	8.7

28. There are people in my neighborhood who encourage me to do my best.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,225)	(n = 4,887)	(n = 4,774)	(n = 3,712)
NO!	12.9%	20.3%	24.8%	28.5%
No	18.3	24.1	25.7	25.9
Yes	35.1	33.7	34.1	30.7
YES!	33.7	21.9	15.3	14.9

There are people in my neighborhood who are proud of me when I do something well. 29. Grade 6 Grade 8

	1 1	-	0	1		0		
					Grade 6	Grade 8	Grade 10	Grade 12
					(n = 4,212)	(n = 4,883)	(n = 4,776)	(n = 3,701)
NO!					13.1%	20.0%	23.6%	26.1%
no					17.0	22.3	23.9	23.9
yes					37.2	36.2	36.3	33.9
YES!					32.8	21.5	16.2	16.1

During the past 30 days, on how many days did you:

30. Smoke cigarettes.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,275)	(n = 4,960)	(n = 4,804)	(n = 3,747)
None	96.0%	87.5%	80.2%	72.4%
1–2 days	2.3	5.0	5.3	6.6
3–5 days	0.5	2.1	2.6	2.3
6–9 days	0.3	1.2	1.4	2.2
10 or more days	1.0	4.2	10.5	16.6
Any use in past 30 days	4.0	12.5	19.8	27.6

31. Smoke cigars, cigarillos, or little cigars.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,270)	(<i>n</i> = 4,941)	(n = 4,796)	(n = 3,742)
None	98.5%	95.7%	92.1%	86.9%
1–2 days	1.0	2.3	4.4	7.8
3–5 days	0.1	0.6	0.7	2.1
6–9 days	0.1	0.2	0.9	0.9
10 or more days	0.4	1.1	1.9	2.3
Any use in past 30 days	1.5	4.3	7.9	13.1

32. Smoke tobacco in a pipe.

34.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,220)	(n = 4,921)	(n = 4,775)	(n = 3,726)
None	99.4%	97.9%	98.1%	98.3%
1–2 days	0.3	1.0	1.1	0.9
3–5 days	0.1	0.4	0.2	0.2
6–9 days	0.1	0.1	0.2	0.3
10 or more days	0.1	0.6	0.4	0.2
Any use in past 30 days	0.6	2.1	1.9	1.7

Smoke bidis ('beedies,' flavored cigarettes). 33.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,244)	(n = 4,905)	(n = 4,760)	(n = 3,724)
None	99.0%	96.7%	95.4%	93.5%
1–2 days	0.5	1.9	3.1	4.3
3–5 days	0.2	0.5	0.5	0.9
6–9 days	0.1	0.5	0.3	0.3
10 or more days	0.2	0.5	0.7	1.0
Any use in past 30 days	1.0	3.3	4.6	6.5
Chew tobacco or use snuff.				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,235)	(n = 4,838)	(n = 4,689)	(n = 3,668)
None	99.2%	97.9%	95.4%	91.2%
1–2 days	0.5	1.0	1.8	3.0
3–5 days	0.1	0.4	0.6	1.3
6–9 days	0.0	0.2	0.4	0.4
10 or more days	0.2	0.4	1.7	4.1
Any use in past 30 days	0.8	2.1	4.6	8.8

35.	Drink a glass, can or bottle of alcohol (beer, wine, wine coolers, hard liquor).					
	-	Grade 6	Grade 8	Grade 10	Grade 12	
		(n = 4,252)	(n = 4,907)	(n = 4,759)	(n = 3,726)	
	None	93.4%	77.7%	62.4%	53.2%	
	1-2 days	4.9	12.5	18.8	21.5	
	3–5 days	0.7	4.0	7.9	11.7	
	6–9 days	0.3	2.3	4.7	6.0	
	10 or more days	0.7	3.5	6.2	7.6	
	Any use in past 30 days	6.6	22.3	37.6	46.8	
36.	Use marijuana or hashish (grass, hash, pot).					
		Grade 6	Grade 8	Grade 10	Grade 12	
		(n = 4,250)	(<i>n</i> = 4,896)	(n = 4,758)	(n = 3,723)	
	None	98.5%	88.0%	78.1%	75.6%	
	1–2 days	0.7	5.1	7.2	8.4	
	3–5 days	0.3	2.1	3.9	4.3	
	6–9 days	0.1	1.5	2.6	2.6	
	10 or more days	0.4	3.3	8.1	9.2	
	Any use in past 30 days	1.5	12.0	21.9	24.4	
37.	Use cocaine or crack (coke, rock, snow).					
		Grade 6	Grade 8	Grade 10	Grade 12	
		(<i>n</i> = *)	(n = 4,885)	(n = 4,740)	(<i>n</i> = 3,716)	
	None	*	98.5%	97.4%	97.2%	
	1–2 days	*	0.9	1.1	1.4	
	3–5 days	*	0.2	0.7	0.4	
	6–9 days	*	0.1	0.3	0.4	
	10 or more days	*	0.3	0.6	0.6	
	Any use in past 30 days	*	1.5	2.6	2.8	
38.	Use inhalants (things you sniff to get high).					
		Grade 6	Grade 8	Grade 10	Grade 12	
		(n = 4,248)	(n = 4,894)	(n = 4,752)	(n = 3,721)	
	None	98.6%	95.1%	96.4%	97.6%	
	1–2 days	0.8	2.8	2.0	1.5	
	3–5 days	0.3	0.9	0.6	0.3	
	6–9 days	0.2	0.5	0.4	0.3	
	10 or more days	0.2	0.6	0.7	0.4	
	Any use in past 30 days	1.4	4.9	3.6	2.4	
39.	Use psychedelics (angel dust, LSD, acid, microdot	, PCP, magic m	ushrooms).			
		Grade 6	Grade 8	Grade 10	Grade 12	
		(n = 4,254)	(n = 4,904)	(n = 4,764)	(n = 3,718)	
	None	99.4%	96.9%	94.2%	93.5%	
	1–2 days	0.3	1.6	3.3	4.0	
	3–5 days	0.2	0.7	1.2	1.6	
	6–9 days	0.0	0.3	0.4	0.5	
	10 or more days	0.1	0.4	0.9	0.3	
	Any use in past 30 days	0.6	3.1	5.8	6.5	

40. Use party drugs (rave, ecstasy, MDMA).

	Grade 6 $(n = 4,215)$	Grade 8 $(n = 4.891)$	Grade 10 $(n = 4,764)$	Grade 12 $(n = 3,723)$
None	99.3%	96.6%	93.8%	93.2%
1–2 days	0.4	1.9	3.5	3.8
3–5 days	0.1	0.5	1.3	1.8
6–9 days	0.0	0.5	0.4	0.5
10 or more days	0.1	0.5	1.0	0.8
Any use in past 30 days	0.7	3.4	6.2	6.8

[Item 41 was used to check honesty.]

42. Use heroin.

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(n = 4,861)	(n = 4,722)	(n = 3,707)
None	*	99.2%	99.0%	99.2%
1–2 days	*	0.4	0.6	0.3
3–5 days	*	0.2	0.1	0.2
6–9 days	*	0.1	0.1	0.1
10 or more days	*	0.1	0.2	0.2
Any use in past 30 days	*	0.8	1.0	0.8

43. Use amphetamines of any kind (speed, uppers, meth, bennies, crank). Do NOT include non-prescription or over-the-counter drugs, or drugs prescribed to you by a doctor.

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(n = 4,875)	(n = 4,741)	(n = 3,722)
None	*	97.3%	95.5%	96.0%
1–2 days	*	1.5	2.0	2.0
3–5 days	*	0.6	1.1	0.9
6–9 days	*	0.1	0.5	0.4
10 or more days	*	0.6	0.9	0.8
Any use in past 30 days	*	2.7	4.5	4.0
		=.,	110	

44. Use methamphetamine specifically (meth, crystal meth, ice, crank).

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(n = 4,856)	(n = 4,730)	(n = 3,718)
None	*	98.8%	97.4%	97.1%
1–2 days	*	0.6	1.4	1.4
3–5 days	*	0.2	0.4	0.7
6–9 days	*	0.2	0.2	0.2
10 or more days	*	0.2	0.6	0.6
Any use in past 30 days	*	1.2	2.6	2.9

45. About how many cigarettes have you smoked in your entire life?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,207)	(n = 4,741)	(n = 4,686)	(<i>n</i> = 3,681)
0	85.8%	64.9%	49.9%	40.5%
1	6.7	9.0	9.5	7.4
2–5	4.0	9.1	9.9	10.5
6–15 (about half a pack)	1.3	4.1	5.3	6.0
16–25 (about a pack)	0.7	2.8	3.2	4.0
More than one pack but less than five	0.6	3.5	6.0	7.5
Five or more packs	0.9	6.6	16.1	24.0

Have you ever, even once in your lifetime, used any of the following drugs?

46.	Cocaine	or crack	(coke,	rock,	snow)	
-----	---------	----------	--------	-------	-------	--

		Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = *)	(<i>n</i> = 4,938)	(n = 4,794)	(n = 3,743)
	No	*	96.7%	94.0%	90.8%
	Yes	*	3.3	6.0	9.2
47. Inhaled substances to get high (snappers, poppers, rush, other things you sniff to get high)					

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,221)	(<i>n</i> = 4,945)	(n = 4,797)	(n = 3,744)
No	97.5%	90.4%	88.1%	86.9%
Yes	2.5	9.6	11.9	13.1

Psychedelics (angel dust, LSD, acid, microdot, PCP, magic mushrooms) 48.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,214)	(n = 4,937)	(n = 4,791)	(n = 3,739)
No	99.2%	95.3%	89.3%	84.9%
Yes	0.8	4.7	10.7	15.1

49. Party drugs (rave, ecstasy)

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = 4,194)	(<i>n</i> = 4,926)	(<i>n</i> = 4,787)	(<i>n</i> = 3,742)
No	99.1%	95.2%	90.7%	86.5%
Yes	0.9	4.8	9.3	13.5

[Item 50 was used to check honesty.]

ers)
ers)

	Grade 6 (<i>n</i> = *)	Grade 8 (<i>n</i> = 4,922)	Grade 10 $(n = 4,781)$	Grade 12 (<i>n</i> = 3,735)
No	*	97.8%	97.1%	97.1%
Yes	*	2.2	2.9	2.9

a 1 **a**

a 1 0

C 1 10

52. Heroin

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = *)	(n = 4,898)	(n = 4,772)	(n = 3,724)
No	*	98.6%	98.1%	97.6%
Yes	*	1.4	1.9	2.4

Amphetamines of any kind (speed, uppers, meth, bennies, crank). Do NOT include non-prescription or over-53. the-counter drugs, or drugs prescribed to you by a doctor.

		Grade 6	Grade 8	Grade 10	Grade 12
		(n = *)	(<i>n</i> = 4,916)	(n = 4,768)	(n = 3,740)
	No	*	95.7%	91.6%	90.0%
	Yes	*	4.3	8.4	10.0
54.	Methamphetamine specifically (meth, crystal met	h, ice, crank)			
		Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = *)	(n = 4,905)	(n = 4,774)	(n = 3,736)
	No	*	98.0%	94.7%	92.5%
	Yes	*	2.0	5.3	7.5

55. Have you ever used a needle to inject an illegal drug?

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(n = 4,924)	(n = 4,781)	(n = 3,742)
No	*	99.0%	98.7%	98.5%
Yes	*	1.0	1.3	1.5

56. Think back over the last 2 weeks. How many times have you had five or more drinks in a row? (A drink is a glass of wine, a bottle of beer, a shot glass of liquor, or a mixed drink.)

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,077)	(n = 4,923)	(n = 4,775)	(n = 3,740)
None	95.3%	85.1%	76.8%	68.2%
Once	2.5	5.8	8.2	11.1
Twice	1.0	3.6	6.0	7.5
3–5 times	0.8	3.3	5.3	7.7
6–9 or more times	0.1	1.0	1.6	2.1
10 or more times	0.3	1.3	2.2	3.4

57. If you drink alcohol, how do you usually get the beer, wine, or liquor you drink?

Grade 6	Grade 8	Grade 10	Grade 12
(n = 4,251)	(n = 4,756)	(n = 4,589)	(n = 3,554)
93.0%	74.9%	55.3%	43.8%
3.8	6.5	6.7	6.5
1.2	4.7	4.0	1.5
1.7	12.4	28.1	37.3
0.3	1.5	5.9	11.0
	Grade 6 (n = 4,251) 93.0% 3.8 1.2 1.7 0.3	Grade 6Grade 8 $(n = 4,251)$ $(n = 4,756)$ 93.0% 74.9% 3.8 6.5 1.2 4.7 1.7 12.4 0.3 1.5	Grade 6Grade 8Grade 10 $(n = 4,251)$ $(n = 4,756)$ $(n = 4,589)$ 93.0% 74.9% 55.3% 3.8 6.5 6.7 1.2 4.7 4.0 1.7 12.4 28.1 0.3 1.5 5.9

58. Do you think that smoke from other people's cigarettes (second-hand smoke) is harmful to you?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,223)	(n = 4,929)	(n = 4,785)	(n = 3,741)
Definitely no	5.9%	4.0%	3.0%	3.4%
Probably no	7.5	7.0	4.8	3.7
Probably yes	28.2	27.7	24.5	20.3
Definitely yes	58.3	61.3	67.7	72.5

59. Do you think that you will smoke a cigarette anytime during the next year?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,245)	(n = 4,916)	(n = 4,781)	(n = 3,733)
Definitely no	83.3%	60.8%	56.2%	51.8%
Probably no	10.5	19.0	17.4	14.3
Probably yes	3.9	11.8	11.1	14.1
Definitely yes	2.2	8.4	15.3	19.9

60. If one of your best friends offered you a cigarette, would you smoke it?

,	5	U	,	Grade 6	Grade 8	Grade 10	Grade 12
				(n = 4,239)	(n = 4,895)	(n = 4,783)	(n = 3,732)
Definitely no				85.5%	64.0%	59.1%	56.7%
Probably no				9.7	18.4	16.7	15.2
Probably yes				3.1	10.2	11.0	10.4
Definitely yes				1.8	7.5	13.3	17.6
5 5							

61. Do you think that you will try a cigarette soon?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,253)	(n = 4,932)	(n = 4,776)	(n = 3,720)
No	90.5%	70.7%	59.3%	51.4%
Yes	1.9	3.6	2.5	3.4
I have already tried smoking cigarettes	7.6	25.7	38.1	45.2

62. During the past year, have you practiced in any of your classes ways to say "no" to tobacco (for example, in role plays)?

Grade 6	Grade 8	Grade 10	Grade 12
(n = 4,222)	(n = 4,910)	(n = 4,765)	(n = 3,724)
30.3%	51.4%	72.5%	85.2%
69.7	48.6	27.5	14.8
	Grade 6 (n = 4,222) 30.3% 69.7	Grade 6Grade 8 $(n = 4,222)$ $(n = 4,910)$ 30.3% 51.4% 69.7 48.6	Grade 6Grade 8Grade 10 $(n = 4,222)$ $(n = 4,910)$ $(n = 4,765)$ 30.3% 51.4% 72.5% 69.7 48.6 27.5

63. Have your parents (or guardians) discussed the dangers of tobacco use with you?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,208)	(<i>n</i> = 4,908)	(n = 4,758)	(<i>n</i> = 3,731)
No	22.2%	28.8%	34.2%	36.2%
Yes	77.8	71.2	65.8	63.8

64. Do you think it is safe to smoke for only a year or two, as long as you quit after that?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,231)	(n = 4,913)	(n = 4,764)	(n = 3,725)
Definitely no	75.5%	69.5%	67.7%	64.3%
Probably no	16.2	21.5	23.0	23.6
Probably yes	6.1	7.1	7.1	9.6
Definitely yes	2.2	1.8	2.2	2.5

65. During the past 30 days, how often have you seen anti-smoking commercials on TV, or heard them on the radio?

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = 4,195)	(n = 4,908)	(n = 4,761)	(n = 3,712)
None	16.5%	9.8%	8.4%	9.5%
1–3 times per month	15.8	13.2	14.0	17.3
1–3 times per week	17.8	17.2	20.5	26.6
Daily	21.9	25.8	29.1	25.7
More than once a day	28.1	34.1	28.1	20.9

66. Do you want to stop using tobacco (cigarettes, cigars, chew, dip, or snuff) right now?

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = 4,216)	(<i>n</i> = 4,911)	(n = 4,762)	(n = 3,712)
I don't use tobacco	93.9%	84.5%	77.5%	70.1%
Definitely no	0.9	2.0	3.2	3.8
Probably no	1.1	4.3	6.5	7.0
Probably yes	1.6	5.2	7.5	11.8
Definitely yes	2.4	4.0	5.3	7.2

Some tobacco companies make items like sports gear, t-shirts, lighters, hats, jackets, and sunglasses that people can buy or receive for free.

67. Would you ever use or wear something that has a tobacco company name or picture on it such as a lighter, T-shirt, hat or sunglasses?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,202)	(n = 4,920)	(n = 4,774)	(n = 3,728)
Definitely no	62.6%	45.5%	39.7%	40.4%
Probably no	24.9	29.1	28.9	24.2
Probably yes	10.3	19.5	22.8	25.6
Definitely yes	2.1	5.9	8.6	9.9

68.	During the past 12 months, did you buy or receive a	nything that ha	s a tobacco co	mpany logo or	picture on it?
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 4,217)	(<i>n</i> = 4,909)	(n = 4,765)	(<i>n</i> = 3,718)
	No	91.3%	83.8%	81.3%	80.0%
	Yes	8.7	16.2	18.7	20.0

How much do you think people risk harming themselves if they:

^{69.} Smoke one to five cigarettes per day?

	Grade 6	Grade 8	Grade 10	Grade 12	
	(n = 4, 165)	(n = 4,877)	(n = 4,741)	(n = 3,701)	
No risk	3.7%	2.8%	2.1%	2.6%	
Slight risk	10.3	9.4	8.6	9.5	
Moderate risk	30.6	36.5	39.9	38.8	
Great risk	41.4	42.8	44.0	45.6	
Not sure	14.0	8.4	5.4	3.5	
Smoke one or more packs of cigarettes per day?					

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 153)	(n = 4,874)	(n = 4,728)	(<i>n</i> = 3,695)
No risk	3.6%	2.0%	1.4%	1.9%
Slight risk	2.2	1.7	1.3	1.3
Moderate risk	10.3	9.8	8.2	5.9
Great risk	77.2	82.0	85.1	87.6
Not sure	6.6	4.5	4.0	3.3

71. Try marijuana once or twice?

70.

72.

5 5	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 132)	(<i>n</i> = 4,850)	(n = 4,702)	(<i>n</i> = 3,688)
No risk	6.4%	14.0%	25.3%	34.3%
Slight risk	19.3	28.8	34.2	33.1
Moderate risk	26.9	25.2	20.8	15.8
Great risk	35.0	23.2	15.0	12.8
Not sure	12.4	8.7	4.7	4.0
Smoke marijuana occasionally?				

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 128)	(n = 4,844)	(n = 4,708)	(<i>n</i> = 3,690)
No risk	4.6%	6.5%	11.1%	15.4%
Slight risk	6.2	12.9	21.3	24.7
Moderate risk	25.0	33.1	34.4	33.8
Great risk	53.2	40.3	29.1	22.4
Not sure	11.1	7.2	4.1	3.7

73. Smoke marijuana regularly?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 149)	(n = 4,793)	(n = 4,668)	(n = 3,648)
No risk	4.1%	4.5%	5.9%	6.9%
Slight risk	1.8	4.1	8.5	9.5
Moderate risk	6.3	11.5	16.4	20.1
Great risk	77.0	73.1	64.9	58.9
Not sure	10.7	6.9	4.3	4.6

74. Take one or two drinks of an alcoholic beverage (wine, beer, a shot, liquor) nearly every day?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 126)	(n = 4,874)	(n = 4,733)	(n = 3,702)
No risk	9.5%	10.9%	9.4%	12.6%
Slight risk	23.1	20.6	21.0	21.9
Moderate risk	28.5	33.1	32.9	30.7
Great risk	27.5	28.8	32.5	30.9
Not sure	11.4	6.6	4.3	3.9

75. Have five or more drinks once or twice each weekend?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,053)	(n = 4,877)	(n = 4,732)	(n = 3,706)
No risk	6.2%	6.7%	5.7%	6.6%
Slight risk	12.5	12.7	12.6	15.4
Moderate risk	28.5	28.5	29.6	31.3
Great risk	40.9	43.3	47.2	42.4
Not sure	12.0	8.9	5.0	4.2

76. Does your school provide a counselor, intervention specialist, or other school staff member for students to discuss problems with alcohol, tobacco, or other drugs?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 171)	(n = 4,870)	(n = 4,737)	(<i>n</i> = 3,689)
No	7.0%	4.0%	4.5%	8.0%
Yes	67.6	76.1	71.7	65.4
I'm not sure	25.4	19.9	23.8	26.6

77. Putting them all together, what were your grades like last year?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,880)	(n = 4,661)	(n = 4,638)	(n = 3,632)
Mostly A's	45.8%	43.1%	40.1%	34.9%
Mostly B's	36.0	31.4	34.8	40.0
Mostly C's	13.5	17.1	18.4	21.1
Mostly D's	2.9	5.1	4.5	3.2
Mostly F's	1.9	3.3	2.2	0.9

78. Are your school grades better than the grades of most students in your class?

Grade 12
(<i>n</i> = 3,649)
5.1%
31.5
44.1
19.2

79.	How often do y	ou feel the so	choolwork vou a	re assigned is m	eaningful and importa	nt?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4, 116)	(n = 4,842)	(n = 4,707)	(n = 3,688)
Almost always	38.6%	18.0%	10.3%	8.0%
Often	27.9	23.0	18.8	17.6
Sometimes	23.8	35.5	38.7	39.4
Seldom	5.6	15.9	25.2	27.8
Never	4.2	7.7	7.1	7.2

How interesting are most of your courses to you? 80.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,974)	(n = 4,805)	(n = 4,699)	(n = 3,680)
Very interesting and stimulating	16.5%	6.7%	5.2%	6.7%
Quite interesting	36.6	21.2	19.5	23.0
Fairly interesting	30.0	40.5	40.9	40.6
Slightly dull	11.0	20.1	22.8	19.5
Very dull	6.0	11.6	11.6	10.1

.

81. How important do you think the things you are learning in school are going to be for your later life?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,073)	(n = 4,841)	(n = 4,714)	(n = 3,678)
Very important	61.6%	35.5%	24.0%	19.0%
Quite important	24.8	31.1	26.7	24.7
Fairly important	10.1	20.3	29.1	31.5
Slightly important	2.5	9.9	16.0	20.5
Not at all important	1.1	3.2	4.2	4.3

82. Think back over the past year in school, how often did you:

A. Enjoy being in school?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,080)	(n = 4,876)	(n = 4,743)	(n = 3,699)
Never	5.6%	8.6%	7.8%	7.7%
Seldom	7.9	16.1	18.6	20.8
Sometimes	24.5	33.0	37.4	38.8
Often	31.4	28.3	26.6	23.4
Almost always	30.6	14.1	9.7	9.3

Β. Hate being in school?

Grade 6	Grade 8	Grade 10	Grade 12
(n = 4,061)	(n = 4,865)	(n = 4,737)	(n = 3,704)
23.6%	9.6%	6.2%	4.9%
33.7	30.3	26.6	24.0
24.5	29.3	32.9	32.5
11.2	18.8	23.3	26.1
7.0	12.0	11.0	12.5
	Grade 6 (n = 4,061) 23.6% 33.7 24.5 11.2 7.0	Grade 6Grade 8 $(n = 4,061)$ $(n = 4,865)$ 23.6%9.6%33.730.324.529.311.218.87.012.0	Grade 6Grade 8Grade 10 $(n = 4,061)$ $(n = 4,865)$ $(n = 4,737)$ 23.6%9.6%6.2%33.730.326.624.529.332.911.218.823.37.012.011.0

С. Try to do your best work in school?

	Grade 6 $(n = 4,076)$	Grade 8 $(n = 4,864)$	Grade 10 $(n = 4,730)$	Grade 12 (<i>n</i> = 3,689)
Never	1.0%	1.8%	2.0%	1.8%
Seldom	1.6	5.2	6.8	8.2
Sometimes	8.0	16.5	23.1	26.2
Often	23.3	29.8	34.1	34.2
Almost always	66.1	46.7	33.9	29.6

83. During the LAST 4 WEEKS how many whole days of school have you missed:

iess'
11.00
Į

Decause of filless.				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,050)	(n = 4,813)	(n = 4,711)	(n = 3,673)
None	64.9%	60.4%	63.0%	58.6%
1	16.2	18.2	17.0	18.6
2	7.5	9.3	9.5	10.5
3	5.6	6.3	5.4	6.0
4–5	3.9	4.2	3.4	4.9
6–10	1.5	1.3	1.0	1.1
11 or more	0.5	0.4	0.7	0.4

Β. Because you skipped or "cut"?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,003)	(n = 4,774)	(n = 4,673)	(n = 3,658)
None	95.5%	91.4%	84.9%	70.7%
1	2.9	4.2	7.1	12.7
2	0.6	1.2	2.2	6.0
3	0.5	1.1	2.2	4.1
4–5	0.2	0.9	2.0	3.7
6–10	0.1	0.5	0.9	1.6
11 or more	0.2	0.7	0.7	1.1

С. For other reasons?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 4,028)	(n = 4,819)	(n = 4,703)	(n = 3,675)
None	74.8%	68.6%	65.8%	55.3%
1	13.7	16.5	18.0	19.4
2	4.5	6.3	7.1	12.4
3	3.1	3.8	4.7	5.8
4–5	2.3	2.7	2.7	4.6
6–10	1.0	1.2	1.1	1.3
11 or more	0.5	0.9	0.6	1.3

84. In my school, students have lots of chances to help decide things like class activities and rules.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,983)	(n = 4,782)	(n = 4,679)	(<i>n</i> = 3,666)
NO!	9.3%	17.8%	18.6%	18.4%
No	24.3	34.1	35.0	37.7
Yes	48.8	39.6	39.4	37.2
YES!	17.6	8.5	6.9	6.8

85. There are lots of chances for students in my school to talk with a teacher one-on-one.

Grade 6	Grade 8	Grade 10	Grade 12
(n = 4,002)	(n = 4,785)	(n = 4,684)	(n = 3,663)
4.4%	8.0%	9.1%	7.1%
13.3	21.6	23.9	22.3
45.9	48.2	51.3	51.6
36.4	22.3	15.8	19.1
	Grade 6 (n = 4,002) 4.4% 13.3 45.9 36.4	Grade 6Grade 8 $(n = 4,002)$ $(n = 4,785)$ 4.4% 8.0% 13.3 21.6 45.9 48.2 36.4 22.3	Grade 6Grade 8Grade 10 $(n = 4,002)$ $(n = 4,785)$ $(n = 4,684)$ 4.4% 8.0% 9.1% 13.3 21.6 23.9 45.9 48.2 51.3 36.4 22.3 15.8

86. Teachers ask me to work on special classroom projects.

Grade 6	Grade 8	Grade 10	Grade 12
(<i>n</i> = 3,911)	(n = 4,725)	(n = 4,655)	(n = 3,647)
14.3%	18.9%	19.9%	20.8%
34.7	42.4	48.1	45.5
37.0	30.2	26.5	28.2
14.0	8.5	5.5	5.5
	Grade 6 (n = 3,911) 14.3% 34.7 37.0 14.0	Grade 6Grade 8 $(n = 3,911)$ $(n = 4,725)$ 14.3%18.9%34.742.437.030.214.08.5	Grade 6Grade 8Grade 10 $(n = 3,911)$ $(n = 4,725)$ $(n = 4,655)$ 14.3% 18.9% 19.9% 34.7 42.4 48.1 37.0 30.2 26.5 14.0 8.5 5.5

87. There are lots of chances for students in my school to get involved in sports, clubs, and other school activities outside of class.

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = 3,991)	(n = 4,800)	(n = 4,692)	(n = 3,668)
NO!	3.6%	3.2%	4.6%	5.1%
no	7.3	6.0	6.5	7.2
yes	36.9	37.1	35.6	38.1
YES!	52.2	53.7	53.3	49.6

88. I have lots of chances to be part of class discussions or activities.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,953)	(n = 4,764)	(n = 4,678)	(n = 3,660)
NO!	3.7%	4.1%	4.6%	3.4%
no	11.1	11.6	11.8	11.1
yes	44.4	54.4	58.7	58.6
YES!	40.8	29.9	24.9	26.9

89. My teacher(s) notices when I am doing a good job and lets me know about it.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,981)	(n = 4,760)	(n = 4,673)	(<i>n</i> = 3,641)
NO!	4.1%	7.1%	8.8%	8.6%
no	10.6	19.8	27.0	26.6
yes	43.8	47.4	48.8	50.0
YES!	41.5	25.7	15.5	14.9

90. The school lets my parents know when I have done something well.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,936)	(n = 4,729)	(n = 4,658)	(n = 3,623)
NO!	12.9%	22.9%	30.4%	31.1%
no	28.8	38.2	44.1	43.8
yes	34.7	26.7	19.2	18.9
YES!	23.7	12.2	6.3	6.3

91. I feel safe at my school.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,933)	(n = 4,704)	(n = 4,639)	(n = 3,624)
NO!	4.7%	7.7%	7.6%	4.1%
no	9.3	14.9	14.9	10.9
yes	38.6	54.4	57.0	58.1
YES!	47.4	23.0	20.5	26.9

		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,907)	(n = 4,716)	(n = 4,653)	(n = 3, 614)
	NO!	10.8%	16.7%	19.5%	16.7%
	no	24.8	33.3	38.5	38.2
	yes	40.9	36.8	33.8	35.9
	YES!	23.5	13.2	8.2	9.2
<i>93</i> .	How often do kids at school:				
A.	Tell lies or spread rumors about you?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,966)	(n = 4,764)	(n = 4,686)	(n = 3,649)
	Never	35.7%	37.1%	38.7%	42.5%
	Sometimes	46.5	44.3	44.6	42.4
	A lot	11.4	11.2	10.3	8.5
	Every day	6.4	7.4	6.3	6.6
B.	Put you down verbally (insult you, call you names)?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,965)	(n = 4,758)	(n = 4,692)	(n = 3,657)
	Never	36.7%	35.3%	40.3%	50.7%
	Sometimes	42.4	43.1	41.5	36.2
	A lot	12.0	11.3	9.2	7.2
	Every day	8.9	10.3	9.0	5.9
C.	Shove, push, hit or trip you?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,960)	(n = 4,754)	(n = 4,680)	(n = 3,648)
	Never	50.4%	52.8%	66.5%	79.2%
	Sometimes	34.8	32.8	24.1	15.1
	A lot	8.0	7.4	4.4	3.0
	Every day	6.8	7.0	5.0	2.7
D.	Threaten to hurt you physically?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,961)	(n = 4,771)	(n = 4,687)	(n = 3,657)
	Never	73.2%	73.6%	77.7%	86.1%
	Sometimes	18.3	18.2	15.8	9.9
	A lot	4.8	4.2	3.0	1.5
	Every day	3.7	4.1	3.5	2.5
94.	If you saw one kid bullying another at school, would	l you:			
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,829)	(n = 4,597)	(n = 4,577)	(n = 3,554)
	Tell that kid to stop?	39.7%	39.9%	40.1%	48.7%
	Walk away or mind your own business?	11.1	21.1	27.3	23.4
	Tell an adult at school?	41.8	18.5	9.9	8.5
	Stay and watch?	7.4	20.5	22.7	19.4

- 95. This is how safe I feel at school in each of these places:
- A. In class

		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,976)	(n = 4,793)	(n = 4,685)	(n = 3,642)
	Very unsafe	1.8%	2.3%	2.7%	2.3%
	A little unsafe	4.8	6.6	5.0	3.2
	Mostly safe	29.9	44.0	41.4	37.0
	Very safe	63.5	47.1	50.9	57.4
R	In the halls or stairs				
Ъ.	In the huns of stuns	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3.859)	(n = 4.777)	(n = 4.673)	(n = 3.632)
	Very unsafe	4 5%	5.0%	4 7%	3.1%
	A little unsafe	16.4	18 5	14 5	10.2
	Mostly safe	41.1	50.0	49.1	44.7
	Very safe	38.0	26.5	31.7	42.0
C.	In the bathroom				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,921)	(n = 4,757)	(n = 4,667)	(n = 3,624)
	Very unsafe	6.0%	7.1%	6.3%	4.4%
	A little unsafe	18.3	15.6	12.6	8.8
	Mostly safe	36.3	45.2	45.1	42.4
	Very safe	39.4	32.1	35.9	44.5
D.	In the locker rooms				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,047)	(n = 4,727)	(n = 4,610)	(n = 3,573)
	Very unsafe	6.1%	6.3%	5.6%	4.0%
	A little unsafe	13.3	12.3	10.7	8.6
	Mostly safe	35.1	47.2	46.3	41.7
	Very safe	45.6	34.3	37.4	45.7
E.	On the playground/school grounds				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,932)	(n = 4,753)	(n = 4,653)	(n = 3,601)
	Very unsafe	6.4%	6.9%	5.7%	3.9%
	A little unsafe	18.4	16.6	13.0	8.9
	Mostly safe	39.2	48.3	48.3	44.2
	Very safe	36.1	28.2	33.0	43.0
F.	In the lunchroom				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,776)	(n = 4,772)	(n = 4,662)	(n = 3,611)
	Very unsafe	3.0%	4.5%	4.6%	3.5%
	A little unsafe	6.7	8.8	8.4	6.8
	Mostly safe	32.6	44.7	44.8	41.6
	Very safe	57.6	42.0	42.2	48.1

G.	On the bus				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,912)	(n = 4,718)	(<i>n</i> = 4,628)	(n = 3,600)
	Very unsafe	6.5%	5.4%	4.9%	3.5%
	A little unsafe	11.4	9.2	6.9	3.0
	Mostly safe	30.3	28.6	20.5	7.6
	Very safe	23.7	19.8	16.2	5.8
	I don't take the bus	28.0	36.9	51.6	80.1
H.	On the way to school				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,902)	(n = 4,746)	(<i>n</i> = 4,652)	(<i>n</i> = 3,592)
	Very unsafe	4.9%	4.2%	3.6%	2.9%
	A little unsafe	11.7	10.5	7.0	5.9
	Mostly safe	37.1	39.4	37.5	36.4
	Very safe	46.4	45.8	51.9	54.8
I.	On the way home after school				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,907)	(n = 4,724)	(<i>n</i> = 4,635)	(n = 3,577)
	Very unsafe	5.6%	5.0%	4.1%	3.3%
	A little unsafe	12.8	14.4	8.9	6.6
	Mostly safe	39.9	42.8	39.5	36.4
	Very safe	41.7	37.9	47.6	53.8

96. During the past 30 days, how many times have you carried a weapon, such as a gun, knife, or club for self-protection or because you thought you might need it in a fight?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = *)	(n = 4,753)	(n = 4,655)	(n = 3, 613)
None	*	89.0%	89.9%	90.7%
1 time	*	4.5	3.6	3.3
2–3 times	*	2.6	2.4	2.1
4–5 times	*	0.9	0.6	0.7
6 or more times	*	3.0	3.5	3.3

97. When was the last time you carried each of the following weapons on school property for self- protection or because you thought you might need it in a fight?

A. Gun

Β.

Never Over 1 year ago Not within the past month, but within the past year Within the past month	Grade 6 (n = 3,931) 98.7% 0.6 0.2 0.5	Grade 8 (n = 4,698) 97.8% 0.9 0.5 0.9	Grade 10 (n = 4,626) 97.2% 1.2 0.6 1.0	Grade 12 (<i>n</i> = 3,606) 97.8% 1.0 0.4 0.8
Knife or razor	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = 3,918)	(<i>n</i> = 4,689)	(<i>n</i> = 4,615)	(<i>n</i> = 3,600)
Never	94.1%	90.6%	89.7%	91.5%
Over 1 year ago	3.1	4.1	4.4	3.4
Not within the past month, but within the past year	1.1	2.0	2.2	2.0
Within the past month	1.7	3.3	3.8	3.1
С. Club, stick, pipe, or other weapon

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,930)	(<i>n</i> = 4,689)	(n = 4,617)	(n = 3,602)
Never	95.9%	93.5%	94.6%	94.4%
Over 1 year ago	2.2	2.5	1.9	2.1
Not within the past month, but within the past year	0.6	1.5	1.6	1.3
Within the past month	1.3	2.4	1.9	2.2

98. How many times in the past year (12 months) have you?

A. Been suspended from school?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = *)	(n = 4,716)	(<i>n</i> = 4,628)	(<i>n</i> = 3,611)
Never	*	86.4%	90.3%	91.7%
1 or 2 times	*	10.2	7.6	6.6
3 to 5 times	*	1.4	1.2	1.3
6 to 9 times	*	0.9	0.4	0.3
10 to 19 times	*	0.5	0.2	0.1
20 to 29 times	*	0.1	0.0	0.0
30 to 39 times	*	0.2	0.0	0.0
40+ times	*	0.3	0.3	0.1

Β. Carried a handgun?

Ç	Grade 6	Grade 8	Grade 10	Grade 12
	(n = *)	(n = 4,702)	(n = 4,623)	(n = 3,605)
Never	*	95.0%	94.6%	95.5%
1 or 2 times	*	2.2	2.4	1.5
3 to 5 times	*	0.8	0.9	0.8
6 to 9 times	*	0.7	0.6	0.7
10 to 19 times	*	0.4	0.6	0.4
20 to 29 times	*	0.2	0.4	0.3
30 to 39 times	*	0.1	0.1	0.1
40+ times	*	0.6	0.5	0.6

C. Sold illegal drugs?

	Grade 6 $(n = *)$	Grade 8 $(n = 4,674)$	Grade 10 $(n = 4,601)$	Grade 12 $(n = 3,579)$
Never	*	95.3%	90.2%	88.2%
1 or 2 times	*	2.4	3.8	5.0
3 to 5 times	*	0.5	1.6	1.9
6 to 9 times	*	0.5	1.1	1.2
10 to 19 times	*	0.2	0.8	0.8
20 to 29 times	*	0.4	0.5	0.5
30 to 39 times	*	0.2	0.3	0.4
40+ times	*	0.5	1.7	2.0

D. Stolen or tried to steal a motor vehicle such as a car or motorcycle?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = *)	(n = 4,701)	(<i>n</i> = 4,619)	(<i>n</i> = 3,604)
Never	*	97.2%	96.5%	97.8%
1 or 2 times	*	1.5	2.0	1.2
3 to 5 times	*	0.5	0.7	0.5
6 to 9 times	*	0.4	0.1	0.1
10 to 19 times	*	0.0	0.2	0.1
20 to 29 times	*	0.1	0.2	0.1
30 to 39 times	*	0.0	0.1	0.0
40+ times	*	0.3	0.2	0.2

E.	Been arrested?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = *)	(n = 4,670)	(n = 4,608)	(n = 3,588)
	Never	*	94.2%	93.7%	93.7%
	1 or 2 times	*	4.2	4.7	5.0
	3 to 5 times	*	0.7	0.8	0.8
	6 to 9 times	*	0.2	0.4	0.1
	10 to 19 times	*	0.2	0.1	0.1
	20 to 29 times	*	0.1	0.1	0.0
	30 to 39 times	*	0.1	0.0	0.0
	40+ times	*	0.3	0.1	0.3
F.	Attacked someone with the idea of seriously hurtin	g them?			
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = *)	(n = 4,676)	(n = 4,617)	(n = 3,601)
	Never	*	87.6%	89.2%	90.9%
	1 or 2 times	*	8.0	6.9	5.5
	3 to 5 times	*	2.2	1.8	2.0
	6 to 9 times	*	0.9	0.9	0.7
	10 to 19 times	*	0.5	0.3	0.4
	20 to 29 times	*	0.2	0.3	0.1
	30 to 39 times	*	0.1	0.1	0.1
	40+ times	*	0.6	0.5	0.3
G.	Been drunk or high at school?				
	-	Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = *)	(n = 4,700)	(<i>n</i> = 4,611)	(<i>n</i> = 3,595)
	Never	*	89.2%	79.4%	76.5%
	1 or 2 times	*	5.8	8.3	8.1
	3 to 5 times	*	1.9	3.2	4.5
	6 to 9 times	*	0.8	1.8	2.2
	10 to 19 times	*	0.6	1.8	1.9
	20 to 29 times	*	0.2	1.3	1.3
	30 to 39 times	*	0.3	0.5	0.6
	40+ times	*	1.2	3.6	4.9
H.	Taken a handgun to school?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = *)	(<i>n</i> = 4,699)	(<i>n</i> = 4,619)	(n = 3,604)
	Never	*	98.8%	98.3%	98.7%
	1 or 2 times	*	0.7	0.8	0.5
	3 to 5 times	*	0.1	0.2	0.2
	6 to 9 times	*	0.1	0.1	0.2
	10 to 19 times	*	0.1	0.1	0.1
	20 to 29 times	*	0.1	0.1	0.1
	30 to 39 times	*	0.0	0.0	0.1
	40+ times	*	0.2	0.3	0.1

- 99. How old were you when you first:A. Smoked marijuana?

	•	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,925)	(n = 4,723)	(n = 4,623)	(n = 3,606)
	Never have	97.8%	80.3%	62.4%	49.5%
	10 or younger	1.0	3.3	2.7	1.7
	11	1.0	3.4	3.5	2.0
	12	0.2	5.7	4.4	4.4
	13	0.0	6.2	7.8	7.3
	14	0.0	1.1	9.9	10.4
	15	0.0	0.0	8.2	9.8
	16	0.0	0.0	1.2	9.5
	17 or older	0.0	0.0	0.0	5.3
B.	Smoked a cigarette, even just a puff?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = 3,896)	(n = 4,713)	(<i>n</i> = 4,618)	(n = 3,598)
	Never have	84.9%	62.9%	47.8%	39.1%
	10 or younger	11.0	15.6	14.5	10.7
	11	3.7	7.8	8.7	6.0
	12	0.4	7.3	7.5	9.3
	13	0.0	5.7	9.1	9.2
	14	0.0	0.6	7.4	8.6
	15	0.0	0.1	4.4	7.4
	16	0.0	0.0	0.6	6.7
	17 or older	0.0	0.0	0.0	3.1
C.	Smoked a whole cigarette?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,883)	(n = 4,687)	(n = 4,579)	(n = 3,580)
	Never have	92.8%	74.7%	59.1%	48.0%
	10 or younger	4.0	6.4	7.0	4.6
	11	2.7	6.5	6.5	4.7
	12	0.5	6.8	6.8	7.3
	13	0.0	4.8	8.0	8.8
	14	0.0	0.8	7.5	8.6
	15	0.0	0.0	4.4	7.5
	16	0.0	0.0	0.7	7.0
	17 or older	0.0	0.1	0.1	3.6
D.	Smoked a cigar, cigarillo, or little cigar for the first	time?			
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,914)	(n = 4,722)	(n = 4,615)	(n = 3,597)
	Never have	96.6%	87.8%	74.2%	58.4%
	10 or younger	2.0	3.0	2.7	2.1
	11	1.2	3.1	2.4	2.0
	12	0.1	2.8	3.6	3.9
	13	0.0	2.8	4.7	5.6
	14	0.0	0.3	6.5	5.5
	15	0.0	0.0	4.7	7.9
	16	0.0	0.0	0.9	9.2
	17 or older	0.0	0.1	0.3	5.4

E. First used smokeless tobacco (chew, dip, or snuff)?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,903)	(n = 4,693)	(n = 4,588)	(n = 3,577)
Never have	98.2%	94.8%	85.7%	75.2%
10 or younger	1.1	1.4	2.0	1.8
11	0.5	1.1	1.1	1.3
12	0.1	1.1	1.4	1.6
13	0.1	1.3	2.4	2.6
14	0.1	0.3	2.9	4.0
15	0.0	0.0	3.8	5.2
16	0.0	0.0	0.4	4.7
17 or older	0.0	0.0	0.1	3.6

-	TT 1 .1 ·			10 1	11	1 . 1
F.	Had more than a sig	o or two of beer,	wine, or hard li	quor (for example	le, vodka,	whiskey, or gin)?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,896)	(n = 4,683)	(n = 4,604)	(n = 3,584)
Never have	78.8%	54.3%	35.0%	24.0%
10 or younger	13.3	15.2	12.3	10.1
11	6.7	8.3	5.8	3.5
12	1.1	9.5	7.6	5.7
13	0.0	10.7	12.4	9.2
14	0.0	1.9	13.6	12.1
15	0.0	0.1	11.8	15.6
16	0.0	0.0	1.3	13.2
17 or older	0.0	0.1	0.2	6.6

G. Began drinking alcoholic beverages regularly, that is, at least once or twice a month?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,851)	(<i>n</i> = 4,687)	(<i>n</i> = 4,583)	(n = 3,582)
Never have	95.9%	84.0%	66.4%	52.2%
10 or younger	2.0	2.3	1.5	0.6
11	1.7	2.8	1.6	0.8
12	0.3	3.8	2.2	1.9
13	0.0	5.8	5.6	3.0
14	0.0	1.1	9.4	6.6
15	0.0	0.1	11.5	9.1
16	0.0	0.0	1.6	15.7
17 or older	0.0	0.1	0.2	10.2

H. Got suspended from school?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,840)	(n = 4,662)	(n = 4,596)	(n = 3,577)
Never have	92.1%	81.2%	79.4%	77.6%
10 or younger	5.1	5.0	3.8	3.1
11	2.3	4.1	2.1	1.5
12	0.4	4.8	3.1	2.7
13	0.0	4.1	4.2	3.4
14	0.0	0.7	4.4	3.9
15	0.0	0.1	2.6	3.4
16	0.0	0.0	0.3	2.6
17 or older	0.0	0.1	0.2	1.7

I.	Got arrested?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,824)	(n = 4,668)	(n = 4,593)	(n = 3,569)
	Never have	97.8%	93.0%	89.8%	88.3%
	10 or younger	1.2	1.1	0.8	0.3
	11	0.8	1.4	0.7	0.4
	12	0.2	1.7	1.4	0.7
	13	0.0	2.3	1.8	1.4
	14	0.0	0.5	2.1	1.5
	15	0.0	0.0	2.4	1.6
	16	0.0	0.0	0.7	2.8
	17 or older	0.0	0.1	0.2	2.9
J.	Carried a handgun?				
	C C	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,807)	(n = 4,675)	(n = 4,568)	(<i>n</i> = 3,569)
	Never have	95.2%	93.2%	92.3%	93.5%
	10 or younger	2.8	2.2	1.7	1.2
	11	1.8	1.7	0.6	0.3
	12	0.2	1.4	1.1	0.6
	13	0.0	1.1	0.9	1.0
	14	0.0	0.3	1.2	0.8
	15	0.0	0.0	1.4	0.9
	16	0.0	0.0	0.7	0.9
	17 or older	0.0	0.1	0.0	0.9
K	Attacked someone with the idea of seriously h	urting them?			
	r Rucked Someone with the field of Schousry h	and the them?			

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,830)	(n = 4,672)	(n = 4,584)	(n = 3,579)
Never have	89.9%	84.7%	83.9%	83.2%
10 or younger	6.0	5.1	4.6	3.4
11	3.1	2.2	1.8	0.9
12	0.8	3.1	1.8	2.2
13	0.1	3.8	2.8	2.1
14	0.0	0.9	2.2	1.5
15	0.0	0.0	2.4	2.5
16	0.0	0.0	0.3	1.9
17 or older	0.1	0.1	0.1	2.3

L. Belonged to a gang (sometimes referred to as an organization, click, clique, set, or posse)?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,831)	(n = 4,668)	(n = 4,588)	(n = 3,583)
Never have	95.0%	88.3%	88.8%	89.7%
10 or younger	2.7	3.8	2.5	1.1
11	1.7	2.1	1.4	1.2
12	0.5	2.6	1.5	1.6
13	0.0	2.7	2.2	1.4
14	0.0	0.4	1.8	1.4
15	0.0	0.0	1.5	1.2
16	0.0	0.0	0.2	1.4
17 or older	0.0	0.1	0.1	1.0

100. How wrong do you think it is for someone your age to:A. Take a handgun to school?

		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,785)	(<i>n</i> = 4,648)	(<i>n</i> = 4,581)	(n = 3,572)
	Very wrong	92.8%	86.3%	87.7%	91.4%
	Wrong	5.6	10.5	9.2	6.2
	A little bit wrong	1.0	2.2	1.8	1.6
	Not wrong at all	0.6	1.0	1.4	0.8
R	Steal anything worth less than \$5?				
D.	Stear anything worth less than ϕ_2 :	Grade 6	Grade 8	Grade 10	Grade 12
		(n - 3.773)	(n - 4.633)	(n - 4.577)	(n - 3.560)
	Very wrong	(n = 3, 773)	(n = 4,000)	(n = 4,377)	(n = 3,300)
	Wrong	35.0	35.6	37.1	3/ 8
	A little bit wrong	11.1	24.0	24.6	18.1
	Not wrong at all	2.1	24.0 6.2	24.0 5 1	10.1
	Not wrong at an	2.1	0.2	5.1	4.5
C.	Steal anything worth more than \$5?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,730)	(n = 4,548)	(n = 4,542)	(n = 3,539)
	Very wrong	80.2%	56.1%	53.7%	59.5%
	Wrong	15.3	29.4	30.0	27.6
	A little bit wrong	3.4	11.2	12.8	10.8
	Not wrong at all	1.1	3.3	3.5	2.1
D.	Pick a fight with someone?				
	C C	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,755)	(n = 4,598)	(n = 4,522)	(n = 3,535)
	Very wrong	54.6%	32.5%	30.8%	35.4%
	Wrong	29.7	35.5	35.9	38.1
	A little bit wrong	11.7	23.8	25.7	21.5
	Not wrong at all	4.1	8.3	7.6	5.0
E.	Attack someone with the idea of seriously burting th	em?			
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3.752)	(n = 4.624)	(n = 4.573)	(n = 3.557)
	Very wrong	83.3%	65.6%	62.1%	65.5%
	Wrong	11.7	21.9	22.7	22.8
	A little bit wrong	33	8.8	11.2	8.8
	Not wrong at all	1.7	3.7	4.0	2.9
_					
F.	Stay away from school all day when their parents thi	nk they are at	school?	G 1 10	G 1 10
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,758)	(n = 4,634)	(n = 4,580)	(n = 3,553)
	Very wrong	81.4%	52.1%	35.0%	27.3%
	Wrong	13.5	29.0	34.1	33.6
	A little bit wrong	3.7	13.8	22.1	28.0
	Not wrong at all	1.5	5.1	8.8	11.1

101. How wrong do you think it is for someone your age to:A. Drink beer, wine, or hard liquor (for example, vodka, whiskey, or gin) regularly?

		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,816)	(n = 4,600)	(n = 4,572)	(n = 3,560)
	Very wrong	81.7%	53.4%	31.7%	22.9%
	Wrong	12.8	24.4	27.2	26.7
	A little bit wrong	4.4	15.5	27.3	32.0
	Not wrong at all	1.1	6.7	13.8	18.4
B.	Smoke cigarettes?				
	C C	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,797)	(n = 4,578)	(n = 4,570)	(n = 3,550)
	Very wrong	84.0%	56.2%	38.7%	27.5%
	Wrong	11.3	25.5	28.6	24.0
	A little bit wrong	3.5	12.0	20.8	24.6
	Not wrong at all	1.2	6.3	11.8	24.0
C.	Smoke marijuana?				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,784)	(n = 4,577)	(n = 4,572)	(n = 3,542)
	Very wrong	92.9%	66.9%	45.2%	36.0%
	Wrong	4.5	15.8	22.6	23.7
	A little bit wrong	1.7	9.5	17.2	22.3
	Not wrong at all	1.0	7.8	15.1	18.0
D.	Use LSD, cocaine, amphetamines or another illegal	drug?			
	-	Grade 6	Grade 8	Grade 10	Grade 12
		(2705)	(1507)	(4,501)	(2550)

Office 0	Ordue 0	Orace 10	Orace 12
(n = 3,795)	(n = 4,597)	(<i>n</i> = 4,581)	(n = 3,558)
95.1%	83.9%	76.1%	75.9%
3.6	10.7	13.5	13.0
0.6	3.2	6.1	7.2
0.8	2.1	4.3	3.9
	(n = 3,795) 95.1% 3.6 0.6 0.8	$\begin{array}{cccc} (n=3,795) & (n=4,597) \\ 95.1\% & 83.9\% \\ 3.6 & 10.7 \\ 0.6 & 3.2 \\ 0.8 & 2.1 \end{array}$	$\begin{array}{ccccc} (n=3,795) & (n=4,597) & (n=4,581) \\ 95.1\% & 83.9\% & 76.1\% \\ 3.6 & 10.7 & 13.5 \\ 0.6 & 3.2 & 6.1 \\ 0.8 & 2.1 & 4.3 \end{array}$

102. Think of your *four best friends* (the friends you feel closest to). In the past year (12 months), how many of your best friends have:

Α. Smoked cigarettes?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,749)	(n = 4,546)	(n = 4,545)	(n = 3,541)
None	85.3%	59.7%	45.2%	35.1%
1	7.8	16.3	16.3	16.0
2	3.5	9.1	13.9	13.2
3	1.0	5.0	7.8	12.7
4	2.5	10.0	16.8	22.9

Β. Tried beer, wine, or hard liquor (for example, vodka, whiskey, or gin) when their parents didn't know about it?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,740)	(n = 4,530)	(n = 4,534)	(n = 3,527)
None	83.0%	50.4%	27.2%	18.8%
1	10.0	17.1	14.8	11.9
2	3.1	10.9	14.0	12.3
3	1.4	6.0	12.7	12.1
4	2.5	15.7	31.3	44.9

C. Used marijuana?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,728)	(n = 4,521)	(n = 4,540)	(n = 3,523)
None	94.1%	67.8%	44.6%	36.0%
1	3.3	11.7	15.4	16.0
2	1.1	5.8	10.9	13.0
3	0.6	4.1	9.4	9.4
4	0.9	10.6	19.7	25.6

D. Used LSD, cocaine, amphetamines, or other illegal drugs?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,742)	(n = 4,542)	(n = 4,542)	(n = 3,535)
None	96.9%	88.6%	78.5%	74.3%
1	1.7	6.1	10.8	11.2
2	0.7	2.1	4.3	5.6
3	0.3	0.8	1.9	2.6
4	0.5	2.4	4.4	6.4

103. What are the chances you would be seen as cool if you:

A. Smoked cigarettes?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,746)	(n = 4,509)	(n = 4,503)	(n = 3,493)
No or very little chance	79.9%	61.2%	60.2%	64.5%
Little chance	10.9	19.9	22.2	20.6
Some chance	4.5	10.9	11.0	10.9
Pretty good chance	2.8	4.9	4.3	2.6
Very good chance	1.9	3.0	2.3	1.5

B. Began drinking alcoholic beverages regularly, that is, at least once or twice a month?

6 6	U	0,00	Grada 6	Grada 8	Grada 10	Grada 12
			Ofade 0	Utaue o	Utade 10	Utaue 12
			(n = 3,731)	(n = 4,501)	(n = 4,487)	(n = 3,492)
No or very little chance			85.0%	59.9%	45.0%	47.1%
Little chance			8.2	19.2	20.3	18.7
Some chance			3.7	11.1	18.7	18.9
Pretty good chance			1.4	5.6	11.3	10.0
Very good chance			1.8	4.3	4.6	5.4

C. Smoked marijuana?

Very good chance

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,737)	(n = 4,492)	(n = 4,488)	(<i>n</i> = 3,490)
No or very little chance	87.2%	62.4%	50.7%	54.2%
Little chance	6.3	15.6	19.5	18.5
Some chance	2.5	9.7	14.9	15.6
Pretty good chance	1.6	6.3	9.1	7.1
Very good chance	2.4	6.0	5.8	4.6
Carried a handgun?				
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,734)	(<i>n</i> = 4,501)	(<i>n</i> = 4,495)	(<i>n</i> = 3,498)
No or very little chance	87.0%	79.6%	83.7%	87.9%
Little chance	5.7	10.7	10.1	7.3
Some chance	3.2	4.5	2.4	2.7
Pretty good chance	1.3	2.3	1.9	1.0

2.9

1.9

1.0

2.9

D.

104.	I think it is okay to take something without asking if	you can get av	way with it.	a	~
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,724)	(n = 4,446)	(n = 4,487)	(n = 3,488)
	NO!	76.1%	53.2%	49.2%	54.5%
	No	19.5	33.3	38.1	34.4
	Yes	3.1	9.8	8.9	8.1
	YES!	1.3	3.6	3.9	3.0
105.	I think sometimes it's okay to cheat at school.				
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,703)	(n = 4,425)	(n = 4,478)	(<i>n</i> = 3,481)
	NO!	73.4%	41.6%	26.0%	26.8%
	no	19.6	32.1	33.1	33.5
	ves	5.5	20.5	32.1	30.3
	YES!	1.5	5.8	8.8	9.3
106.	It is all right to beat up people if they start the fight.				
100.		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3.695)	(n = 4.412)	(n = 4.475)	(n = 3.476)
	NO	64 9%	39.0%	33.2%	32 5%
	no.	16.4	22.1	22.270	21.7
	Vec	10.4	20.1	22.2	24.7
	VESI	7.0	18.8	23.4	24.7
	125:	1.9	10.0	21.2	21.2
107.	It is important to be honest with your parents, even if	f they become	upset or you g	et punished.	G 1 10
		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 3,718)	(n = 4,417)	(n = 4,464)	(n = 3,455)
	NO!	19.5%	14.4%	12.3%	9.6%
	no	5.2	13.1	14.6	15.9
	yes	16.1	32.1	37.3	38.0
	YES!	59.2	40.4	35.9	36.4
108.	You're looking at CDs in a music store with a friend.	You look up	and see her slip	p a CD under h	er coat. She
	smiles and says "Which one do you want? Go ahead,	, take it while	nobody's aroui	nd." There is no	obody in
	sight, no employees and no other customers. What w	ould you do n	ow?		
		Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = *)	(n = 4,372)	(n = 4,446)	(n = 3,444)
	Ignore her	*	21.2%	29.2%	33.8%
	Grab a CD and leave the store	*	12.4	13.4	9.6
	Tell her to put the CD back	*	37.9	28.4	29.4
	Act like it's a joke, and ask her to put the CD back	*	28.4	29.0	27.3
109.	It's 8:00 on a weeknight and you are about to go over	r to a friend's h	nome when yo	ur mother asks	vou where
	you are going. You say "Oh, just going to go hang ou	ut with some f	riends." She sa	iys, "No, you'll	just get into
	trouble if you go out. Stay home tonight." What wou	ld you do now	/?		
		Grade 6	Grade 8	Grade 10	Grade 12
		(<i>n</i> = *)	(<i>n</i> = 4,291)	(n = 4,380)	(n = 3,387)
	Leave the house anyway	*	8.9%	9.5%	10.8%
	Explain what you are going to do with your friends,	*	68.5	71.8	75.3
	tell her when you'd get home, and ask if you can				

*

*

13.3

9.4

7.8

10.9

go out Not say anything and start watching TV Get into an argument with her

4.3

9.6

110. You are visiting another part of town and you don't know any of the people your age there. You are walking down the street and some teenager you don't know is walking toward you. He is about your size, and as he is about to pass you, he deliberately bumps into you and you almost lose your balance. What would you say or do?

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(n = 4,286)	(n = 4,351)	(<i>n</i> = 3,390)
Push the person back	*	11.2%	11.7%	11.4%
Say nothing and keep on walking	*	45.2	45.7	44.0
Say "Watch where you're going" and keep on	*	28.3	27.4	30.3
walking				
Swear at the person and walk away	*	15.3	15.1	14.3

111. You are at a party at someone's house, and one of your friends offers you a drink containing alcohol. What would you say or do?

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(n = 4,313)	(n = 4,376)	(n = 3,397)
Drink it	*	26.5%	44.1%	52.0%
Tell your friend "No thanks, I don't drink" and	*	33.6	24.9	19.5
suggest that you and your friend go and do				
something else				
Just say, "No, thanks" and walk away	*	30.9	25.3	24.9
Make up a good excuse, tell your friend you had	*	8.9	5.7	3.6
something else to do, and leave				

112. When I am an adult I will smoke cigarettes.

the first and an addit i will billone ergarettest				
C C	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,671)	(n = 4,327)	(n = 4,394)	(<i>n</i> = 3,419)
NO!	77.3%	68.9%	73.8%	71.9%
no	17.5	22.5	16.7	17.7
yes	4.0	6.1	6.7	7.0
YES!	1.2	2.5	2.8	3.4

113. When I am an adult I will drink beer, wine, or liquor.

which I am an addit I will armik beer, which of hequor	•			
	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,631)	(n = 4,309)	(n = 4,385)	(<i>n</i> = 3,418)
NO!	50.3%	31.1%	23.5%	20.2%
no	23.8	23.6	17.5	15.2
yes	22.3	34.8	41.3	42.7
YES!	3.6	10.5	17.8	21.9

114. When I am an adult I will smoke marijuana.

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,648)	(n = 4,327)	(n = 4,376)	(<i>n</i> = 3,411)
NO!	91.8%	76.9%	70.3%	69.3%
no	6.6	13.0	15.7	16.7
yes	1.1	5.4	8.3	8.2
YES!	0.5	4.7	5.8	5.8

115. During the past 12 months, did you ever feel so sad or hopeless almost every day for two weeks or more in a row that you stopped doing some usual activities?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,613)	(n = 4,222)	(n = 4,307)	(n = 3,378)
No	76.9%	69.6%	66.0%	68.0%
Yes	23.1	30.4	34.0	32.0

116.	How honest	were vou	in filling	out this	survev?
	110			000000000	our of the

	Grade 6	Grade 8	Grade 10	Grade 12	
	(n = 3,630)	(n = 4,304)	(<i>n</i> = 4,361)	(<i>n</i> = 3,411)	
I was very honest	88.7%	84.2%	87.2%	89.8%	
I was honest pretty much of the time	11.3	15.7	12.8	10.2	
I was honest some of the time	0.1	0.1	0.0	0.0	
I was honest once in a while	Surveys pulled				
I was not honest at all		Surveys	pulled		

117.	How many	times. i	f anv. l	have voi	u tried to	auit using	tobacco	(smoking	or chew/dip)?
								(

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 3,626)	(n = 4,269)	(n = 4,330)	(n = 3,390)
I have never used tobacco	92.4%	77.3%	65.6%	55.3%
None	2.7	7.8	12.3	14.7
1 time	2.3	7.2	9.8	10.8
2 times	1.1	2.7	5.3	7.2
3–5 times	0.8	2.4	3.3	6.6
6–9 times	0.2	0.7	0.9	1.6
10+ times	0.6	2.0	2.8	3.9

118. When you last tried to quit using tobacco (smoking or chew/dip), how long did you stay off it?

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = *)	(n = 2, 101)	(n = 2,219)	(<i>n</i> = 1,894)
I have never used tobacco	*	59.1%	39.8%	28.0%
I have never tried to quit	*	10.7	16.5	19.8
Less than a day	*	4.2	4.9	5.1
1–7 days	*	4.7	8.3	14.6
More than 7 days but less than 30 days	*	2.5	5.5	7.0
More than 30 days but less than 6 months	*	4.7	6.1	8.4
More than 6 months but less than a year	*	2.5	3.0	3.2
More than a year	*	11.5	16.0	13.9

119. Have you ever participated in a program to help you quit using tobacco?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = *)	(n = 2,068)	(n = 2,227)	(<i>n</i> = 1,897)
I have never used tobacco	*	55.8%	37.5%	24.5%
No	*	41.0	58.4	72.3
Yes	*	3.2	4.2	3.2

120. Have you ever used tobacco regularly, that is, at least one cigarette or chew/dip every day for 30 days?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 1,476)	(n = 2,049)	(n = 2,213)	(<i>n</i> = 1,889)
No	94.6%	84.3%	69.7%	55.4%
Yes	5.4	15.7	30.3	44.6

121. If you use tobacco (cigarettes or chew/dip), how do you usually get it?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 1, 141)	(n = 1,453)	(n = 1,540)	(n = 1,338)
I don't use tobacco	88.6%	66.4%	49.0%	32.2%
Buy it from a store	0.8	2.7	8.3	37.9
From my parents	1.2	2.1	3.6	2.4
From someone under 18	2.5	8.6	9.6	2.7
From someone 18 or older	2.3	8.4	22.6	21.6
Take or steal it from a store or family member	1.3	5.3	2.4	0.7
Get it some other way	3.3	6.5	4.6	2.5

122. If you currently use tobacco (smoking or chew/dip), how many times have you seriously tried to quit in the past 12 months?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 1, 115)	(n = 1,476)	(n = 1,578)	(n = 1,354)
I don't use tobacco	88.0%	66.5%	54.4%	36.9%
None	4.6	13.0	19.0	22.7
1 time	3.2	9.2	11.7	17.2
2–3 times	1.5	6.1	9.1	16.1
4–6 times	1.2	2.2	3.0	4.0
7 or more times	1.5	3.0	2.7	3.0

123. During the past 30 days, on the days you smoked, about how many cigarettes did you smoke per day?

	Grade 6	Grade 8	Grade 10	Grade 12
	(<i>n</i> = 1,116)	(n = 1,476)	(n = 1,579)	(n = 1,348)
I do not smoke	89.9%	68.6%	57.8%	45.8%
Less than 1 cigarette per day	4.2	8.6	8.6	6.4
1 cigarette per day	2.1	5.7	5.6	5.8
2–5 cigarettes per day	2.1	10.4	17.4	20.7
6–10 cigarettes per day	0.5	2.5	6.0	14.1
11–20 cigarettes per day	0.6	1.5	3.2	5.1
More than 1 pack per day	0.5	2.7	1.5	2.1

124. Are the cigarettes you usually smoke menthol cigarettes?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = *)	(n = 1,479)	(n = 1,576)	(n = 1,351)
I do not smoke cigarettes	*	66.0%	55.5%	42.5%
No	*	20.8	34.6	46.3
Yes	*	13.2	9.9	11.2

141. Does anyone who lives with you now smoke cigarettes?

,	U	Grade 6 (<i>n</i> = 1,529)	Grade 8 $(n = 1, 139)$	Grade 10 $(n = 1,868)$	Grade 12 (<i>n</i> = 1,405)
No Yes		57.5% 42.5	61.3% 38.7	62.6% 37.4	63.6% 36.4

142. During the past 7 days, on how many days were you in the same room with someone who was smoking cigarettes?

	Grade 6	Grade 8	Grade 10	Grade 12
	(n = 1,519)	(n = 1, 140)	(n = 1,877)	(n = 1,403)
None	62.0%	54.3%	51.0%	44.8%
1–2 days	13.6	16.0	20.2	20.7
3–4 days	6.4	6.9	7.1	9.5
5–6 days	2.3	4.5	3.3	4.8
7 days	15.7	18.2	18.4	20.1

143. During the past 7 days, on how many days did you ride in a car with someone who was smoking cigarettes?

Grade 6	Grade 8	Grade 10	Grade 12
(n = 1,526)	(n = 1, 135)	(n = 1,866)	(n = 1,403)
62.1%	60.0%	59.6%	56.8%
16.0	15.7	15.8	18.0
6.4	6.9	8.1	7.8
4.4	4.5	4.6	4.3
11.0	12.9	12.0	13.1
	Grade 6 (n = 1,526) 62.1% 16.0 6.4 4.4 11.0	Grade 6Grade 8 $(n = 1,526)$ $(n = 1,135)$ 62.1% 60.0% 16.0 15.7 6.4 6.9 4.4 4.5 11.0 12.9	Grade 6Grade 8Grade 10 $(n = 1,526)$ $(n = 1,135)$ $(n = 1,866)$ 62.1% 60.0% 59.6% 16.0 15.7 15.8 6.4 6.9 8.1 4.4 4.5 4.6 11.0 12.9 12.0

144. Other illegal drugs

Yes

		Grade 6	Grade 8	Grade 10	Grade 12
		(n = 4, 148)	(<i>n</i> = *)	(<i>n</i> = *)	(<i>n</i> = *)
	None	99.0%	*	*	*
	1–2 days	0.5	*	*	*
	3–5 days	0.2	*	*	*
	6–9 days	0.1	*	*	*
	10 or more days	0.3	*	*	*
	Any use in past 30 days	1.0	*	*	*
145.	Other illegal drugs				
	0 0	Grade 6	Grade 8	Grade 10	Grade 12
		(n = 4, 190)	(<i>n</i> = *)	(n = *)	(n = *)
	No	97.6%	*	*	*
	Yes	2.4	*	*	*

2.4

Appendix B Selected Results by Ethnic Group

	Percent of survey participants					
Grade/Ethnicity	None	1–2	3–5	6-9	10+	n
Grade 6***						
Black, non-Hispanic	94.6	2.3	1.5	0.8	0.8	259
Asian or Pacific Islander	96.6	2.0	0.6	0.0	0.9	351
American Indian or Alaskan Native	89.9	7.7	0.3	0.7	1.4	286
Hispanic	95.5	2.4	0.9	0.3	0.9	331
White, non-Hispanic	96.8	1.7	0.4	0.1	1.1	2,755
Grade 8***						
Black, non-Hispanic	82.5	5.4	4.1	1.3	6.7	315
Asian or Pacific Islander	92.6	4.1	1.7	0.7	1.0	418
American Indian or Alaskan Native	75.4	6.0	4.3	3.0	11.2	232
Hispanic	83.4	8.1	2.7	1.0	4.9	409
White, non-Hispanic	88.9	4.5	1.7	1.2	3.7	3,433
Grade 10***						
Black, non-Hispanic	77.4	3.8	0.5	3.2	15.1	186
Asian or Pacific Islander	85.4	3.4	3.4	1.1	6.7	357
American Indian or Alaskan Native	58.7	7.9	6.3	0.8	26.2	126
Hispanic	79.1	4.3	3.0	1.7	11.9	464
White, non-Hispanic	80.8	5.6	2.5	1.2	9.8	3,545
Grade 12***						
Black, non-Hispanic	70.6	14.0	1.5	2.2	11.8	136
Asian or Pacific Islander	82.9	4.0	1.3	1.3	10.4	298
American Indian or Alaskan Native	55.1	5.6	0.0	15.7	23.6	89
Hispanic	77.2	4.9	1.9	2.3	13.7	263
White, non-Hispanic	71.3	6.7	2.6	1.8	17.7	2,848

Table B1:Prevalence of 30-Day Cigarette Use by Ethnic Group
Percent of Survey Participants

		Percent of survey participants						
Grade/Ethnicity	None	1–2	3–5	6–9	10+	n		
Grade 6***								
Black, non-Hispanic	88.7	7.8	0.8	0.4	2.3	257		
Asian or Pacific Islander	95.4	3.4	0.6	0.3	0.3	348		
American Indian or Alaskan Native	87.4	7.4	1.4	0.7	3.2	285		
Hispanic	86.9	10.1	1.5	0.6	0.9	327		
White, non-Hispanic	94.8	3.9	0.5	0.3	0.4	2,745		
Grade 8***								
Black, non-Hispanic	73.2	12.1	7.0	3.5	4.2	313		
Asian or Pacific Islander	83.3	8.7	2.9	1.9	3.1	414		
American Indian or Alaskan Native	71.2	15.7	7.4	1.7	3.9	229		
Hispanic	71.6	16.5	4.7	4.4	2.7	405		
White, non-Hispanic	78.7	12.2	3.5	2.1	3.6	3,396		
Grade 10***								
Black, non-Hispanic	58.2	8.8	15.9	6.0	11.0	182		
Asian or Pacific Islander	68.3	17.8	5.9	3.1	4.8	353		
American Indian or Alaskan Native	46.8	20.2	12.9	14.5	5.6	124		
Hispanic	56.8	19.9	7.6	7.1	8.6	463		
White, non-Hispanic	63.3	19.1	7.8	4.0	5.8	3,517		
Grade 12***								
Black, non-Hispanic	52.6	23.4	10.9	2.2	10.9	137		
Asian or Pacific Islander	62.2	18.4	8.4	5.0	6.0	299		
American Indian or Alaskan Native	40.9	22.7	20.5	5.7	10.2	88		

Table B2:Prevalence of 30-Day Alcohol Use by Ethnic Group
Percent of Survey Participants

Hispanic	47.1	19.5	14.2	7.3	11.9	261
White, non-Hispanic	53.0	21.9	11.6	6.3	7.2	2,836

***p < .01.

Table B3:Prevalence of 30-Day Marijuana Use by Ethnic Group
Percent of Survey Participants

	Percent of survey participants					
Grade/Ethnicity	None	1–2	3–5	6-9	10+	n
Grade 6***						
Black, non-Hispanic	98.0	1.2	0.4	0.0	0.4	254
Asian or Pacific Islander	98.3	0.6	0.3	0.6	0.3	350
American Indian or Alaskan Native	95.5	2.8	0.7	0.0	1.0	289
Hispanic	96.9	1.8	0.9	0.0	0.3	326
White, non-Hispanic	99.1	0.4	0.2	0.1	0.2	2,741
Grade 8***						
Black, non-Hispanic	74.2	9.9	5.1	2.2	8.6	314
Asian or Pacific Islander	91.2	5.6	1.0	0.0	2.2	410
American Indian or Alaskan Native	79.9	8.7	3.1	3.5	4.8	229
Hispanic	83.5	6.2	2.2	2.5	5.7	406
White, non-Hispanic	90.0	4.2	1.8	1.3	2.6	3,389
Grade 10***						
Black, non-Hispanic	59.4	7.8	5.6	5.6	21.7	180
Asian or Pacific Islander	83.2	4.3	2.8	3.1	6.5	352
American Indian or Alaskan Native	61.0	15.4	6.5	4.9	12.2	123
Hispanic	71.2	7.8	5.6	4.1	11.3	462
White, non-Hispanic	80.0	7.2	3.7	2.3	6.8	3,519
Grade 12**						
Black, non-Hispanic	73.0	5.8	3.6	5.8	11.7	137

Asian or Pacific Islander	80.5	7.4	3.0	0.7	8.4	297	
American Indian or Alaskan Native	70.0	13.3	7.8	6.7	2.2	90	
Hispanic	74.0	7.3	5.3	3.1	10.3	262	
White, non-Hispanic	75.5	8.6	4.2	2.5	9.2	2,836	

p < .01. *p < .001.

Table B4:Prevalence of Two Week Binge Drinking Use by Ethnic Group
Percent of Survey Participants

		Percent of survey participants					
Grade/Ethnicity		None	1–2	3–5	6–9	10+	n
Grade 6***							
Black, non-Hispanic	91.0	5.3	1.2	2.0	0.0	0.4	244
Asian or Pacific Islander	97.9	1.5	0.0	0.3	0.0	0.3	337
American Indian or Alaskan Native	90.2	3.3	2.5	2.5	0.4	1.1	275
Hispanic	90.4	5.8	1.9	1.0	0.3	0.6	313
White, non-Hispanic	96.2	2.2	1.0	0.5	0.0	0.1	2,634
Grade 8***							
Black, non-Hispanic	79.4	6.4	4.2	6.1	1.3	2.6	311
Asian or Pacific Islander	88.9	4.6	2.4	2.7	0.7	0.7	415
American Indian or Alaskan Native	78.9	5.2	6.0	7.3	0.9	1.7	232
Hispanic	78.3	11.4	3.6	2.4	1.2	2.9	411
White, non-Hispanic	86.3	5.3	3.4	2.9	1.0	1.1	3,403
Grade 10***							
Black, non-Hispanic	73.1	8.8	6.6	6.6	2.2	2.7	182
Asian or Pacific Islander	82.4	5.7	4.2	4.0	0.8	2.8	353
American Indian or Alaskan Native	55.3	8.1	13.0	13.8	4.9	4.9	123
Hispanic	69.5	9.3	8.2	4.8	3.2	5.0	463

White, non-Hispanic	78.3	8.1	5.6	5.1	1.4	1.6	3,528
Grade 12**							
Black, non-Hispanic	77.2	5.1	5.9	8.1	2.2	1.5	136
Asian or Pacific Islander	76.2	9.2	4.4	7.1	1.4	1.7	294
American Indian or Alaskan Native	63.3	10.0	13.3	6.7	3.3	3.3	90
Hispanic	53.8	15.3	8.8	9.2	5.7	7.3	262
White, non-Hispanic	68.5	11.3	7.7	7.7	1.9	2.9	2,842

***p < .001.

Table B5:Prevalence of 30-Day Weapon Carrying by Ethnic Group
Percent of Survey Participants

Grade/Ethnicity	None	1–2	3–5	6-9	10+	n
Grade 8***						
Black, non-Hispanic	82.9	5.1	4.8	2.4	4.8	292
Asian or Pacific Islander	88.1	5.4	1.7	1.5	3.2	404
American Indian or Alaskan Native	82.2	6.4	6.4	2.7	2.3	219
Hispanic	84.1	6.5	2.2	1.6	5.7	371
White, non-Hispanic	90.8	3.8	2.3	0.5	2.5	3,324
Grade 10***						
Black, non-Hispanic	81.1	7.7	4.1	1.8	5.3	169
Asian or Pacific Islander	90.9	3.8	1.5	0.9	2.9	342
American Indian or Alaskan Native	86.4	5.9	3.4	0.0	4.2	118
Hispanic	82.9	6.7	3.6	0.7	6.1	445
White, non-Hispanic	91.4	2.8	2.1	0.6	3.0	3,463
Grade 12***						
Black, non-Hispanic	90.4	3.2	0.8	0.8	4.8	125
Asian or Pacific Islander	93.7	2.1	2.1	0.7	1.4	284

American Indian or Alaskan Native	88.2	1.2	3.5	1.2	5.9	85
Hispanic	87.1	5.6	4.0	1.6	1.6	248
White, non-Hispanic	90.8	3.3	1.9	0.7	3.3	2,761

Note. Grade 6 students were not administered this question. ***p < .001.

	Percent of survey participants		
Grade/Ethnicity	No	Yes	n
Grade 6***			
Black, non-Hispanic	73.2	26.8	183
Asian or Pacific Islander	73.4	26.6	289
American Indian or Alaskan Native	67.4	32.6	233
Hispanic	69.3	30.7	290
White, non-Hispanic	79.2	20.8	2,397
Grade 8**			
Black, non-Hispanic	68.1	31.9	254
Asian or Pacific Islander	70.1	29.9	371
American Indian or Alaskan Native	57.3	42.7	178
Hispanic	65.5	34.5	319
White, non-Hispanic	70.9	29.1	2,988
Grade 10***			
Black, non-Hispanic	69.3	30.7	150
Asian or Pacific Islander	67.8	32.2	301
American Indian or Alaskan Native	47.7	52.3	107
Hispanic	59.1	40.9	372
White, non-Hispanic	67.0	33.0	3,275
Grade 12*			
Black, non-Hispanic	61.9	38.1	113
Asian or Pacific Islander	65.6	34.4	259
American Indian or Alaskan Native	58.8	41.3	80
Hispanic	62.8	37.2	218
White, non-Hispanic	69.3	30.7	2,609

Table B6:Prevalence of 12-Month Depressive Feelings by Ethnic Group

*p < .05. **p < .01. ***p < .001.